MSc Thesis Report Positive tipping points in the adoption of sustainable energy technologies

Sara Raposeiro, 4827554





MSc Thesis Report

Positive tipping points in the adoption of sustainable energy technologies

Sara Raposeiro, 4827554

University:Delft University of Technology, DelftStudy Programme:MSc Sustainable Energy TechnologyFirst Supervisor:Dr. L. M. KampSecond Supervisor:Dr. A. CorreljéDeloitte Supervisor:Dr. J. van den BerkDate:May 28, 2024Word count (main text):41,448

Cover: Gijs Broere. (2023). Handboekje Duurzamer Leven. Dwars. Retrieved from https://dwars.org/overdwars/ handboekje-duurzamer-leven/



Preface

Since a young age I have been passionately driven to contribute to fighting climate change and to safeguarding a better future for us and for the next generations. I felt like the best way to do so was to work on the energy field, which is full of tangible options to rapidly reduce anthropogenic emissions. This ambition led to me to pursue the MSc Sustainable Energy Technologies. This inter-faculty programme has provided me with an overview of different sustainable energy technologies as well as how they fit in the broader scope of a country's energy needs.

This thesis is a culmination of this systems perspective, and I am grateful for the help of several people during the process of completing it. I would like to thank my first supervisor, Linda M. Kamp, for her constant guidance and feedback over the last months, which allowed the work done to have a higher quality than it would otherwise have had. The critical attitude of my second supervisor, Aad Correljé, was also essential to have a fresh perspective on the topics analysed, such that relevant aspects were not overseen. I am also thankful for all the interviewees who agreed to speak to me, who sparked my creativity and helped me validate my work.

I want to thank Deloitte for this thesis opportunity and for the people it led me to meet. I received a warm welcome from the Sustainability Team, who is always willing to lend a helping hand. In particular, I am very grateful for my Deloitte supervisor, Jelle van den Berk, who was present for my thesis colleagues and I, always having an advice or comforting words to give. I would also like to thank my fellow thesis interns, Isabel Klennert and Rosita Tombari, whose companionship and help made these last nine months more enjoyable.

Writing a thesis is full of challenges and overcoming them is made easier by finding support and comfort in your friends and family. A thank you note goes to my close university friends, my second family, who were by my side in every step of these last six years. I would also like to thank my boyfriend, António, in specific, who has been a major source of support. Finally, I am grateful for my family and the support they have always given me. The TU Delft experience would not have been possible without them.

This Master's thesis concludes my six-year journey at TU Delft. It has been a truly enriching chapter of my life, full of challenges but also of many rewarding moments.

Sara Raposeiro Delft, May 2024

Summary

The Earth is at a critical juncture due to anthropogenic greenhouse gas (GHG) emissions and ecosystem degradation, which may lead to irreversible shifts in its climate system. Crossing climate tipping points could trigger catastrophic ecosystem changes. *Tipping points* represent critical thresholds in complex systems where small changes can lead to significant and irreversible shifts. Feedback loops play a crucial role, where reinforcing loops amplify change, and dampening loops resist it. *Positive tipping points*, unlike climate tipping points, are intentional and can be leveraged for transformative change. They involve deliberate actions to strengthen reinforcing feedback loops and weaken dampening ones, leading to rapid shifts toward desirable states, such as sustainability and decarbonisation.

The energy sector is one of the biggest contributors to GHG emissions. To reform it, *sustainable energy technologies* (SETs) are crucial. Sustainable energy technologies include, among others, renewable energy sources, energy storage, clean transportation, and smart grids. Positive tipping points in the uptake of SETs are identified by a marked acceleration in their rate of adoption. Triggering positive tipping points requires coordinated actions across technological, regulatory, economic, and social domains.

Geels and Ayoub (2023) analyse positive tipping points in the context of socio-technical transitions, which depend both on social and technological developments. The authors elaborate a feedback loop model comprising of seven feedback loops between four actor groups and the technology being deployed. The framework is tested by the authors through two case studies: the uptake of offshore wind and electric vehicles (EVs) in the United Kingdom (UK). Geels and Ayoub's model was used as a starting point for this thesis, which aimed to refine and expand it.

The main research question reads as "How can the feedback loop model of Geels and Ayoub be refined and expanded to understand how positive tipping points can be triggered in the adoption of sustainable energy technologies?". In this quest, the focus was on understanding which actors and interactions among these (referred to as actor roles) were the most relevant to trigger positive tipping points in the adoption of clean technologies. Special attention was given to the role of policymakers since several authors, including Geels and Ayoub, have acknowledged that policies play an essential role in enabling transitions.

As for the method used, first, the model of Geels and Ayoub was refined, before making any additions to it. Second, literature related to SETs and positive tipping points in general was examined to understand how the model could potentially be expanded. Third, four case studies where a positive tipping point had been reached were analysed, testing the hypotheses derived from the literature. The case studies were offshore wind energy in the UK, wind energy in Portugal, EVs in the UK, and EVs in the Netherlands. Fourth, the insights from the case studies were used to derive the final feedback loop model. Finally, all findings were validated through four semi-structured interviews with experts.

In terms of findings, both additional actors and actor roles were identified. On the one hand, Geels and Ayoub included policymakers, technology firms, adopters, and the wider public in their model. However, based on the literature initially analysed, academic and research institutes, trade associations, non-

governmental organisations, and financial institutions were also foreseen to have had a key role. All eight actor groups were identified in the wind energy case studies. However, financial institutions did not appear relevant in the EV case studies.

On the other hand, the interactions in the model of Geels and Ayoub can be aggregated into seven overarching actor roles. These are 1) technology development, 2) technology adoption, 3) legitimising technology, 4) providing financing incentives/options, 5) investing in infrastructure, 6) establishing legislation, and 7) lobbying. Based on the literature initially examined, six other actor roles were identified, them being 8) raising awareness, 9) fostering collaboration, 10) transferring knowledge, 11) giving and receiving feedback, 12) establishing market-based mechanisms, and 13) promoting education. In the wind energy case studies, all these actor roles appeared relevant except for actor role 13. Actor roles 12 and 13 were not found to be relevant in the EV case studies.

As for the role of policymakers, Geels and Ayoub only included economic and regulatory policy instruments in their framework. Economic instruments affect financial decisions, while regulatory instruments are tools that affect the addressees' behaviour. On the one hand, the literature analysed suggested that additional economic and regulatory instruments could be relevant. Indeed, market-based mechanisms were not included in Geels and Ayoub's model but played an essential role in the wind energy case studies. Similarly, strategic planning was an important regulatory instrument in all four case studies. On the other hand, it was also found that information instruments - which relate to fostering collaboration, knowledge exchange, and public awareness - can be important. While the two former were found in all case studies, policymakers did not appear to be involved in raising public awareness. The results thus suggest that policy mixes combining economic, regulatory, and information instruments are effective.

Due to the differences found in the case studies, two final feedback loop models were created. Figure 1 relates to the wind energy case studies, while Figure 2 was based on the EV case studies. The grey blocks and black labels were already included in Geels and Ayoub's model. The green and yellow blocks and labels correspond to the additions made. In green the actors and interactions that overlap between Figure 1 and Figure 2 are shown. Yellow is used to illustrate the additions that are specific of only one of the final feedback loop models.

The findings provide a conceptual framework to understand the actors and actor roles involved in enabling positive tipping points in the adoption of SETs. In particular, Figure 1 could be used as a reference for the deployment of business-to-business (B2B) technologies, typically adopted in large numbers by businesses. Figure 2 could be used as a baseline for the uptake of business-to-consumer (B2C) technologies, mainly adopted in small numbers by individual buyers.

This research addressed knowledge gaps regarding positive tipping points in the adoption of SETs. By enhancing, expanding and validating the feedback loop model proposed by Geels and Ayoub (2023), this MSc thesis contributed to the development of a pertinent theoretical framework in the field. Additionally, the four case studies provide empirical knowledge on the dynamics of socio-technical tipping points. The results of this study allow to better understand the actors and actor roles involved in triggering a positive tipping point in the adoption of B2B and B2C sustainable energy technologies. This was enabled by the use of feedback loop models, which allow to explore the interaction between the technologies and the socio-technical system. Furthermore, this thesis contributed to understanding how policies can effect technological and behavioural changes, as well as how policy mixes affect socio-technical transformation and vice-versa. Overall, this research's findings may provide insights on how to intentionally trigger future technological transitions.



Figure 1: Feedback loop model showing the relevant actors and respective roles in triggering a tipping point in the adoption of wind energy. This model might be used as a reference for the uptake of other B2B sustainable energy technologies

The research has several limitations, including a limited number of case studies focusing on two technologies and European countries, which limits the generalisability of the findings. More case studies could provide a broader empirical understanding. Additionally, some actor groups currently encompass a wide range of actors, and could be expanded further. The research also focused on the tipping point enablers and less on its barriers, which are typically more prominent at an initial stage. Future studies could analyse more years prior to the tipping point, and dampening feedback loops could be included in future models. The latter could be facilitated by using causal loop diagrams (CLDs), which also allow to more easily incorporate feedback loops involving multiple actors and create a numerical model of the system. Furthermore, the operationalisation of positive tipping points remains challenging, and future studies could define metrics that reliably indicate a tipping point and analyse whether multiple positive tipping points exist. Quantitative methods could help understand the relative importance of different interactions. Besides, the sequence of events could be compared between case studies through temporal dynamics analysis. Finally, future research could study the initiation of feedback loops.



Figure 2: Feedback loop model showing the relevant actors and respective roles in triggering a tipping point in the adoption of electric vehicles. This model might be used as a reference for the uptake of other B2C sustainable energy technologies

Despite this study's limitations and avenues for future research, it holds political and practical significance. In terms of its political relevance, this thesis emphasises the importance of integrating knowledge from socio-technical transitions and political science, rather than examining policies in isolation. The study highlights the value of using policy mixes and understanding the roles and interventions of policymakers in successful case studies. This helps government agencies design and implement effective policies, especially in wind energy and EVs. Concerning its practical applicability, industry players and investors can benefit from understanding the policy landscape and its impact on adoption, contributing to strategic decision-making. The research underscores the importance of considering the interactions among the actors in the system. Analysing interactions between actors and identifying which roles are not being successfully fulfilled can ensure all relevant actor roles are covered, potentially leading to a positive tipping point in the technology's adoption.

This research provides insights on how to deliberately trigger positive tipping points in the uptake of sustainable energy technologies. The possibility of doing so gives hope for realising a successful energy transition and securing a sustainable future for future generations.

Contents

Pr	eface	•	i			
Sι	umma	ımary ii				
No	omen	clature	ix			
1	Intro	oduction	1			
2	Вас	kground	2			
	2.1	Climate Crisis & Energy Transition	2			
	2.2	Tipping Points	3			
	2.3	Climate Tipping Points	3			
	2.4	Positive Tipping Points	3			
	2.5	Socio-Technical Tipping Points	5			
3	Res	earch Design	7			
	3.1	Literature Review	7			
	3.2	Knowledge Gap	12			
	3.3	Research Aim	13			
	3.4	Research Questions	13			
	3.5	Significance of the Study	14			
4	Con	ceptual Framework	15			
	4.1	Geels & Ayoub's Framework	15			
		4.1.1 Multi-Level Perspective	15			
		4.1.2 Feedback Loop Model	16			
	4.2	Feedback Loop Model Refinement	19			
		4.2.1 Modifications to Blocks	20			

		4.2.2	Modifications to Arrows	20
	4.3	Feedba	ack Loop Model Expansion	22
		4.3.1	Actors & Actor Roles	22
		4.3.2	Initial Version of Expanded Feedback Loop Model	25
5	Cas	e Studie	es Methodology	26
	5.1	Case S	Studies Selection and Analysis	26
		5.1.1	Case Studies Selection	26
		5.1.2	Case Studies Analysis	27
		5.1.3	Case Studies Comparison	28
	5.2	Data C	ollection	29
		5.2.1	Data Collection Methods	29
		5.2.2	Collecting Relevant Data	31
	5.3	Validati	ion	32
6	Cas	e Studie	es	33
	6.1	Wind E	nergy in the UK (2002 - 2014)	33
	~ ~			•••
	6.2	Wind E	nergy in Portugal (2000 - 2007)	42
	6.2 6.3			
	6.3	Electric	nergy in Portugal (2000 - 2007)	42
7	6.3 6.4	Electric	Inergy in Portugal (2000 - 2007) Vehicles in the UK (2015 - 2021) Vehicles in the Netherlands (2015 - 2021)	42 50
7	6.3 6.4	Electric Electric cussion	Inergy in Portugal (2000 - 2007) Vehicles in the UK (2015 - 2021) Vehicles in the Netherlands (2015 - 2021)	42 50 59
7	6.3 6.4 Disc	Electric Electric cussion Model	Inergy in Portugal (2000 - 2007) Vehicles in the UK (2015 - 2021) Vehicles in the Netherlands (2015 - 2021)	42 50 59 68
7	6.3 6.4 Disc	Electric Electric cussion Model 7.1.1	Energy in Portugal (2000 - 2007) Inergy in Portugal (2000 - 2007) Inergy in Portugal (2015 - 2021) Inerg	42 50 59 68 68
7	6.3 6.4 Disc	Electric Electric cussion Model 7.1.1 7.1.2	Inergy in Portugal (2000 - 2007) Ic Vehicles in the UK (2015 - 2021) Ic Vehicles in the Netherlands (2015 - 2021) Ic Vehicles in the Netherlands (2015 - 2021) Ic Vehicles in the Netherlands (2015 - 2021) Ic Final Feedback Loop Models	42 50 59 68 68 68
7	6.3 6.4 Disc	Electric Electric cussion Model 7.1.1 7.1.2 7.1.3	Inergy in Portugal (2000 - 2007) Vehicles in the UK (2015 - 2021) Vehicles in the Netherlands (2015 - 2021) Expansion Final Feedback Loop Models Actors	42 50 59 68 68 68 71
7	6.3 6.4 Disc	Electric Electric Cussion Model 7.1.1 7.1.2 7.1.3 7.1.4	inergy in Portugal (2000 - 2007) c Vehicles in the UK (2015 - 2021) c Vehicles in the Netherlands (2015 - 2021) Expansion Final Feedback Loop Models Actors Actor Roles	42 50 59 68 68 68 71 72

		7.2.2 Final Feedb	ack Loop Models Validation		83
	7.3	General Remarks .			84
8	Con	clusions & Recom	mendations		85
	8.1	Answering the Res	earch Questions		85
	8.2	Scientific Contribut	ion		86
	8.3	Limitations & Reco	mmendations		88
		8.3.1 Research S	соре		88
		8.3.2 Actors & Ac	tor Roles		88
		8.3.3 Feedback L	oop Models		89
		8.3.4 Application	of Other Frameworks		90
		8.3.5 Operational	isation of Positive Tipping Points		90
		8.3.6 Importance	and Sequence of Interactions		91
	8.4	Policy Relevance &	Practical Implications		91
	••••			• • •	51
Bi		raphy			93
		raphy			
A	bliog Data	raphy			93
A B	bliog Data Surr	raphy			93 110
A B	bliog Data Sum Sum	raphy a nmary of Actor Role nmary of Policies			93 110 111 115
A B	Data Data Sum Sum C.1	raphy mary of Actor Role mary of Policies Policy Categorisation	es		93 110 111 115
A B	Data Data Sum C.1 C.2	raphy mary of Actor Role mary of Policies Policy Categorisation	es on		93 110 111 115 115
A B C	Data Data Sum C.1 C.2	raphy mary of Actor Role mary of Policies Policy Categorisation Case Studies Polic	es on		 93 110 111 115 115 118
A B C	Data Sum C.1 C.2 Sum	raphy mary of Actor Role mary of Policies Policy Categorisation Case Studies Polic mary of Validation	es on		 93 110 111 115 115 118 122
A B C	Data Sum C.1 C.2 Sum	raphy mary of Actor Role mary of Policies Policy Categorisation Case Studies Polic mary of Validation Interviewee 1 D.1.1 Wind Energ	es on	· · · · · · · ·	93 110 111 115 115 118 122 122
A B C	Data Sum C.1 C.2 Sum D.1	raphy mary of Actor Role mary of Policies Policy Categorisation Case Studies Polic mary of Validation Interviewee 1 D.1.1 Wind Energ D.1.2 EVs in the U	es on	· · · · · · · ·	 93 110 111 115 118 122 122 122
A B C	Data Sum C.1 C.2 Sum D.1	raphy mary of Actor Role mary of Policies Policy Categorisation Case Studies Polic mary of Validation Interviewee 1 D.1.1 Wind Energ D.1.2 EVs in the U Interviewee 2 - Win	es on	· · · · · · · · · · · ·	 93 110 111 115 115 118 122 122 122 123 124

Nomenclature

Abbreviations

Abbreviation	Definition
B2B	Business-to-business
B2C	Business-to-consumer
BEV	Battery electric vehicle
CLD	Causal loop diagram
EU	European Union
EV	Electric vehicle
GDP	Gross domestic product
GHG	Greenhouse gas
ICE	Internal combustion engine
ICEV	Internal combustion engine vehicle
IEA	Internation Energy Agency
LEZ	Low-emission zone
NGO	Non-governmental organisation
PHEV	Plug-in hybrid electric vehicle
R&D	Research and development
RD&D	Research, development, and demonstration
SET	Sustainable energy technology
SMEs	Small and medium enterprises
TSO	Transmission System Operator
UK	United Kingdom
WWF	World Wildlife Fund

Specific to UK case studies (Section 6.1 and Section 6.3)

CfD	Contracts for Difference	
CE	Crown Estate	
DECC	Department of Energy & Climate Change	
ETI	Energy Technologies Institute	
EU ETS	European Emissions Trading System	
FED	Fuel Excise Duty	
FiDeR	Final Investment Decision Enabling for Renewables	
FoE	Friends of the Earth	
NAREC	National Renewable Energy Centre	
RO	Renewable Obligation	
ROCs	Renewable Obligation Certificates	
RSPB	Royal Society for the Protection of Birds	
SMMT	Society of Motor Manufacturers and Traders	

Continued on next page

Abbreviation	Definition		
TSB	Technology Strategy Board		
ULEV	Ultra Low Emission Vehicle		
ULEZ	Ultra Low Emission Zone		
VED	Vehicle Excise Duty		
Specific to wind	in Portugal case study (Section 6.2)		
APREN	Portuguese Association for Renewable Energies (Associação Portuguesa de Energias Renováveis)		
FEUP	Engineering Faculty of the Porto University (Faculdade de Engenharia da Universidade do Porto)		
INESC	Mechanics Engineering and Industrial Management Institute (Instituto de Engenharia de Sistemas Computadores)		
INETI	National Institute of Engineering, Technology and Innovationm (<i>Instituto Nacional de Engenharia, Tec</i> nologia e Inovação)		
ISEP	Porto Higher Engineering Institute (Instituto Superior de Engenharia do Porto)		
IST	Higher Technical Institute (Instituto Superior Técnico)		
Specific to EV in	n the Netherlands case study (Section 6.4)		
ANWB	General Dutch Cyclists' Union (Algemene Nederlandse Wielrijdersbond)		
BOVAG	Association of Automobile Dealers and Garage Owners (BOnd Van Automobielhandelaren en Garage houders)		
EVO	Own Carrier Organization (Eigen Vervoerders Organisatie)		
HBO	University (Hogeschool)		
HBO MBO	University (<i>Hogeschool</i>) Secondary vocational education (<i>Middelbaar beroepsonderwijs</i>)		
МВО	Secondary vocational education (Middelbaar beroepsonderwijs) National Charging Infrastructure Knowledge Platform Foundation (Nationaal Kennisplatform Laadii		
MBO NKL	Secondary vocational education (<i>Middelbaar beroepsonderwijs</i>) National Charging Infrastructure Knowledge Platform Foundation (<i>Nationaal Kennisplatform Laadin frastructuur</i>)		
MBO NKL RAI	Secondary vocational education (<i>Middelbaar beroepsonderwijs</i>) National Charging Infrastructure Knowledge Platform Foundation (<i>Nationaal Kennisplatform Laadin frastructuur</i>) Bicycle and Automotive Industry (<i>Rijwiel en Automobiel Industrie</i>)		
MBO NKL RAI TLN	Secondary vocational education (<i>Middelbaar beroepsonderwijs</i>) National Charging Infrastructure Knowledge Platform Foundation (<i>Nationaal Kennisplatform Laadin frastructuur</i>) Bicycle and Automotive Industry (<i>Rijwiel en Automobiel Industrie</i>) Transport and Logistics Netherlands (<i>Transport en Logistiek Nederland</i>)		
MBO NKL RAI TLN TU	Secondary vocational education (<i>Middelbaar beroepsonderwijs</i>) National Charging Infrastructure Knowledge Platform Foundation (<i>Nationaal Kennisplatform Laadin frastructuur</i>) Bicycle and Automotive Industry (<i>Rijwiel en Automobiel Industrie</i>) Transport and Logistics Netherlands (<i>Transport en Logistiek Nederland</i>) University of Technology (<i>Technische Universiteit</i>)		

Table 1 – continued from previous page

Key Definitions

Term	Definition
Climate Tipping Points	Crucial thresholds that when crossed trigger rapid and potentially irreversible changes to a stable state in the Earth's climate system (Lenton et al., 2008)
Positive Tipping Points	Intentional interventions causing large-scale systemic changes through small actions, leading to transformative shifts toward low-carbon systems (Lenton et al., 2022)
Feedback Loops	When a change in one parameter leads to subsequent changes, which amplify (positive feedback) or oppose (negative feedback) the initial change (Geels & Ayoub, 2023)
Sustainable Energy Technologies	Technologies designed to reduce environmental impact, particularly in the context of energy production and consumption (Shahbaz, Siddiqui, Siddiqui, Jiao, & Kautish, 2023)

1 Introduction

As evidenced by a plethora of scientific literature, there is no doubt of the effects of anthropogenic climate change. In 2023, six of the nine planetary boundaries - important metrics for climate-related thresholds - had been crossed (Richardson et al., 2023). This increases the risk of crossing climate tipping points, defined as critical thresholds in the Earth climate system that when crossed can lead to irreversible changes to ecosystems (Lenton et al., 2008). To avoid such a scenario, current research has focused on triggering positive tipping points to convert high-carbon systems into low-carbon ones (Lenton et al., 2022), for which sustainable energy technologies are essential (Shahbaz et al., 2023).

Positive tipping points are deliberate and depend on reinforcing feedback loops to trigger swift changes. Tipping points in the adoption of sustainable energy technologies occur in a socio-technical context, involving both technological and societal factors. Due to the complexity involved, research on how to trigger socio-technical tipping points is still limited, even though the interest is rapidly growing (Lenton et al., 2022, 2023). With their model Geels and Ayoub (2023) took a step towards understanding which reinforcing feedback loops between actors were involved in the deployment of a new technology.

This thesis aimed to *understand how the model developed by Geels and Ayoub (2023) could be refined and expanded to understand how positive tipping points can be triggered in the adoption of sustainable energy technologies.* To do so, it was investigated which other actors and actor roles were relevant in the uptake of sustainable energy technologies. The findings were supported by four case studies: wind energy in the United Kingdom (UK) and in Portugal, and electric vehicles in the UK and the Netherlands.

This report is structured in the following manner. Firstly, in Chapter 2 the necessary background is provided. Secondly, the research design is covered in Chapter 3, which touches upon aspects such as the literature review conducted and the knowledge gap found, the research aim and questions, and the significance of this study. In Chapter 4 the conceptual framework is described. Then, Chapter 5 describes the methodology utilised to analyse the case studies. This chapter should provide the reader with the tools for replicating the work performed in this research, if desired. This is followed by the four case studies, in Chapter 6. A subsequent discussion of the results is then provided in Chapter 7. Here, the main insights and generalisations are presented, as well as the final feedback loop models. To close, Chapter 8 provides the answers to the research questions, clarifies the scientific contribution, policy relevance, and practical insights of this thesis, and highlights limitations and recommendations.

2 | Background

In this chapter, the necessary background about key concepts is provided. Section 2.1 puts the problem of climate change into perspective, highlighting the need for an energy transition. In this context, climate and positive tipping points have been the focus of much research. Section 2.2 therefore explains the concept of tipping points. Then, Section 2.3 clarifies what are climate tipping points, while Section 2.4 does so for positive tipping points. At last, Section 2.5 delves into socio-technical tipping points, relevant to promote energy-related shifts.

2.1. Climate Crisis & Energy Transition

Human activities, particularly greenhouse gas (GHG) emissions are to blame for global warming. The energy sector is one of the biggest contributors, having accounted for 34% of net global GHG emissions in 2019. Another 21% came from transportation and buildings (IPCC, 2023a), which cannot be sustainable if the energy sector itself is not. There is therefore an increasing effort to realise a transition towards sustainable energy technologies.

Sustainable energy technologies (SETs) are the focus of this Master's thesis. These refer to adapted or innovative environmentally responsible solutions, products, and systems designed to meet current energy needs while minimising negative environmental impacts, reducing GHG emissions and promoting the sustainable use of natural resources. Therefore, SETs relate to producing, gathering, converting, storing, transporting, and operating electrical power. Consequently, they encompass not only renewable energy sources but also technologies related to energy storage, carbon capture, and clean transportation. In addition, smart grid systems and sustainable building technologies can also be considered SETs (Bossink, Blankesteijn, & Hasanefendic, 2023; Shahbaz et al., 2023).

Although the trend towards sustainability has been accelerating (Geels, Sovacool, Schwanen, & Sorrell, 2017), out of the fifty sustainable technologies monitored by the IEA (2023c), only three - solar photovoltaics, electric vehicles, and lighting - are considered to be fully on track with the net zero by 2050 scenario. Upon the slow progress of SETs, global GHG emissions have kept rising. As a consequence, extreme weather and climate events have been negatively affecting Earth's atmosphere, ocean, cryosphere, and biosphere, as well as people's health and safety (IPCC, 2023b).

If decarbonisation does not occur at a higher pace, there is the risk of triggering several climate tipping points, in which case life on Earth might become very different from what is known (Lenton et al., 2008; Roberts et al., 2018). Triggering instead positive tipping points can have the power to prevent such a catastrophic future (Fesenfeld, Schmid, Finger, Mathys, & Schmidt, 2022). Before delving into these concepts, Section 2.2 explains the characteristics and dynamics of tipping points in a broader sense.

2.2. Tipping Points

Tipping points act as crucial thresholds in complex systems, where even a small change in one parameter can have non-linear consequences and trigger a significant, abrupt, and potentially irreversible change to a stable state (Geels & Ayoub, 2023). Irreversibility is tightly linked to hysteresis, meaning that returning to the initial stable state involves more effort than the one needed to cross the tipping point. For example, should the Greenland Ice Sheet completely melt due to global warming, it would not be enough to stop emissions and have global warming under control for the ice sheet to form again. Instead, global cooling would be necessary. This assumes, however, that it is still possible (even if hard) to revert back to the initial state, which might not be true (Lenton et al., 2022, 2023).

Underlying the dynamics of tipping points there are feedback loops. In positive or reinforcing feedback loops an initial change is amplified, leading to either vicious or virtuous cycles. Conversely, negative or dampening feedback loops are characteristic of a stable system, where an initial change is opposed¹ (Sterman, 2000). The latter often pose as the barriers for crossing a tipping point and the former as the enablers. Indeed, a tipping point is crossed because reinforcing feedback loops overshadow dampening feedbacks, making the process self-perpetuating. Once in the new stable state, dampening feedback loops contribute to an (almost) irreversible process (Lenton et al., 2022).

2.3. Climate Tipping Points

A specific type of tipping points are climate tipping points, which hold a negative connotation. Climate tipping points are pivotal thresholds within the Earth's climate system. Crossing them can result in substantial and irreversible shifts in the planet's climate and ecosystems and in life as is known. Additionally, triggering one tipping point can potentially cause another to activate, and so on, due to the possibility of upward scaling tipping cascades (Lenton et al., 2022, 2023).

For instance, the tipping of the Greenland Ice Sheet would lead to a substantially different Earth climate system, with higher temperatures due to a lower Earth albedo, as well as major biodiversity impacts (Lenton et al., 2008)². At the same time, triggering this tipping point could lead to a critical transition in one of the major components responsible for regulating Earth's temperature and supporting the marine ecosystems. These disruptions together cause sea-level rise and Southern Ocean heat accumulation, which could in turn accelerate ice loss from the west Antarctic Ice Sheet (Steffen et al., 2018).

Lenton et al. (2008) pioneered the research on tipping points by identifying policy-relevant climate tipping points at the risk of being triggered this century. According to Lenton et al. (2023), five climate tipping points are already at the imminent danger of being crossed.

2.4. Positive Tipping Points

To mitigate the possibility of crossing climate tipping points, the tipping point dynamics can be used in favour of society (Lenton et al., 2023). Enabling *positive* tipping points offers a glimmer of hope (Fesenfeld et al., 2022) as it might be the only way to avoid the worst consequences of climate change (Alkemade & de Coninck, 2021) or, in other words, propel climate action (Winkelmann et al., 2022).

¹'Positive' and 'negative' are not meant to transmit a sense of good or a bad. Rather, this is the commonly used terminology for reinforcing/amplifying and dampening/balancing feedback loops, respectively (Sterman, 2000).

²For other examples of climate tipping points, consult Lenton et al. (2008) or Lenton et al. (2023) (page 21).

Positive tipping points relate to causing large-scale systemic changes through small and smart interventions aiming at converting high-carbon into low-carbon systems (Lenton et al., 2022). Crossing a positive tipping point can also have a cascading effect. For example, a tipping point electric vehicles' adoption might be associated with a reduction in battery costs. This would allow to upscale storage capacity, facilitating a tipping point in green ammonia production (Lenton et al., 2023).

Positive tipping points have all the characteristics explained under Section 2.2. Nevertheless, unlike climate tipping points, positive tipping points are intentional, requiring deliberate forcing to strengthen reinforcing feedback loops and weaken dampening ones (Roberts et al., 2018). By understanding and triggering positive tipping points, there is the potential to induce rapid and transformative shifts towards desirable states that promote global decarbonisation and sustainability (Fesenfeld et al., 2022).

Geels and Ayoub (2023) consider that a positive tipping point occurs when innovation dynamics change from a fragile state to a self-sustaining one, i.e. when the technology has reached a stable design and its adoption picks up momentum³. Moreover, Lenton et al. (2022) identify a positive tipping point as going up the typical S-curve of adoption. These definitions are in agreement with the one provided in Lenton et al. (2023) and illustrated in Figure 2.1. According to the authors, after a tipping point is crossed an accelerating phase, characterised by non-linear changes and reinforcing feedbacks, is entered, after which the system stabilises at a more sustainable state.



Figure 2.1: Concept of a positive tipping point, with the system (circle) moving through three different phases (Lenton et al., 2023)

³Explained in more detail under Section 4.1.1.

Figure 2.1 shows the system evolving through three phases: enabling, accelerating, and stabilising. During the *enabling phase*, initially dominated by dampening feedback loops, agents can strategically intervene to trigger a positive tipping point. This is done by creating enabling conditions to weaken dampening feedbacks and enhance reinforcing ones. The *acceleration phase* is entered once the tipping point is crossed, being characterised by non-linear and self-perpetuating changes driven by strong reinforcing feedbacks. This is followed by the *stabilising phase*, during which the system stabilises at a qualitatively different state. If a positive tipping point is successfully triggered, the end result is a more sustainable, safe, and just world. Nevertheless, as demonstrated in Figure 2.1, this attempt might not be well succeeded, resulting in shallower outcomes or, in the worst case, unintended consequences.

This thesis focused, for one, on the enabling phase to understand the interventions that enable reinforcing feedback loops to dominate and a positive tipping point to be crossed. For another, it studied how these attempts resulted in a successful tipping point by analysing the beginning of the acceleration phase. Identifying the interventions, conditions, and mechanisms that lead to a positive tipping point in a particular context is especially relevant to understand how to foster such a self-perpetuating change.

2.5. Socio-Technical Tipping Points

Positive tipping points can be observed in a diverse set of systems, with energy transitions developing in socio-technical systems. These are characterised by complex environments involving socio-economic and techno-economic developments as well as actor reorientations. Social factors include changes to cultural norms, values, public perception and acceptance, while technological aspects refer to innovation, R&D, infrastructure expansion, and interoperability. Economic factors involve the market dynamics, the technology's affordability, funding and investment. The political domain often plays an important role through the policies, incentives, and governance in place. Thus, to understand how to trigger tipping points in the adoption of SETs it is vital to take a systems-thinking approach, understanding the actor dynamics involved (Alkemade & de Coninck, 2021; Geels & Ayoub, 2023; Geels et al., 2017; Lenton et al., 2022, 2023; Sovacool, 2016; Stadelmann-Steffen, Eder, Harring, Spilker, & Katsanidou, 2021).

Specific interventions can promote socio-technical tipping points, such as technological and social innovations, behavioural nudges, spreading public information, private and public investments, and policy intervention (Brescia, 2019; Kekäle & Helo, 2014; Lenton et al., 2022; Newell, Twena, & Daley, 2021). Concerning the latter, efficient economic, regulatory, and information instruments can create synergies between technological and behavioural changes, leading to virtuous feedback loops (Fesenfeld et al., 2022; Roberts et al., 2018; Rogge & Reichardt, 2016). Some of these are shown in Table 2.1.

	Primary purpose			
Primary type	Technology push	Demand pull	Systemic	
Economic in- struments	RD&D grants and loans, tax incentives, state equity assistance	Subsidies, feed-in tariffs, trading systems, taxes, levies	Tax and subsidy reforms, infras- tructure provision, cooperative RD&D grants	
Regulation in- struments	Patent law, intellectual prop- erty rights	Technology/performance stan- dards, prohibition of products/prac- tices, application constraints	Market design, grid access guaran- tee, priority feed-in, environmental liability law	
Information in- struments	Professional training and qualification, entrepreneurship training, scientific workshops	Training on new technologies, rat- ing and labelling programs, public information campaigns	Education system, thematic meet- ings, public debates, cooperatives RD&D programs, clusters	

Table 2.1: Examples of economic, regulatory and information policy instruments (Rogge & Reichardt, 2016)

Interventions such as the ones described above can create the necessary enabling conditions to trigger strong, self-perpetuating reinforcing feedback loops and, subsequently, positive tipping points. Enabling conditions might involve a technology's performance, price, symbolism and accessibility. At the same time, they might also relate to the population size, the tightness of the social network, the information available, and the technology's complementarity with other products. In the energy and transportation sectors, the technology's performance, cost, and associated infrastructure were found to have the largest impact. Furthermore, strategic interventions might strengthen reinforcing feedback loops. Economies of sale, learning-by-doing, social contagion, and technological reinforcement were found to be the most relevant reinforcing feedbacks in the energy and transportation sectors (Kekäle & Helo, 2014; Lenton et al., 2022, 2023; Otto et al., 2020; Sharpe & Lenton, 2021).

This chapter provided background on the topic of sustainable energy transitions and tipping points. Climate tipping points, positive tipping points, and socio-technical tipping points were covered. Comprehending socio-technical tipping points is a pressing necessity to support governance and decisionmaking processes, aimed at minimising climate damage and fostering transitions towards sustainability (Lenton et al., 2023). Chapter 3 explains how this Master's thesis aimed at contributing to this goal.

3 | Research Design

This chapter outlines the research design. Section 3.1 presents a literature review, while Section 3.2 covers the identified knowledge gaps. The research aim is explained in Section 3.3, followed by the research questions in Section 3.4. At last, the significance of the study is detailed in Section 3.5.

3.1. Literature Review

There is plenty of research on enabling either technological (Mercure et al., 2021; Rogge & Reichardt, 2016; Sharpe & Lenton, 2021) or social (Eder & Stadelmann-Steffen, 2023; Newell et al., 2021; Otto et al., 2020; Stadelmann-Steffen et al., 2021; Winkelmann et al., 2022) tipping points. Nevertheless, studies that integrate these two perspectives and look at socio-technical tipping points are still scarce (Geels & Ayoub, 2023; Lenton et al., 2022). An overview of important work on the field is given below.

"Sociotechnical transitions for deep decarbonisation"

Geels et al. (2017) addressed the complexity of deep decarbonisation by developing a socio-technical framework and demonstrating how co-evolutionary interactions between technology and socio-cultural groupings might accelerate low-carbon transitions. First, they point to aligning innovations and systems so that several innovations are linked. Second, they highlight the relevance of building societal and business support for fast technological transitions. Third, they claim that the phase-out of polluting technologies and supply chains can remove barriers for niche innovations.

For future research, the authors advise exploring the interaction between innovations and socio-technical systems, emphasising that consumer acceptance, business models, and socio-political factors are often overlooked. They also highlight the need to align innovation policy with sector-specific policy in polycentric efforts. Lastly, they suggest combining model-based analysis with socio-technical research to develop cost-effective and socio-politically feasible policy approaches for deep decarbonisation.

"The politics of accelerating low-carbon transitions: towards a new research agenda"

Even though Roberts et al. (2018) do not develop any framework on the field, they examine the politics of deliberate acceleration by combining insights from political science, policy analysis, and sociotechnical transition studies. Their research sets the stage for prospective investigations within this domain. For one, they stress the importance of increased collaboration between socio-technical transition and political science scholars for a deeper understanding of acceleration policies.

Identified areas for further research include the sustainability of positive feedback loops generated by transition policies, crafting policies that are both popular and self-expanding, distinguishing between policy feedback effects and mechanisms, and exploring the deliberate cultivation of reinforcing feedbacks by policymakers. Additionally, they highlight research opportunities in understanding the dynamics of power, agency, and politics, as well as considering the broader context of low-carbon transitions.

"The co-evolution of policy mixes and socio-technical systems: Towards a conceptual framework of policy mix feedback in sustainability transitions"

Edmondson, Kern, and Rogge (2019) advocate for policy mixes in socio-technical transformations and feedback mechanisms, arguing that individual policy instruments are not adequate in this context. Therefore, they develop a framework aiming to clarify how policy mixes affect socio-technical transformation (via resource, interpretive, and institutional effects) and how modifications to the socio-technical system in turn influence the evolution of the policy mix (through socio-political, administrative, and fiscal feedbacks). At the same time, their framework, illustrated in Figure 3.1, also includes the influence of exogenous factors. Their concept emphasises the need to overcome obstacles that could eventually erode political support while establishing incentives for beneficiaries to mobilise support.



Figure 3.1: Interactions between policy mixes and socio-technical systems by Edmondson et al. (2019)

Even though Edmondson et al. (2019), Geels et al. (2017), and Roberts et al. (2018) focus on sustainable transitions in socio-technical systems and their insights can be used to deliberately trigger tipping points, they do not revolve around positive tipping points. In contrast, the studies presented below do so, offering clear insights into the dynamics of socio-technical tipping points.

"Operationalising positive tipping points towards global sustainability"

Lenton et al. (2022) focus on social-technical-ecological systems, presenting enabling conditions for positive tipping, how these influence reinforcing feedbacks, and which actions can trigger positive tipping in the adoption of sustainable technologies and behaviours, as summarised in Figure 3.2. The authors identify the need for future research to focus on understanding (and possibly designing a guiding map on) how to identify and intentionally trigger positive tipping. The authors emphasise the importance of intervening in different places at the same time, for instance by considering coalitions of shared interests and how agents' interventions can initiate system-wide positive tipping points. Moreover, they advocate for integrating what already exists in inventive ways instead of coming up with new ideas.



Figure 3.2: Framework developed by Lenton et al. (2022) on triggering positive tipping points

"The politics of enabling tipping points for sustainable development"

Fesenfeld et al. (2022) argue that policy strategies are essential to leverage political feedback from both technological advancements and behavioural changes and thereby facilitate the creation and crossing of positive tipping points. The authors propose a framework, illustrated in Figure 3.3, that links research on feedback mechanisms to three sustainability principles: efficiency, sufficiency, and substitution. The article stress that there is limited research on how policies can effectively lead to behavioural and technological changes. Fesenfeld et al. (2022) emphasise the need to comprehend how policy feedbacks concerning technological and behavioural changes occur in different contexts, as well as how to overcome the political opposition against sufficiency and substitution practices.



Figure 3.3: Behavioural and technological policies on enabling sustainable development (Fesenfeld et al., 2022)

Lenton et al. (2022) and Fesenfeld et al. (2022) present relevant frameworks concerning socio-technical tipping points. However, they leave it to the reader to draw a mental map of the interactions between different social groups. By comparison, Geels and Ayoub (2023) presented a *visual* framework showing reinforcing feedback loops between actors with routines, capabilities, beliefs, and interests.

"A socio-technical transition perspective on positive tipping points in climate change mitigation: Analysing seven interacting feedback loops in offshore wind and electric vehicles acceleration"

Geels and Ayoub (2023) developed a framework in the context of socio-technological tipping points that highlights the co-evolutionary interactions between technological advancements and actor re-orientations. Drawing on political science, discourse theory, business studies, consumption theory, and innovation studies, the paper employs the Multi-Level Perspective (Geels, 2019) to identify seven feedback loops in tipping point dynamics. These are shown in Figure 3.4, which is explained in Section 4.1.2 as it was used as the foundation for the conceptual framework of this thesis.



Figure 3.4: Feedback loop model of Geels and Ayoub (2023)

With their work, the authors paved the way to analysing the interaction between innovations and sociotechnical systems, as invoked by Geels et al. (2017). They also contribute to understanding how to deliberately enable reinforcing feedbacks and positive tipping points using existing interventions as suggested by Lenton et al. (2022), and through policymaking as pointed out by Roberts et al. (2018). Moreover, their two case studies - offshore wind and electric vehicles in the United Kingdom (UK) reveal the practical significance of interacting feedback loops and demonstrate different sequences of tipping point dynamics between the two sectors. Therefore, their case studies contribute to understanding how feedbacks exist in different contexts, as recommended by Fesenfeld et al. (2022).

Despite the relevant value of the study conducted by Geels and Ayoub (2023), several points can be improved concerning their work. Starting with the consistency of the diagram shown in Figure 3.4, the first improvement point relates to the blocks. While some represent an actor (*policymakers*, *firms*, and *wider publics*), another one has a complete sentence (*more adoption by users*), while the centre block

represents a *new technology*. Apart from potential improvements in the labelling of the blocks, it would be clearer that different blocks represent distinct things if they had different shapes. For instance, the *new technology* block would benefit from having a distinct shape compared to the other blocks. Furthermore, the *firms* block currently represents a very wide group of actors as there are uncountable types of firms. Lastly, even though the case studies in Geels and Ayoub (2023) are performed for two SETs, the central block in their diagram reads *new technology*. It is, however, unclear whether their diagram and analysis can be applied generically to any new technology or solely to SETs.

The second improvement point related to the consistency of Figure 3.4 concerns the arrow labels. On the one hand, some of them include an interaction and the consequence of the same. Examples are *increasing visibility for wider publics leading to more debate* (feedback 5) and *stronger policy plans and accompanying narratives shape public debates* (feedback 7). On the other hand, some arrows have a sense of direction through the use of words as "higher", "more" or "increasing" while others do not. It can be argued that what is relevant is that the interaction occurred and not whether that factor increased or decreased compared to a past period of time. For instance, it is by itself relevant that policymakers release loans, capital grants, R&D subsidies and purchase subsidies. It is not necessarily preferable to have more of these incentives if the ones implemented are effective.

Additionally, improvements could be made to the analysis and descriptions of the case studies. To start with, the attainment of a positive tipping point involves the participation of an expanded set of actors compared to the ones in Figure 3.4. This is suggested by Geels and Ayoub (2023) themselves and supported by Geels and Turnheim (2022), where a description of the diffusion of offshore wind and electric vehicles in the UK is given. Perhaps Geels and Ayoub (2023) did not intend for their feedback loop model to become too complex and therefore focused only on four actor groups. Nevertheless, the result was a simplified model that does not represent the whole reality of the factors involved.

Furthermore, the model of Geels and Ayoub (2023) only includes one interaction (i.e. arrow) that is not part of a particular feedback loop. This is *higher purchase subsidies*, from the policymakers to the users. Nevertheless, such 'isolated' interactions might play a big role in the technology's deployment, at times being the ones responsible for triggering a reinforcing feedback loop in the first place.

Concerning the data collected and analysed, Geels and Ayoub (2023) lack a good criterion for which time frame and data is relevant, resulting in a broad scope. For instance, both case studies were analysed since the (early or late) 2000s until the present moment. Nevertheless, the positive tipping point occurred in 2009 for offshore wind energy and in 2019 for EVs, begging the question of whether it is relevant to analyse a similar time frame for both case studies. Just as what happened in the late 2000s might not have had a large impact in the EVs tipping point in 2019, the events that occurred after 2020 probably do not explain the 2009 tipping point for offshore wind energy. It is also unclear which factors were the most critical for the events in their two case studies to unfold the way they did. Particularly in this respect is the fact that Geels and Ayoub (2023) identify the relevance of policies and policymakers but do not, for instance, mention which policies were determinant in triggering the positive tipping point. The policy aspect could therefore be developed further in order to contribute to the knowledge gap identified by Fesenfeld et al. (2022) on understanding how specific policies promote technological and behavioural change through intentional reinforcing feedback loops.

The lack of good criteria to select relevant data and to identify the main contributing factors to trigger the positive tipping points was also revealed through pieces of information that did not seem to help create the feedback loop model or in reaching the conclusions taken. For instance, for the case study of EVs in the UK, the authors covered global EV developments by including China in the case study. However, the few information provided on the roll-out of EVs in China was quite broad and was only given for some actor groups during certain time periods. The role of this information in understanding the accelerated deployment of EVs in the UK is unclear. At the same time, Geels and Ayoub (2023) included a thorough description of EV-related plants, jobs, and supply chain firms in the UK after 2017. There is not a clear link between changes in these factors and the increased adoption rate after 2019. Besides, these considerations have not been added to their feedback loop model.

The data collected by Geels and Ayoub (2023) was written chronologically. Each piece of information was linked to the corresponding feedback loop by specifying its number between squared brackets. It is not always clear how each piece of information connects to that specific feedback loop. An example, among others, is when the authors state that purchase subsidies and positive public discourses contributed to a higher uptake of EVs after 2015, identifying feedback loops 4 and 5 as relevant. However, this information does not seem to directly relate to neither feedback loop. For one, providing purchase subsidies does not seem to belong to any particular feedback loop. For another, it is true that positive public discourses can lead to increased desirability for EVs, leading to more adoption. Nevertheless, in Figure 3.4, increased adoption is not a consequence of feedback 5 - only an increase in desirability.

Moreover, Geels and Ayoub (2023) utilised the electricity generated as a measure for deployment in offshore wind energy. However, this metric is contingent on weather conditions and may not accurately reflect actual roll-out figures. In comparison, the cumulative installed capacity would offer a more accurate reflection of the technology's adoption rate.

One final point that could be questioned about the analyses performed by Geels and Ayoub (2023) is that they classify significant shifts in actor attitudes as positive tipping points in that specific actor sphere (e.g., political, social, technological). However, defining when a tipping point occurs in firms, users, wider publics, or policymaker perspectives can be ambiguous as it is not as straightforward as basing the choice on quantitative deployment figures.

Having provided a sketch of the literature landscape on the topic of socio-technical transitions and tipping points, Section 3.2 identifies a series of knowledge gaps.

3.2. Knowledge Gap

Several knowledge gaps were identified from the literature review conducted, as enumerated below. These relate to socio-technical tipping points in general, and to the policies used in this context.

- 1. Lack of knowledge on how to deliberately trigger socio-technical tipping points
 - 1.1 Which actors are essential to trigger a socio-technical tipping point
 - 1.2 How actors promote a tipping point through interventions and interactions with other actors
 - 1.3 How do the socio-technical tipping point dynamics differ in different contexts
 - 1.4 How to deliberately cultivate reinforcing feedback loops
 - 1.5 How to identify a positive tipping point
 - 1.6 Performing socio-technical research with the help of model-based analysis
- 2. Lack of knowledge with respect to the role of policies in triggering socio-technical tipping points
 - 2.1 How do policies lead to behavioural and technological changes

- 2.2 How to overcome the political opposition against sufficiency and substitution practices
- 2.3 How sustainable are the positive feedbacks generated by transition policies

In this thesis, it was attempted to address knowledge gaps 1.1, 1.2, 1.3, 1.4 and 2.1. Knowledge gap 1.5 is also covered in this thesis, with the caveat that this is done in retrospect, i.e., looking at historical data. Another interesting research gap is understanding how to identify that a positive tipping point is about to the crossed or has just been crossed. However, this is not the goal of this thesis. This and the remaining knowledge gaps are left to be investigated in further studies. Having the relevant knowledge gaps in mind, Section 3.3 clarifies the goal of this Master's thesis.

3.3. Research Aim

Considering the knowledge gaps this thesis aimed to tackle (see Section 3.2), the main goal was to contribute to the lack of research on the dynamics of socio-technical tipping points and how to intentionally trigger them. In this quest, it is important to understand how the actor dynamics, the policies and the context play a role in promoting, triggering, and sustaining positive tipping points. It can be said that every framework presented in Section 3.1 lacks something, creating avenues for further research.

At the same time, the model of Geels and Ayoub (2023) in particular had potential to be refined and expanded (see Section 3.1). This research aimed to refine the authors' model by tackling inconsistencies in the labels of the blocks and arrows. In terms of expanding the model, this thesis aimed to include in it more actors and interactions among these, understand the role of policymakers, and contribute to expanding the knowledge base by conducting more case studies. An associated goal was to not make the model overly complex, which could compromise its readability. In Section 5.2.2 the criteria to decide which data was deemed relevant to collect and to add to the model is provided.

3.4. Research Questions

To achieve the goals of this research (see Section 3.3) a main research question and five sub-questions were formulated. The proposed research question was *"How can the feedback loop model of Geels and Ayoub (2023) be refined and expanded to understand how positive tipping points can be triggered in the adoption of sustainable energy technologies?"*. Answering this question represented one more step in the direction of understanding how to accelerate the global shift towards sustainability. To answer the main research question, the following five sub-questions were answered.

- 1. How can the feedback loop model of Geels and Ayoub (2023) be refined to improve its consistency and readability?
- 2. Apart from the actor groups selected by Geels and Ayoub (2023), which other actors are relevant in promoting positive tipping points in the deployment of sustainable energy technologies?
- 3. What are the main actor roles in promoting a positive tipping point in the deployment of sustainable energy technologies?
- 4. Apart from the policy types identified by Geels and Ayoub (2023), which other policies are relevant in promoting positive tipping points in the deployment of sustainable energy technologies?
- 5. How can the feedback loop model of Geels and Ayoub (2023) be expanded to include a wider set of actors and interactions between these while maintaining its readability?

The main final output of the thesis was, firstly, four case studies which contribute to building empirical repertoire on the topic. Two case studies concerning wind energy, and two case studies concerning electric vehicles were analysed. Secondly, the thesis concludes with two feedback loop models¹ similar to that in Figure 4.2 where more actors and feedback loops are included.

Chapter 5 explains the methodology used to answer the questions above and to select the case studies. Sub-question 1 is answered in Section 4.2. Sub-questions 2, 3, and 4 are covered in Section 7.1.2, Section 7.1.3, and Table 7.15, respectively. Finally, Section 7.1.1 answers sub-question 5.

3.5. Significance of the Study

This study is significant within the broader landscape of socio-technical tipping points, adding depth to the existing knowledge base. The commitment to improve and build upon the model of Geels and Ayoub (2023) fostered the evolution of a relevant theoretical framework within the field, at the same time contributing to validate it. A key contribution lied in taking one more step in the direction of a visual framework of how actors interact with each other and intervene in socio-technical systems.

The case study analysis performed explored how feedback loops and positive tipping points manifest in diverse contexts, as suggested in the literature. By understanding which actor groups were relevant and how the actor dynamics unfolded in four case studies, this investigation contributed to understanding how to deliberately promote and trigger positive tipping points in the adoption of SETs in distinct settings. In interpreting the results, special attention was given to the role of policymakers, due to their importance in creating the necessary enabling conditions (Edmondson et al., 2019; Fesenfeld et al., 2022; Geels & Ayoub, 2023; Geels et al., 2017; Lenton et al., 2022; Roberts et al., 2018).

Conducting four case studies enhanced the applicability and generalisability of the findings, contributing to a more nuanced understanding. The actors and actor roles in the wind energy case studies differed slightly from those in the electric vehicle cases. However, they were consistent between the two wind energy studies and between the two electric vehicle studies. This suggests that tipping point dynamics may be similar within business-to-business and business-to-consumer clean energy technologies.

The findings of this research are relevant both for the scientific community, as well as for policymakers and industry players seeking to navigate the complexities of socio-technical transitions by facilitating sustainable technology adoption and behavioural shifts. Therefore, this Master's thesis provides a valuable contribution to the ongoing discourse on socio-technical tipping points.

This chapter described the research aim, how it fits within the existing literature, and how it contributes to relevant knowledge gaps. Chapter 4 proceeds by explaining the conceptual framework used, which was based on the work of Geels and Ayoub (2023).

¹Two feedback loops are presented since key differences were found between the wind energy and the EV case studies.

4 | Conceptual Framework

This chapter presents the conceptual framework used in this investigation. Section 4.1 discusses the framework presented by Geels and Ayoub (2023), used as the basis for this thesis' conceptual framework. Following, Section 4.2 explains the refinements performed to this framework. The chapter finalises in Section 4.3 with possible expansions to this model concerning actors and actor roles.

4.1. Geels & Ayoub's Framework

The framework presented by Geels and Ayoub (2023) to study positive tipping points in socio-technical transitions was used as the basis for this thesis' conceptual framework. The author's proposed framework is rooted in the Multi-Level Perspective, initially presented in Geels (2002) and later on refined in Geels (2019). Section 4.1.1 starts by delving into the Multi-Level Perspective. Afterwards, Section 4.1.2 describes the conceptual framework itself, i.e. the feedback loop model of Geels and Ayoub (2023).

4.1.1. Multi-Level Perspective

The Multi-Level Perspective (MLP) built on Geels' foundational work is shown in Figure 4.1. It is called 'multi-level' because it considers transitions and innovations as occurring across multiple interconnected levels within a societal system, presented in the vertical axis of Figure 4.1. The lower one, the *niche innovation*, refers to an emerging, innovative, and radical technology. Then, the *sociotechnical system* corresponds to the regime with its established practices. At the top, the *landscape* represents the broader societal context in which events unfold. Then, when the innovation is developing, four phases can be distinguished, on the horizontal axis: 1) experimentation, 2) stabilisation, 3) diffusion and disruption, and 4) institutionalisation and anchoring. *Experimentation* (phase 1) is characterised by the emergence of radical niche-innovations in niches via small projects. In the subsequent *stabilisation* period (phase 2) the innovation gains a foothold in small market niches and the technical design rules stabilise. This is followed by *diffusion and disruption* (phase 3), where the innovation enters mass markets, competing with existing systems and established technologies. Lastly, *institutionalisation and anchoring* (phase 4) occur when the new system becomes anchored in regulatory programs, user habits, and professional standards (Geels & Ayoub, 2023).

Geels and Ayoub (2023) define a positive tipping point as occurring between phases 2 and 3 of the MLP, as highlighted in Figure 4.1. When crossing this threshold, innovation dynamics transition from requiring substantial protective and developmental efforts to becoming self-sustaining through strong reinforcing feedbacks and alignment of socio-technical elements. This means that, when looking at a graph of the technology adoption over time, a positive tipping point is identified when the rate of adoption substantially increases from one year to the next and this rate is (approximately) maintained in the subsequent years, being eventually followed by a stabilisation phase. This behaviour can be visualised in Figure 3 and Figure 12 of Geels and Ayoub (2023), where a tipping point can be identified in 2009 and 2019 for offshore wind and electric vehicles in the UK, respectively.

The similarities with Figure 2.1 are significant. It could be said that phases 2, 3 and 4 in Figure 4.1 correspond to the enabling, accelerating, and stabilising phases in Figure 2.1, respectively. This definition also agrees with how Lenton et al. (2022) and Fesenfeld et al. (2022) identify a positive tipping point. Therefore, in this investigation the same definition of a positive tipping point was used.



Figure 4.1: Multi-level perspective where the tipping point is identified by an ellipse (Geels & Ayoub, 2023)

4.1.2. Feedback Loop Model

The feedback loop model of Geels and Ayoub (2023), shown in Figure 4.2¹, was based on the MLP. It focuses on co-evolutionary feedback loops between a new technology and four actor groups. The framework integrates insights from wider social sciences, moving beyond traditional tipping point models. It focuses on actors with routines, capabilities, beliefs, and interests, emphasising reactive feedback loops across various dimensions (Geels & Ayoub, 2023). Below, a description of the model's blocks and feedback loops is given, alongside the actor roles that can be identified.

Explanation of Blocks

The article lacks an explanation of the blocks included in Figure 4.2. Therefore, definitions are provided below, corresponding to the ones used in the remaining of this report. These definitions are based on the information provided in Geels and Ayoub (2023) with the help of supporting sources.

¹This is the same figure as Figure 3.4. It was here added again to support the explanatory text provided in this subsection.



Figure 4.2: Feedback loop model of Geels and Ayoub (2023)

Geels and Ayoub (2023) focus on a *new technology* in the context of socio-technical tipping points, analysing two case studies concerning sustainable energy technologies. Representing the new technology separately allows to visualise perspectives about and interactions with (the idea of) the technology, irrespective of which firm is developing it. For instance, an actor might be against a certain firm, due to e.g., its principles, and yet support one or some of the technologies developed or sold by this firm. It is possible that by new technology the authors mean a clean(er) technology, despite not stating this explicitly. Since this thesis focused on promoting the energy transition, the new technology should be interpreted as a *sustainable energy technology* (defined in Section 2.1) from here onwards.

Since the framework proposed by Geels and Ayoub (2023) is based on the MLP, the new technology likely intends to represent a niche innovation trying to break into the socio-technical regime. The term 'niche innovation' is associated with radical innovations that are taking their first steps (Geels, 2019). Nevertheless, in this thesis the technology was not referred to as a niche nor radical innovation. This is because by now technologies such as electric vehicles or wind turbines are not radical innovations anymore and yet they are not part of some countries' socio-technical regime. Additionally, many sustainable energy technologies are not just introduced by new entrants since they are typically researched and developed by incumbents as well (Geels & Ayoub, 2023). In this thesis the focus was put on phases 2 and 3 of Figure 4.1, which considers a technology which has stabilised in a dominant design.

It can be deduced that the block *firms* is used to represent to *technology firms*, which design, develop, manufacture, and provide innovative technologies, products, and services (Geels & Turnheim, 2022). These firms are at the forefront of technological advancements. Within this category, two groups can be distinguished: incumbents and new entrants (Fang, Li, & Govindan, 2024; Geels & Ayoub, 2023).

Incumbent technology firms are well-established and often large companies (Karttunen et al., 2021) that have been operating for an extended period. Incumbents have a history of producing and distributing traditional technologies, typically having substantial resources, industry expertise, and infrastructure (Fang et al., 2024). An example of an incumbent which recently aligned their vision with a more sustainable future is Siemens (2023). After a long history in conventional power plants and transmission technologies, in 2004 they diversified to wind turbines and in 2020 Siemens Energy was founded.

New entrants are usually the ones beginning new technology cycles, having fresh perspectives, disruptive solutions, and emerging technologies (Geels, 2019; Karttunen et al., 2021). New entrants might be start-ups or relatively young and small companies (Fang et al., 2024) driven by having a positive environmental impact (Karttunen et al., 2021). Tesla (2023) is an example of a new entrant with a disruptive approach. It was not an established automaker with a history of internal combustion engine vehicles. Instead, it focused on electric mobility bringing fresh ideas and innovation to the market.

User engagement and preferences influence the adoption of SETs (Ekim, Mattsson, & Bernardo, 2023; Geels & Turnheim, 2022) and therefore their demand and returns profile (Geels & Ayoub, 2023). It can be concluded that *users* aim to represent the entities and individuals who buy and/or utilise SETs, being referred hereinafter by *adopters* to avoid confusion. While the term adopter and user can most of the times be used interchangeably when it comes to EVs, it is not as straightforward for wind energy. For instance, while energy or utility firms might adopt wind turbines for their own use, it might also be that private investors purchase and install a wind farm (therefore being viewed as adopters) but sell it to a final owner after the instalment, making the latter the actual user of the technology. Furthermore, confusion might also arise between the users of the technology versus the users of the electricity.

Adopters can be businesses (or "user firms" as described by Geels and Ayoub (2023)) or individual consumers. Some technologies are adopted by both business and individual consumers, such as solar panels and energy-efficient solutions for buildings (Ekim et al., 2023; Geels & Turnheim, 2022). However, others strongly depend on being adopted by either businesses (e.g., wind turbines) or individual buyers (e.g., EVs) (Geels & Ayoub, 2023). The former are commonly referred to as business-to-business (B2B) technologies, while the latter as business-to-consumer (B2C) technologies.

Wider publics refers to a broad and diverse set of stakeholders including the general public, communities, advocacy groups, and society at large, that impact and shape the adoption and advancement of SETs. More positive public debate and discourse about a new technology increase its legitimacy and desirability (Geels & Ayoub, 2023). This can lead to more adopters, helping accelerate technological diffusion. Conversely, negative public opinion can hinder the development and uptake of a technology, as exemplified by nuclear energy (Ming, Yingxin, Shaojie, Hui, & Chunxue, 2016). Hence, considering the wider public is crucial to incorporate behavioural changes in the model (Fesenfeld et al., 2022).

Finally, *policymakers* encompass "political parties, Parliament, courts, and lobby groups" (Geels & Ayoub, 2023) responsible for formulating, implementing, and overseeing policies, laws, regulations, and guidelines at various government levels, from local and regional authorities to national and international bodies (Sultana, Dwivedi, & Moktadir, 2023; Vivalt & Coville, 2023).

Explanation of Feedback Loops

Figure 4.2 portraits seven feedback loops among the five blocks explained above. Each interaction in the model illustrates the role of different actors in influencing technology adoption, development, and policy shaping. Feedback loops 1 and 2 represent the economically-oriented feedbacks, where the technology firms interact with the adopters. The remaining loops correspond to the socio-political feedback loops. Socio-political processes have special significance in early transition phases when economic feedbacks tend to be negative and the economic actors (technology firms and adopters) are reluctant to invest. Overcoming negative feedbacks, such as high costs and inertia, is a challenge in these early phases, often addressed by radical innovations emerging in niches with protective policy support (Geels & Ayoub, 2023). The seven feedback loops are briefly explained below².

- *Feedback 1* (*users and technology*): Increasing user adoption reduces technology costs, stimulating further adoption. This is also referred to as increasing returns to adoption;
- *Feedback 2* (*firms, technology, users*): Firms invest more in new technologies with increasing demand, improving technical performance, and lowering costs;
- Feedback 3 (firms and policymakers): Growing technology firms have greater political access and hence influence on shaping policies;
- *Feedback 4* (*policymakers and technology*): Policymakers shape technology development and deployment through various instruments;
- *Feedback 5* (*users and wider publics*): Increasing adoption enhances learning-by-using processes, improving familiarity and perceptions of new technologies;
- *Feedback 6* (*wider publics and technology*): Positive public debates enhance the cultural meanings and societal legitimacy of new technologies;
- *Feedback* 7 (*wider publics and policymakers*): Increasing public attention creates pressures on policymakers to introduce new or strengthen existing policies.

Actor Roles

It can be said that Figure 4.2 capture with their framework seven overarching actor roles, being them 1) technology development, 2) technology adoption, 3) legitimising technology, 4) providing financing incentives/options, 5) investing in infrastructure, 6) establishing legislation, and 7) lobbying.

The framework of Geels and Ayoub (2023) provides a comprehensive understanding of positive tipping points in socio-technical transitions, emphasising the interplay between actors, feedback loops, and contextual factors. Nevertheless, the model can benefit from improvements, as detailed in Section 3.1. The refinements applied to the original feedback loop model are presented in Section 4.2.

4.2. Feedback Loop Model Refinement

In refining the feedback loop model of Geels and Ayoub (2023), the blocks' labels as well as the labels of the arrows connecting two blocks were modified, following the arguments presented in Section 3.1. This is explained in Section 4.2.1 and Section 4.2.2, respectively. The refined version of the model is illustrated in Figure 4.3, which should be compared to Figure 4.2.

²For more information consult Geels and Ayoub (2023).



Figure 4.3: Refined version of the feedback loop model of Geels and Ayoub (2023)

4.2.1. Modifications to Blocks

Firstly, the block with *more adoption by users* was replaced by *adopters*. Secondly, the *new technology* block was changed into an ellipse so that it is clear that it does not correspond to an actor like the remaining blocks. Its label was also modified to *sustainable energy technology* since that is the focus of this thesis. This should not have an impact on the overall meaning of the diagram developed by Geels and Ayoub (2023). Rather, it helped in making the diagrams developed during this investigation more specific as the conclusions taken at the end of this research might not be applicable to all new technologies but solely to new *sustainable energy* technologies. Thirdly, the block labelled *firms* was changed into *technology firms*, following the reasoning given in Section 4.1.2.

Moreover, each block has now only its label and not a brief description of how the respective views, strategies and goals evolved over time due to the feedback loops. This reduced the complexity of the initial version of the model, which was especially relevant since the goal of this thesis was to add more actors and interactions to the model, inevitably increasing its overall complexity. The information conveyed by these labels was, nevertheless, not neglected and was incorporated in the explanatory text that accompanies the feedback loop model(s) presented in the remaining of this report.

4.2.2. Modifications to Arrows

Just like the block's labelling was thought to lack consistency, the labels of the arrows in Figure 4.2 could also benefit from improvements, as detailed in Section 3.1. For one, the consequence of an interaction was removed from the arrow's label such that the arrow label would only represent the interaction itself. For another, comparative adjectives such as "higher", "more" or "increasing" were removed.

Additionally, a system was created to build a sentence using: label block 1 + verb + *arrow label* + connector + label block 2. Block 1 is the one connected to the start of the arrow, while block 2 connects to the arrow's head. An example is: *policymakers* + give + *loans, capital grants and R&D subsidies* + to + *technology firms* (feedback 3). These points resulted in three rules for the labels of the arrows that were applied not only when modifying the arrows in the model of Geels and Ayoub (2023) but also when adding new arrows to the baseline model. The rules are summarised below.

- Rule 1: The arrow's label should only represent an interaction between two actors or an actor and the technology. The consequence of that interaction should not be included. *Example:* The label stronger policy plans and accompanying narratives shape public debates should not include shape public debates³;
- Rule 2: Arrows should not have a sense of direction through the use of words such as "higher", "more" or "increasing". *Example: higher purchase subsidies* should be solely *purchase subsidies*;
- Rule 3: It should be possible to formulate a sentence using the following elements: block 1 label
 + verb + arrow label + connector + block 2 label. Example: The sustainable energy technology + provides + value + to + the adopters.

Apart from performing adjustments to several arrow labels such that these complied with the three rules presented above, two feedback loops suffered bigger adjustments. In feedback loop 1 it is not possible to formulate a sentence using rule 3 for the label *increasing purchase and sales* since it would resemble the following: the technology provides increasing purchase and sales to the adopters. It is true, however, that feedback 1 is meant to represent increasing returns to adoption, defined as a reduction in the technology's costs and increase in its performance as adoption rises (Geels & Ayoub, 2023). To try to capture this idea while following the established rules, feedback loop 1 now reads as: the *SET* presents (a higher) *value*⁴ to the *adopters*, which complete (more) *purchases* of the *SET*.

Feedback loop 5 also suffered modifications. Presumably, Geels and Ayoub (2023) meant that more adoption by users increase the technology's for the wider public, leading to more public debate and desirability to buy the product, resulting in more adoption. First, the fact that there are more public debates is a *consequence* of the interaction between the adopters and the wider public and should not be part of the arrow label according to rule 1. Second, the higher desirability for the technology is at most an interaction between the wider publics and the technology, not the adopters. Thirdly, the resulting increased adoption is represented in the label of the users' block in Figure 4.2 but not in the arrow connecting the actors. For these reasons, the arrow going from the wider publics to the adopters was changed from *increasing desirability* to *further adoption*. It is thought that this does not change the meaning of feedback loop 5 while making it clearer. Lastly, the arrow label going from the adopters to the wider publics was also modified as to follow the rules established. Accordingly, now it reads as: *adopters* contribute to *expanded technology's visibility* for the *wider publics*.

Combining these labelling modifications, the model shown in Figure 4.3 is a refined version of the one in Figure 4.2. Nonetheless, based on evidence from literature it is expected that more actors and actor roles have a relevant contribution to the accelerated deployment of sustainable energy technologies. Section 4.3 covers potential expansions to the feedback loop model concerning these aspects.

³This specific arrow label also suffered other modifications in order to comply with rules 2 and 3.

⁴A higher value is associated with a better performance and a lower cost.

4.3. Feedback Loop Model Expansion

The model presented by Geels and Ayoub (2023) and explained in Section 4.1 has the advantage of being simple in the sense that it only includes four actor groups. However, this comes with the possible negative consequence of excluding important actors and respective functions. Section 4.3.1 provides an explanation for four actor groups, and associated actor roles, that other scientific sources deem relevant in socio-technical transitions. According to the information found about these actors, an initial version of an expanded feedback loop model is proposed in Section 4.3.2.

4.3.1. Actors & Actor Roles

In Section 4.1.2, the four actor groups included in the model of Geels and Ayoub (2023) are described. However, the authors themselves identified that their feedback loop model could benefit from the addition of more actor groups. Nevertheless, apart from financial firms, the authors did not provide other suggestions for additions. Apart from financial institutions, literature repeatedly refers to three other actor groups apart from the ones presented by Geels and Ayoub (2023). These are academic and research institutes, trade associations, and non-governmental organisations (NGOs). A description of these four actors and the functions they typically perform is provided below.

Academic & Research Institutes

Academic and research institutes function as innovation, learning, and knowledge hubs by researching and testing innovative and radical solutions. They include universities, research centres, technical schools, and academic departments. A range of partnerships exist between these institutes and technology firms, including student-centred partnerships focusing on workforce development (Becker & Brown, 2000) as well as research partnerships driven by innovation and sponsorship opportunities (Fraser et al., 2011). The translation of scientific advances into commercial products is facilitated by funding partnerships, where technology firms act as research sponsors (S. Chai & Shih, 2016). These partnerships are crucial for knowledge exchange and talent development.

NGOs

NGOs shape public debates and opinion via public awareness campaigns and supporting (and sometimes starting) grassroot movements. Moreover, NGOs actively advocate for policy changes - often in the name of the wider public - through direct and indirect lobbying (De Bruycker & Beyers, 2019) and their research activities. For instance, in the UK NGOs played a key role not only in advocating in favour or against certain technologies but also in drafting the 2008 Climate Change Act (Farmer et al., 2019; Geels & Turnheim, 2022). Furthermore, NGOs are involved in litigation and electioneering processes which directly affect policymaking and policymakers. Geels (2019) stresses that policymakers should aim for a more open and inclusive governance style by involving NGOs, which in turn leads to a more participatory democracy (Junk, 2016). The World Wildlife Fund (2023) is an example of an NGO that works on renewable energy initiatives, advocating for policies that promote clean energy adoption. They also partner with governments and businesses to support sustainable energy practices.

Financial Institutions

Public resources are limited and so financial institutions are needed to cover the large capital costs (Falchetta, Michoud, Hafner, & Rother, 2022) required by novel SETs. For instance, on the offshore wind case study of Geels and Ayoub (2023), it was found that although policymakers were able to drive firm strategies and technology deployment, it was essential that private investors were mobilised in order for costs to decrease and adoption to increase further. An increased access to equity and debt finance (like pension funds, insurance funds, sovereign wealth funds) led to lower risk profiles and reduced capital costs, which increased adoption. Nonetheless, the authors left this actor group out of their model. Other authors also identified private investment, such as bank loans (Xu, Kasimov, & Wang, 2022), as crucial for fostering universal access to electricity (Falchetta et al., 2022; Hinestroza-Olascuaga, Carvalho, & de Jesus, 2023), for developing the renewable energy industry, and for promoting the conservation of resources and of the environment (Xu et al., 2022).

Financial institutions facilitate the financing of SETs by providing financial services, investment, and funding They encompass a wide range of entities, including banks, investment firms, venture capital organisations, private equity firms, and public financing bodies, that contribute to the growth and advancement of SETs. For instance, the Green Investment Group (2023) is a specialist green investment entity known for its focus on investing in and financing renewable energy projects. They have committed or arranged more than £26 billion and contributed to more than 90 GW of green energy assets.

Trade Associations

Trade associations allow for better communication, knowledge exchange, and collaboration between businesses, professionals, and stakeholders, which constitutes an advantage for firms and consumer surplus. Furthermore, they represent and advocate for the collective interests of their members next to policymakers. These associations do not have a direct commercial interest and therefore they are better able to advocate, in the name of the individual companies, for policies, standards, and practices that support the development, deployment, and adoption of clean and renewable energy technologies. Besides, trade associations might help financial institutions assess industry trends, risks, and opportunities, engage in public awareness campaigns involving the wider public, and collaborate with NGOs on sustainability initiatives and policy advocacy (Boleat, 1996; Kirby, 1988).

Geels and Turnheim (2022) highlight the role of trade associations in low carbon transitions in the UK, namely concerning electricity consumption, offshore wind, building systems, and biomass heating. Different types of trade associations include industry, multi-industry, specialised, company, federal, national, and international associations. Besides, a specific example is the international trade association for wind power, the Global Wind Energy Council (2023). They conduct authoritative research and analysis on the wind power industry. Moreover, they work with governments and international institutions to provide them with transparent information about the advantages and potential of the technology. Besides, they foster collaboration between policymakers from various countries.

Policymakers

The literature did therefore suggest that the role of policymakers was more extensive than suggested by Geels and Ayoub (2023). In addition to what has been included in the descriptions above, policymakers can address resistance from incumbents by providing incentives to promote firm reorientation, which

might include phase-out policies, professional training and qualification programs (Geels, 2019; Lenton et al., 2022; Rogge & Reichardt, 2016). Policymakers also appear as relevant actors in cultivating a positive public opinion on SETs since policies shape public debates. One way in which policymakers can foster public acceptance is via knowledge and education programmes (Sultana et al., 2023).

Indeed, a large list of policy instruments is typically utilised to promote cleaner technologies. To explore the full extent of the role of policymakers, this thesis analysed which specific policies had been relevant to trigger the positive tipping point of each case study. Due to the plenitude of policy instruments available, and since specific policies (names) differ between case studies, policy instruments were categorised according to their objective. Policies can be divided into three main groups: economic, regulatory, and information instruments (Rogge & Reichardt, 2016). Each instrument type was further sub-divided into three categories, as briefly explained below. This is explained in detail in Section C.1.

Firstly, **economic instruments** might involve financial incentives, market-based mechanisms, and R&D support schemes. *Financial incentives* aim to stimulate investment in renewable energy projects and promote the adoption of sustainable practices. Examples include purchase subsidies, feed-in tariffs, tax rebates, and green bonds (Cox, 2016). *Market-based mechanisms* leverage market forces to internalise the costs of environmental externalities and promote emissions reductions. Examples are carbon taxes, tenders, and net metering (Joskow & Schmalensee, 1998). *R&D support* aim to foster innovation and technological advancements through financial support for R&D and demonstration projects (Huergo & Moreno, 2014).

Secondly, **regulatory instruments** can relate to compliance and standards, strategic planning, or market facilitation. *Standards* ensure that products, practices, and operations meet specific requirements or guidelines. Examples include mandating products or practices, intellectual property rights, renewable portfolio standards, technology/performance standards, and product labelling (ISO, 2024). *Strategic planning* involves the setting of goals, targets, and long-term roadmaps to guide policy development and decision-making (Gregersen & Johnson, 2010; Newell & Simms, 2021). *Market facilitation* focuses on creating conducive environments for the development, growth, and integration of renewable energy markets and infrastructure. This is done via local market design/development, infrastructure provision, and grid access guarantees (Steinbach & Bunk, 2024).

Thirdly, **information instruments** encompass knowledge transfer, collaboration and networking, and outreach initiatives. *Knowledge transfer* is designed to facilitate the exchange of information, expertise, and skills among stakeholders. Examples include professional and entrepreneurship trainings and scientific workshops (Kochenkova, Grimaldi, & Munari, 2015). *Collaboration and networking* incentives aim to facilitate partnerships, cooperation, and knowledge sharing among stakeholders, which is accomplished through public-private partnerships, cooperative RD&D programs, thematic meetings, and clusters (Q. Chai & Zhang, 2010; Lehmann, 2006). *Outreach* instruments aim to raise awareness, educate the public, and engage stakeholders. Examples include rating and labelling programs, public information campaigns, and organised public debates.

The descriptions given in this subsection point at six additional actor roles compared to the seven identified in the model Geels and Ayoub (2023) (see Section 4.1.2). These would be 8) raising awareness, 9) fostering collaboration, 10) transferring knowledge, 11) giving & receiving feedback, 12) establishing market-based mechanisms, and 13) promoting education. Therefore, this subsection suggests that Geels and Ayoub (2023) did not consider important actors and actor roles in their analysis and model. In Section 4.3.2, a few additions are proposed to the feedback loop model of Figure 4.3.
4.3.2. Initial Version of Expanded Feedback Loop Model

Through this initial literature investigation, four additional actors were found to have an important contribution to triggering a positive tipping point in SET development and deployment. This came with additional roles, considered relevant to include in the feedback loop model. Based on the descriptions given in Section 4.3.1, it is possible to draft an expanded version of the feedback loop model, presented in Figure 4.4. While it was attempted to keep the diagram as simple as possible, these additions were thought to be essential as simplicity should not come at the cost of omitting relevant information. Therefore, despite more complex, Figure 4.4 is more complete than Figure 4.3. The grey blocks and black arrow labels were already included in the refined version of the model (see Figure 4.3). The orange blocks and orange italic labels correspond to the additions proposed.



Figure 4.4: Proposed expanded version of the feedback loop model based on the literature presented in Section 4.3.1

This initial expanded version of the framework was tested with four case studies, allowing to understand whether these additions were found in practice and whether other important factors were still missing. Chapter 5 covers the case studies methodology utilised to conduct the research presented in this report.

5 | Case Studies Methodology

This chapter explains the methodology used to analyse the case studies. Section 5.1 provides the reasoning behind the chosen case studies and explains how they were analysed. In Section 5.2 the methods used to collect data are enumerated and it is detailed how the relevant data was selected. The chapter ends in Section 5.3 with the method used for the validation.

5.1. Case Studies Selection and Analysis

The concept of positive tipping points is fairly recent, with limited literature focusing on socio-technical tipping points (Geels & Ayoub, 2023; Lenton et al., 2022). An exploratory approach is thus adequate since it allows to understand the interactions between the different actors and among these and the SET. Case studies are a valuable research tool in exploratory research concerning under-researched fields (Fitzgerald, 1999). Besides, case studies are appropriate in examining qualitative dimensions like evolving perceptions and strategies, intricate interactions, and the tracing of processes. Lastly, a comparative design is useful to demonstrate distinct patterns and results (Geels & Ayoub, 2023).

To investigate the aforementioned propositions and to expand on the conceptual framework presented in Chapter 4, a comparative research design was used. Four case studies in which a positive tipping point had occurred were analysed. Section 5.1.1 gives the rationale behind the selection of the case studies, while in Section 5.1.2 the method used to analyse them is clarified.

5.1.1. Case Studies Selection

It is proposed that the actors engaged, their role, and the tipping points dynamics may exhibit variations not only when looking at B2B versus B2C, but also when looking at the deployment of the same technology in different contexts. For a B2B technology, such as wind or solar farms, the target customers are energy or utility firms that purchase large quantities of the technology to produce and sell energy. Comparatively, a B2C technology, like electric vehiles (EVs) or roof solar panels, is mainly purchased in small numbers by individuals, i.e. the end-consumer itself, and hence behavioural factors are expected to play a larger role (Geels & Ayoub, 2023). Further, it is proposed that the deployment dynamics of the same technology in different countries most likely involves distinct interactions between the actors and between these and the technology.

To compare the dynamics involved in the roll-out of B2B and B2C technologies, the adoption of wind turbines was compared with that of electric vehicles. Then, the impact of the context was evaluated by analysing the uptake of each technology in two countries. This design ensures that one variable, namely the type of technology remains constant, facilitating the examination of contextual changes in actors, actor roles, and feedback loops. Following this logic, four case studies were analysed, being them offshore wind in the UK, onshore wind in Portugal, EVs in the UK, and EVs in the Netherlands. Figure 5.1 gives a graphical representation of the chosen case studies.



Figure 5.1: Matrix of the selected case studies, showing whether they are representative of business-to-business (B2B) or business-to-consumer (B2C) technologies

Offshore wind and EVs in the UK drew inspiration from the investigation of Geels and Ayoub (2023). Using their work as a starting point, the objective was to investigate whether additional actors were engaged in these two case studies and to what extent, i.e. what role they played and which feedback loops they contributed to. This was done by analysing information provided by other sources. Then, to observe the effect of context, the deployment of onshore wind turbines and of EVs was analysed in two other countries. Western-European countries were chosen because choosing a country with a completely different context and government structure (e.g., China) could mean that the observations differed so much from one case to the other that no parallel could be drawn.

For one, onshore wind energy in Portugal was examined since there was a noticeable acceleration post-2004 and an apparent stabilisation phase after 2010, as evident in the S-shaped curve of its cumulative capacity graph (see Figure 6.3). The onshore wind sector was analysed as the offshore sector is only now taking its first steps in Portugal (Costa, Simões, Couto, & Estanqueiro, 2021). This was, nonetheless, considered to be a relevant case study to compare with offshore wind in the UK for two reasons. First, onshore wind turbines are also a B2B technology. Second, both onshore and offshore wind are based on the same technology (even though the application constraints differ in each case). For another, the uptake of EVs in the Netherlands, which saw a substantial surge post-2019 (see Figure 6.9), was investigated and compared to the roll-out of the same technology in the UK.

The term 'electric vehicle' might have different interpretations (Contestabile, 2019; Geels & Turnheim, 2022). In this report, as in Geels and Ayoub (2023), it refers to cars able to operate fully electric, without the assistance of an internal combustion engine (ICE) or a fuel cell, i.e. in zero-emission mode. Many organisations use the term in this way, including the International Energy Agency (IEA, 2023b), from which the data to construct the graphs of EV stock over time (Figure 6.5 and Figure 6.9) was retrieved. EVs hence encompass battery-electric vehicles (BEVs) and plug-in hybrid vehicles (PHEVs). BEVs only rely on electric motors and batteries (Geels & Turnheim, 2022). PHEVs can also run on an electric motor and battery but use an ICE or fuel cell when the battery is discharged (Contestabile, 2019).

5.1.2. Case Studies Analysis

As pointed out in Section 3.1, Geels and Ayoub (2023) analysed a very broad time period, where perhaps not all events contributed to triggering the positive tipping point. In this thesis the dynamics of

positive tipping points were used to select the relevant time period to analyse. Leveraging quantitative data on the deployment figures of each technology, a positive tipping point in adoption was identified as occurring when user adoption markedly accelerated, as described in Section 2.4. For the wind energy case studies, data on the cumulative installed capacity was used¹. For the EV case studies, the stock of EVs was utilised. A graph of adoption over time was created in Python for each case study ². By adding to the graph the trend line before and after the identified tipping point it was possible to better observe how the rate of adoption changed from before to after the tipping point was crossed.

Having the goal of understanding what triggered each positive tipping point, a zoomed-in period around the tipping point was analysed for each case study. This way, events that did not directly contribute to triggering the positive tipping point were not analysed. The length of this time period was case-specific as different transitions take a different duration to unfold (Grubler, Wilson, & Nemet, 2016; Smil, 2016).

The period immediately before - belonging to the enabling phase in Figure 2.1 - was analysed as it creates the conditions for the tipping point to be triggered and the accelerated adoption to take place. The analysis started once a clear change in the attitudes of the main actors involved was observed. This means the actors became confident in investing significant resources and time into the technology, which had reached a stabilised design (Geels, 2019) (see phase 2 of Figure 4.1).

Additionally, in the years after the tipping point, that change of attitudes and investment decisions had to be sustained such that the self-perpetuating growth that occurs after a tipping point was present. Hence, for each case study the years immediately after the positive tipping point were also analysed. This covered part of the accelerating phase in Figure 2.1, but not all of it. One of the reasons was that, since the interest was in the period surrounding the tipping point, the analyses should not cover the years approaching the transition from the accelerating to the stabilising phase. The other reason was that for three of the case studies the technology is currently still in the accelerating phase, with the exception of the uptake of wind energy in Portugal, which has entered a stabilisation phase.

In each case study the actors and their respective roles were analysed, and a representative feedback loop model was created. Figure 4.3 was used as a starting point. If the blocks or arrows existent in this initial feedback loop model did not apply to that case study, they were removed. Subsequently, the additional actors found were added as blocks and any relevant interactions between actors and between these and the technology were added as arrows, using an appropriate label. An important aspect in comparison with Geels and Ayoub (2023) is that interactions that were not part of any feedback loop in specific were also added to the model. At the end, each case-specific model was compared to what had been found in the initial literature analysed (see Section 4.3.2). Each case study's feedback loop model was then described, providing an explanation for each arrow label of the model. This is considered an improvement compared to Geels and Ayoub (2023), where the corresponding feedback loop was added throughout the text between square brackets. As discussed in Section 3.1 this made it hard at times to understand which information exactly contributed to that feedback loop.

5.1.3. Case Studies Comparison

After analysing the enabling phase and part of the accelerating phase of each case study, the findings were compared among each case in order to understand whether a pattern existed among different

¹Geels and Ayoub (2023) used figures on electricity production, which might not be the most adequate data in this case (see Section 3.1).

²These graphs correspond to Figure 6.1, 6.3, 6.5, and 6.9

cases. For one, the actors involved were compared. To do so, the actors were divided into different categories (corresponding to the blocks in the feedback loop models), making the implicated stakeholders comparable among case studies. This is similar to what Geels and Ayoub (2023) have done in their own model. As an example, even though government departments in different countries differ in name and possibly nature, they all fall under the 'policymakers' category.

For another, the roles played by the actors were compared through comparing the four case-specific feedback loop models. This comparison allowed to draw conclusions on the general actor roles present when triggering a socio-technical tipping point. This comparison was important since it was expected that the overall functions performed remained (almost) the same, even if that function was performed by different actor groups. One actor role that had special attention, including in the research sub-questions, was that of policymakers and their policy instruments. This was because there seems to be a large knowledge gap on how policymakers can enable positive tipping points. Hence, the policies employed in each case study were compared. To enable this comparison, economic, regulatory, and information instruments were divided into sub-categories (see Section 4.3.1 and Appendix C).

Based on this comparison, two final feedback loop models were derived. Namely, the two wind energy feedback loop models were overlapped, as was done for the EV models. The reason for two and not one final model lies in the fact that significant differences were identified between the B2B and B2C case studies concerning the relevant actors and actor roles. To make the similarities and distinctions between the models clearer, different colours were used (as explained in Section 7.1.1).

5.2. Data Collection

Having explained the case studies' selection and analysis, Section 5.2.1 covers the data collection methods, while Section 5.2.2 explains how the relevant data was selected among all the data available.

5.2.1. Data Collection Methods

Geels and Ayoub (2023) employ a process tracing methodology in their case studies, emphasising its suitability for exploring phenomena marked by multiple causal pathways. This approach is particularly apt for investigations interested in temporal flow, sequences, and interacting feedback loops, and was therefore used in this Master's thesis. Process tracing seeks to unveil explanations detailing how a process unfolds over time, focusing on tracing conjunctions and steps in a developmental sequence. In this investigation, the aim was for an analytic explanation, which involved converting a historical narrative into an analytical causal explanation. By combining quantitative and qualitative information, the investigation delved into how agency, techno-economic developments, and interacting feedback loops contributed to the emergence of socio-technical tipping points.

In terms of collecting the necessary data for the case studies, the articles from Vasseur, Kamp, and Negro (2013), Mazur, Contestabile, Offer, and Brandon (2015), Rosenbloom, Berton, and Meadowcroft (2016), Kamata, Khosla, and Narayanamurti (2020) and Geels and Ayoub (2023) provide a good overview of common practices. The case studies employed quantitative data extracted from (national) statistical databases to track deployment figures. This included sources such as Statista (2024), BVG Associates (2021), and RenewableUK (2024)³ for offshore wind in the UK, Costa et al. (2021) for on-

³For the case of offshore wind in the UK three sources were necessary to build its deployment graph (provided in Figure 6.1) since each of them provided data for different time periods.

shore wind in Portugal, and IEA (2022) for EVs both in the UK and in the Netherlands. These sources were found through the Google search engine by searching for: *technology's name + country's name + statistics/deployment rate/over time*. An example would be: *offshore wind + UK + statistics*. Quantitative data was also collected concerning the number of public EV chargers in each country over time by searching for instance *public chargers + electric vehicles + country's name*. In this respect, DfT (2023) was used for the UK and Statista (2023) was used for the Netherlands.

While a positive tipping point is in this thesis identified by a rapid increase in technology deployment or user adoption, numerous factors accumulate gradually, allowing the crossing of this threshold. As emphasised by Geels and Ayoub (2023), substantial changes manifest in companies' perspectives and (investment) strategies. Additionally, policymakers may enact considerable alterations in policy goals and instruments over time. Furthermore, noteworthy surges in public attention or shifts in discursive content could contribute to crossing a positive tipping point. These significant changes in strategies and perspectives were not, however, identified in this thesis as positive tipping points in themselves, unlike what was done by Geels and Ayoub (2023). In the current study these dynamics simply contributed to the interactions and feedback loops that led to each socio-technical tipping point.

Qualitative data sourced from newspaper databases was utilised to monitor the nature of interactions between actors. Newspapers prove valuable as mass media coverage is seen as the most relevant means of attention and political communication (Geels & Ayoub, 2023). For example, information about protests against or in favour of a technology that might influence its deployment profile can usually be found in newspapers. Newspapers also showed valuable in discovering large investments and industry coalitions. Keywords such as 'offshore wind', 'wind energy' and 'electric vehicles' were used, as well as the year of interest. Such data was especially relevant when conducting the two fully new case studies, i.e. wind in Portugal and EVs in the Netherlands. This is because for the case studies researched by Geels and Ayoub (2023) data had already been collected concerning these aspects.

To identify shifts in perceptions, strategies, and objectives, websites, (annual) reports, white papers, and newsletters, as well as academic, industry and governmental publications were consulted. These provided information on the timeline of events, on the policies employed, and on the interactions between the different actors. Interviews with organisations and field experts were not used to collect insights on these aspects since written sources are more reliable than the partial perspective of interviewees. While most of these data sources are accessible via the Google search engine using appropriate key words academic publications were searched for differently. Some were found through the reference list of other relevant articles. Others were found via article and book finders such as Google Scholar, ScienceDirect (2024), Connected Papers (2024) and Elicit (2024).

As an example, for the case study of EVs in the Netherlands⁴, Science Direct was searched for '*Netherlands* + *EV*' and for '*Netherlands* + *electric vehicles*'. Besides, searches in Google Scholar were performed using these same entries, both in English and Dutch. Through the articles found in this way, other relevant articles, reports and websites were found via the reference list. These sources provided invaluable information on the evolution of the technology over the past years, a lot of the times with a focus on the policies employed. At last, in order to find additional information, sometimes related to a specific actor group, other reports and websites were searched for with the Google search bar. Examples include '*ICCT electric vehicles* + *Europe/Netherlands* + *[year]*', '*electric vehicles* + *Netherlands* + *[year]*', '*electric vehicles* + *Netherlands*', again searched in English and in Dutch.

⁴A similar method was used for the other case studies.

5.2.2. Collecting Relevant Data

Concerning which data was relevant to collect, firstly, a general understanding had to be gained about each case study such that the main actors, events and milestones before and after the positive tipping point were understood. For this, sources that provided an overview of the technology's development over time were crosschecked. Also at this stage, information related to changes in actors' perspective and strategies was used to determine the duration of the zoomed-in period surrounding the positive tipping point. This information was only used to understand the context and the background of the deployment history and to determine the years to include in the case study analysis.

Then, the analysis continued on the identified zoomed-in period. This way, the remaining information collected was narrowed down and the focus was put on information about the actors involved (who were the active actors?), and the actions of those actors (what did actors do that affected another actor or the technology development?). Specific information was searched concerning the interactions included in the model of Geels and Ayoub (2023) and the ones proposed in Figure 4.4. Information that did not provide any insights on the actors and actor roles involved was in general not included in the report unless it was necessary to provide background information to the reader.

Two Word documents were used to collect the relevant data for each case study. In the first one, an entry was created for each visited source, under which all the relevant information was pasted. In the second document this information was assigned to the relevant feedback loop(s) and actor interactions. There was an entry for each interaction identified so far in the previous feedback loop models. This document had a structure similar to that of the sections in Chapter 6. The second Word document also had an entry for information concerning the introduction and the conclusion to the case study. This way, the information found in the different sources could be attributed to the corresponding document entry for that specific case study: interactions between actors, case introduction, or case conclusion. If an interaction was found to be important for the technology developments but was not yet captured in the model, a new entry was added to the document. This process was iterative. For instance, if a new interaction was found on the second case study, the first case study would be re-evaluated to understand whether that interaction had been missed by accident. At last, the raw information collected on the second Word document was converted into the case study descriptions in Chapter 6.

Thus, to be relevant information had to comply with at least one of three criteria. First, it had to correspond to an interaction between two actors or between an actor and the technology during the period analysed. Second, it corresponded to general information on the deployment of that technology in the respective country, particularly on adoption figures, during the period being analysed. The third option was that the information provided direct insights into which were the main contributing criteria to triggering the positive tipping point.

As much information as possible was collected from the most 'easy-to-find' sources. The second step was to check which entries in the second Word document were still missing information. Information concerning these specific topics was searched for. To limit the complexity of the case-specific and final feedback loop models, only the most relevant actors and interactions were included in the models. Usually, if an actor group was 'relevant', meaning it had a significant role in promoting the fast development or adoption of the technology, it was referred to in (several) articles and/or reports that provided an overview of the technology's development in that country. Examples include policymakers, technology firms, adopters, academic and research institutes, and NGOs. At other times, however, it

might be that a certain actor group is not mentioned at all or is only briefly referred to in one or two of the visited sources. This is already a sign that most likely this actor category was not (as) essential to trigger the positive tipping point. If after searching for more specific information on that actor group or interaction not a lot of information was found and/or this information was very hard to find, then it was assumed that it was not a big contributor to the accelerated technology adoption.

For instance, in the EV case studies there were financial institutions that provided leasing options and made it easier for private car owners to buy a (electric) vehicle. However, information on this topic was not provided in any of the scientific articles and official reports analysed, most of which provided an overview of the main developments concerning EV uptake in the UK, in the Netherlands, or in Europe. Some, very segregated information, was only found after searching specifically for leasing options and for the name of particular banks that operate in each country. However, even so, financial solutions did not appear to specifically target electric vehicles. As this actor group did not seem to have a decisive role in promoting EVs in neither country, it was decided to not add it to the respective feedback loop model. This way, the complexity of the model was not increased unnecessarily. These findings agree with sources such as Muratori et al. (2021).

5.3. Validation

The case studies and the final feedback loop models were validated through semi-structured interviews with relevant stakeholders who worked in the specific sector and country of one of the case-studies, or who were very familiar with the concept of positive tipping points. The interviewees are kept anonymous but some general information about them and the topic they validated is provided in Table 5.1.

Three of the four validation interviews were performed via a video call on Microsoft Teams, which was recorded with the interviewees' consent⁵. The remaining interview was performed via e-mail. In all the validation interviews, a PowerPoint presentation was prepared where the main findings (relevant to the interviewee) were included. The interviewee was then asked to provide feedback on them and comment on whether it was a good representation of reality. Based on the feedback received, a few adjustments were made to the case studies, as described in Section 7.2.

Interviewee	Employer	Position	Validation topic
1	Big 4 consultancy firm	Manager	Offshore wind and EVs in the UK (Section 6.1 and Section 6.3)
2	Company focused on wind en- ergy production	Project manager	Wind in Portugal (Section 6.2)
3	Big 4 consultancy firm	Senior manager	EVs in the Netherlands (Section 6.4)
4	Dutch technical university	Postdoctoral researcher	Conclusions and final feedback loop model (Section 7.1 and Section 7.3)

Table 5.1:	Description	of experts intervie	wed for the validation
10010 0.11	Description	or experte intervie	weallor the validation

Having explained the methodology related to selecting, analysing, and validating the case studies, as well as collecting the necessary data, Chapter 6 presents the four case studies conducted.

⁵These recordings were deleted once notes from the interviews were taken.

6 | Case Studies

In this chapter the case studies are presented. Section 6.1 and Section 6.2 cover the expansion of the (offshore) wind sector in the UK and in Portugal, respectively. Then, Section 6.3 and Section 6.4 elaborate on the deployment of electric vehicles in the UK and in the Netherlands, in this order.

6.1. Wind Energy in the UK (2002 - 2014)

The United Kingdom has established itself as a global leader in offshore wind deployment, representing 21.6% of the world's cumulative installed capacity by 2023, with only China having a larger market share (GWEC, 2023). This growth is evidenced by a significant increase in offshore wind electricity generation, rising to 13.8% of the total electricity produced in the UK in 2022 (RenewableUK, 2023).

As shown by Figure 6.1, the offshore wind industry in the UK markedly accelerated after 2009, going from a slow to a rapid growth phase. Offshore wind kept growing steadily afterwards. The exceptions were 2014-2016 due to an overlap between two similar policies (McNally, 2022), and 2019-2021 in light of the Covid-19 pandemic (BVG Associates, 2021). However, in both cases, the industry quickly picked up afterwards. It can thus be said that in 2009 a positive tipping point was crossed (Geels & Ayoub, 2023). In 2002 policymakers introduced the Renewable Obligation (RO), which represented a clear shift towards renewable energy (Geels & Ayoub, 2023). Since then, this policy was amended and other policies were introduced. Therefore, 2002 was defined as the beginning of the enabling phase. With the tipping point occurring in 2009, it was decided to analyse the case study until 2014, the year in which Contracts for Difference were introduced and effectively a new stage within the accelerating phase started (McNally, 2022). With this being said, this case study was analysed from 2002 until 2014.



Figure 6.1: Offshore wind energy in the United Kingdom from 2003 to 2022 (BVG Associates, 2021; RenewableUK, 2024; Statista, 2024). The detailed data is given in Table A.1

As pointed out by Geels and Ayoub (2023), policymakers, firms, adopters, and wider publics were four of the main actors involved. Nonetheless, the deployment of offshore wind in the UK involved a highly networked coalition of actors, including large utilities, government bodies, research institutes, environmental NGOs, and industry networks. This collaboration has been crucial in boosting the credibility of offshore wind and channelling resources into its development. The involvement of both public and private entities underscores the complex and multifaceted nature of the offshore wind sector in the UK (Geels & Turnheim, 2022; Kern, Smith, Shaw, Raven, & Verhees, 2014; Toke, 2011).

Figure 6.2 shows a representation of the interactions among these actors, as well as among them and the technology for the period of 2002 until 2014. Ultimately, this is representative of the role that each actor played in the accelerated deployment of offshore wind in the UK after 2009. In Figure 6.2, the grey blocks and black labels were retrieved form Geels and Ayoub (2023). The orange blocks and labels correspond to actors or interactions that were proposed based on general literature (see Figure 4.4.) The blue labels are additions made solely based on the case study analysis.

As it can be observed, in some occasions information was added to the interactions identified by Geels and Ayoub (2023). Besides, several other interactions were included in the model. As a result, three new feedback loops were identified, alongside several 'loose' interactions. Overall, it can be said that the complexity of the model significantly increased in comparison to Figure 4.3, with twice as many actors and more interactions. Nonetheless, it is argued that the model in Figure 6.2 provides a better understanding of how the development of offshore wind in the UK unfolded in the period of time immediately before and after the positive tipping point.

The involvement of *technology firms* included wind turbine manufacturers such as Vestas (since the early 2000s) and Siemens (since the late 2000s), as well as civil engineering contractors (e.g., Balfour Beatty since the late 2000s) involved in developing the transmission infrastructure (Kern et al., 2014). *Adopters* refer to buyers of offshore wind turbines, who either utilise the electricity directly or sell it to end-consumers. Initially, the buyers of wind turbines were the energy firms themselves, who acted as project developers. Nevertheless, in the mid-2010s offshore wind farms started being developed by alliances of project developers and investors who then sold them to operators (Geels & Ayoub, 2023).

Below, an explanation of the role that the relevant stakeholders played is given, starting with the feedback loops and concluding with the interactions that are not associated with any particular feedback loop. For clarity, a title in bold is added before each explanation; these correspond to the arrow labels in Figure 6.2. As in the figure, the new labels are shown in italic and in either orange or red.

$\textbf{Feedback 1\& Feedback 2: Technology Firms} \leftrightarrow \textbf{SET} \leftrightarrow \textbf{Adopters}$

Investment in R&D, production, *and infrastructure:* The period from 2005 to 2009 was especially relevant as it marked a shift towards commercialisation in the UK's offshore wind sector. This was accompanied by equipment manufacturers of offshore wind turbines investing large amounts in R&D to improve their technical and data gathering abilities (Geels & Ayoub, 2023; Higgins & Foley, 2014). Besides, civil engineering contractors (e.g., Balfour Beatty since the late 2000s) were involved in developing the transmission infrastructure (Kern et al., 2014). As a result, the time period between 2002 and 2014 was characterised by significant technological advancement, leading to the development of wind turbines with higher capacities, as well as initiating an intense rivalry among turbine manufacturers to produce larger and more efficient turbines (McNally, 2022).



Figure 6.2: Feedback loop model showing the relevant actors and respective roles in the diffusion of offshore wind in the UK for the period of 2002-2014

Value: The commitment of technology firms to technological advancements resulted in wind turbines with higher sizes, capacities, and efficiencies, especially after 2010, meaning an overall better value for adopters (McNally, 2022). With the improved technical performance came lower costs too as the cost of components, seabed foundations, undersea cables, installation, commissioning, operation and maintenance reduced (Geels & Ayoub, 2023).

Purchases; Revenues and profit: The better value proposition of offshore wind turbines due to technological improvements and the amended Renewable Obligation in 2009 in combination with positive discourses led project developers to invest increasing amounts in offshore wind energy, in turn increasing the revenue and profit of technology firms (Geels & Ayoub, 2023).

Feedback 3: Policymakers \leftrightarrow Technology Firms

Lobbying: After the policy changes of 2009 the confidence and interest of potential project developers in offshore wind energy increased. However, they perceived the strong dependency of the market on subsidies as a risk. Therefore, both project developers and manufacturers (technology firms) lobbied for policy changes via the Offshore Wind Developers Forum and the Offshore Wind Cost Reduction Task Force (Kern et al., 2014).

Loans and capital grants: The Offshore Wind Capital Grants Scheme was launched in 2002 by the government's Department of Trade and Industry in order to promote the early development of offshore wind. With the intention of encouraging technology firms to improve their products, processes, and capabilities related to offshore wind energy it had the goal of reducing costs and increasing confidence in the technology. The twelve projects, all completed in 2003 had a total funding of £117 million (Higgins & Foley, 2014; IEA, 2014). Other grant schemes include the capital grants given by the Environmental Transformation Fund, which between 2009 and 2011 provided three projects with £26 million in order to promote understanding of supply chain technologies, as well as the Offshore Wind Manufacturing Funding which provided £130 million between 2011 and 2015 for the establishment of manufacturing capabilities at ports (Kern et al., 2014).

R&D subsidies: Public funding accelerated after the mid-2000s, with R&D subsidies aiming at improving efficiency, reducing costs, and developing larger and more robust turbines. The Offshore Wind Accelerator, funded by the Carbon Trust between 2008 and 2014, is an example of a collaborative RD&D programme having the goal of reducing the costs related with offshore wind by 10% until 2015 (Kern et al., 2014).

Foster collaboration: The Department of Energy & Climate Change (DECC), established in 2009, set up critical policies, provided funding, and promoted collaboration between technology developers and support innovation. Another actor which, among other things, was involved in fostering collaboration through organising supply chain events was the Crown Estate (CE), which actively promoted offshore wind energy since 2000. Despite being a private body, the CE was established by the government (interviewee 1) and has the overall goal of bringing prosperity for the nation (The Crown Estate, 2023). Hence, its contribution was included in interactions involving policymakers. Besides, the Technology Strategy Board (TSB) also aimed to facilitate knowledge sharing (Kern et al., 2014).

Feedback 4: Policymakers \leftrightarrow SET

Policies, goals, and coalitions adjustments: The amended Renewable Obligation was a direct result of evaluating the efficiency of the original policy, which did not bring about the desired outcome. This led policymakers to adjust the RO and start providing stronger support to more expensive technologies, which effectively meant that offshore wind energy received twice as much support as before (Geels & Ayoub, 2023). Furthermore, as the technology advanced and offshore wind became more attractive and feasible, policymakers were more inclined to form coalitions with industry stakeholders to support the continued development of offshore wind projects. This is showcased by the collaborative opportunities offered by the CE, the DECC, and the TSB (see Feedback 3). Additionally, policymakers adjusted their goals related to renewable energy production and offshore wind energy based on the evolving capabilities and potential of the technology. For instance, both UK and European renewable energy targets have been updated over time (see the interaction Policymakers \rightarrow Adopters).

Infrastructure investment: In 2009 the DECC introduced the Interim Connect and Manage programme, helping streamline the grid connection process, typically complex and time-consuming. This was followed by the Connect and Manage programme, fully implemented in 2011 by the DECC, through which new projects applied for a quick grid connection. The difference was mostly that the latter programme was established as a more permanent framework for allocating and managing grid capacity for offshore wind projects (Metcalfe & Sasse, 2023; Ofgem, 2013).

Regulations and standards: Some regulations and standards concerned the technology in general rather than technology firms or adopters in specific, even though these actor groups had to comply to such regulations. For example, an Environmental Impact Assessment had to be completed for offshore wind projects, ensuring projects complied with environmental regulations. Furthermore, there were regulations related to habitats and species protection, creating a safety zone around the offshore wind farm, and decommissioning the wind farm (SEA, 2009). A last example are grid connection regulations, which were established to govern the connection of offshore wind farms to the onshore electricity grid (BVG Associates, 2019).

Another important regulation was the European Emissions Trading System (EU ETS), introduced in 2005, which imposed a cap on emissions that can be emitted by the different companies. Tradable emission allowances are then issued to these entities, which can be bought or sold on the carbon market. Ultimately, this meant that offshore wind projects became more economically attractive compared to fossil fuel-based energy generation (European Commission, 2021). The EU ETS was complemented, in 2013, by the introduction of the Carbon Price Floor, which established a minimum price on carbon emissions from the power sector. This policy promoted low-carbon energy generation by further increasing the cost of emitting carbon dioxide and setting a stable and predictable carbon price, which helped reduce investment uncertainty for renewable energy projects like offshore wind (HM Revenue & Customs, 2018).

Feedback 5: Adopters \leftrightarrow Wider Public

Further adoption; Expanded technology's visibility: The enhanced deployment of offshore wind farms in the UK in the early 2010s was propelled by both the 2009 policy adjustments and positive discourses among the industry and the wider public, which sparked the interest of international and British energy firms in the technology. Furthermore, the increased adoption of the technology led to economies of scale, created more debates, and gave it further visibility, which in turned spurred more interest among investors (Geels & Ayoub, 2023).

Feedback 6: SET \leftrightarrow Wider Public

Solution to societal problems; Legitimacy: In 2009/2010 there was a surge in public attention for offshore wind due to the positive debates on the RO amendment. These public and political discourses touched upon topics such as decarbonisation, jobs, energy security, industry growth, UK's offshore conditions, and the social acceptance of offshore wind compared to onshore (Kern et al., 2014). Overall this helped frame offshore wind as part of the solution to tackle climate change, which legitimated the technology (Geels & Ayoub, 2023).

Feedback 7: Policymakers \leftrightarrow Wider Public

Public opinion; Strong policy plans and accompanying narratives: During the 2000s public concerns about climate change grew and started to pressure the policymakers. In response, the UK government started taking more climate action. For instance, in 2008 the Climate Change Act was introduced, defining that in 2050 the country's emissions should have lowered by 80% compared to 1990 levels. This was also followed by the 2009 adjustments to the RO which gave an appealing support premium to offshore wind projects. As a result, in the early 2010s there were both public and political positive discourses (Geels & Ayoub, 2023; Kern et al., 2014).

Feedback 8: Technology Firms \leftrightarrow Academic & Research Institutes

Funding: Interviewee 1 highlighted the relevance of adding this interaction to the model. Indeed, Siemens, a major player in wind turbine manufacturing, established a research centre at the University of Sheffield in 2010. This collaboration aimed to advance offshore wind technology through joint research projects, knowledge exchange, and training programs (University of Sheffield, 2019). Furthermore, ScottishPower Renewables, another major player in the UK offshore wind sector, collaborated with academic institutions like the University of Strathclyde. These partnerships focused on research areas such as grid integration, offshore wind farm optimisation, and novel turbine technologies (University of Strathclyde, 2022).

Knowledge transfer: Universities such as the University of Strathclyde and the University of Edinburgh provided from 2009 and 2012, respectively, doctoral programmes with offshore renewable energy training, alongside entrepreneurial and commercial skills development. Some of this training was provided by businesses such as EDF Energy, E.ON, BP, Shell, Caterpillar and Rolls-Royce. From such programmes resulted job-seekers that were very specialised in offshore wind energy and could join the technology firms as experts. Knowledge transfer also happened from the knowledge institutes towards the technology firms. Indeed, academic and research institutes were essential to the development of offshore wind energy in the UK, namely when it came to technology's cost reductions and performance improvement. Three of the most important publicly funded organisations that contributed to research and innovation were the Carbon Trust, the Energy Technologies Institute (ETI), and the National Renewable Energy Centre (NAREC). All were active in the field of offshore wind since the late 2000s, with the ETI starting a dedicated offshore wind programme in 2009 (Geels & Turnheim, 2022; Kern et al., 2014). The knowledge obtained in such institutions would then be transferred to the industry through published reports and articles, as well as collaboration opportunities such as the Offshore Wind Cost Reduction Task Force (RenewableUK, 2012). Furthermore, knowledge and expertise was shared for instance by NAREC, which provided test and technical services since the late 2000s (BVG Associates, 2021).

Feedback 8: Policymakers \leftrightarrow Trade Associations

Consultation; Foster collaboration: Several trade associations were involved in representing its members next to the government (see Feedback 9), that way fostering collaboration between the industry and policymakers. Examples include RenewableUK, the Offshore Wind Developers Forum, the Offshore Wind Industry Council, the Global Underwater Hub (previously SubseaUK), and the Scottish Renewables (Global Underwater Hub, 2024; Kern et al., 2014; RenewableUK, 2012; Sottish Renewables, 2024). Trade associations were not the only ones approaching policymakers, with the latter seeking advice from the former (interviewee 1). For instance, in 2009

the DECC consulted 56 parties, including trade associations, to obtain feedback on their most recent Strategic Environmental Assessment for offshore wind leasing (DECC, 2009). Besides, Offshore Wind UK, who claim to be the "leading representative body for the UK's offshore energy industry", mention they engage with regulators and governments (Offshore Energies UK, 2024).

Feedback 10: Technology Firms \leftrightarrow Financial Institutions

Long-term certainty: The amended Renewable Obligation in 2009 effectively doubled the subsidy for electricity produced by offshore wind turbines, making the technology more attractive for investors (McNally, 2022). The amended RO provided generous and long-term (20 years) support, creating a subsidised market for offshore wind and boosting investor confidence (Geels & Ayoub, 2023). The confidence of financial institutions, however, increased more significantly after the introduction of the Contracts for Difference, in 2014, which helped bring down the cost of offshore wind energy projects (BEIS, 2019).

Financing solutions: This long-term certainty led private investors to gradually lower their interest rates which helped secure private financing (Geels & Ayoub, 2023). Nevertheless, most of the sponsoring came directly from energy and utility firms, with financial institutions taking a smaller role. Besides, most of the banks and institutional investors involved were state-owned or state-backed as these were willing to invest in higher-risk projects like offshore wind (BEIS, 2019).

$\textbf{Policymakers} \rightarrow \textbf{Adopters}$

Feed-in tariffs: The RO, released in 2002, required utility firms to have a specific amount of renewable electricity in their portfolio. Renewable Obligation Certificates (ROCs) were distributed among utility firms, with the same number of ROCs attributed to every renewable energy technology. Utilities could then trade these certificates until each one had their required amount (Geels & Ayoub, 2023). In 2009, the RO was amended and the number of ROCs per megawatt-hour (MWh) increased for offshore wind. This was a consequence of introducing banding to adjust the support based on the technology's maturity (Geels & Turnheim, 2022). This adjustment effectively doubled the subsidy for electricity produced by offshore wind turbines and provided an estimated premium on top of the retail price of £100/MWh (Geels & Ayoub, 2023).

From 2013 to 2015 the Final Investment Decision Enabling for Renewables (FiDeR) system was in place to help developers make investment decisions before the Contracts for Difference (CfD) regime was established, in 2015 (DECC, 2013). The CfD provided a fixed 'strike price' during 15 years to the winners of a bi-annual auction. This provided protection against wholesale prices volatility, attracting private investors (BVG Associates, 2021).

Tenders: The regulator of gas and electricity markets in the UK, Ofgem¹, was established in 1990 and played a key role in arranging the necessary grid connections (Geels & Turnheim, 2022; Kern et al., 2014). In 2009, the Ofgem and the DECC were involved in launching competitive tenders for offshore wind transmission licenses (Metcalfe & Sasse, 2023), which aimed to promote efficient transmission infrastructure development. Furthermore, the CE, which manages the seabed in UK waters, offered leases for offshore wind farm development. Interested parties would bid or apply for these leases through a competitive process. Successful applicants would then be granted the rights to develop and operate offshore wind farms in the designated areas. During the period analysed there were two lease rounds, Round 2 and Round 3. Round 3, conducted in 2009, was

especially important and marked a significant expansion in scale. It was the first time that the CE identified itself the zones with the most potential to be explored and co-invested next to the developers (Kern et al., 2014).

(Binding) targets: Since 1990, the UK government has set or signed up to ambitious targets, some of which were binding, which played an essential role in setting the industry's direction. Firstly, in 1990 the Non Fossil-Fuel Obligation specified the amount of capacity that should be generated from non-fossil sources (McNally, 2022). In 2007, the UK government signed up to the binding EU targets of the European Renewable Energy Directive, which required 15% of the country's energy to come from renewable energy sources by 2020 (McNally, 2022). This directive provided a broader framework for renewable energy adoption, including offshore wind. This was followed, in 2008, by the Climate Change Act, which set the UK's long-term climate targets in law for the first time. The Act aimed to reduce GHG emissions by 80% from 1990 levels by 2050. In 2009, the European Renewable Energy Directive came into force (Metcalfe & Sasse, 2023).

Additionally, the 2011 Renewable Energy Roadmap outlined the government's long-term commitment to increasing the share of renewable energy in the UK's energy mix. By setting clear targets and objectives for renewable energy deployment, including offshore wind, the roadmap provided policy certainty and signalled to investors and developers that offshore wind was a priority to the government (DECC, 2011). This was followed by the 2013 Offshore Wind Industrial Strategy, which provided a roadmap for the development of the offshore wind sector in the UK, addressing aspects such as policy, investment, supply chain, cost reduction, and job creation to support its growth and competitiveness. By laying out a long-term strategy, the government aimed to reduce investors' uncertainty and boost their confidence in the offshore wind sector (BIS & DECC, 2013).

Foster collaboration Several government initiatives involved adopters and developers in collaboration opportunities. An example is the Offshore Wind Investment Organisation. The in Northern Ireland the Offshore Wind Support Programme bridged wind farm developers with supply chain companies. Besides, the Carbon Trust Offshore Wind Accelerator is a public-private programme joining more than 75% of offshore wind developers in the UK (HM Government, 2014).

$\textbf{Policymakers} \rightarrow \textbf{Financial Institutions}$

Public financing departments or businesses: Apart from the subsidies and support premiums (see Policymakers \rightarrow Adopters), the UK government also financed offshore wind projects through state-owned or state-backed institutions. For example, the CE, the UK's sea bed leaser, co-invested in projects together with the project developers (Kern et al., 2014). The CE is an independent private organisation established by the government (interviewee 1). Besides, the Green Investment Bank (later acquired by Macquarie Group Limited) was established in 2012 (Green Investment Group, 2024) as an attempt to tackle the funding issue associated with large renewable energy projects and mobilise private financing into the sector (Higgins & Foley, 2014).

$\textbf{Policymakers} \rightarrow \textbf{Academic \& Research Institutes}$

R&D subsidies: From 2000 to 2004, the offshore wind energy sector in the UK experienced a focus on R&D and demonstration projects, stimulated by government support (Geels & Turnheim, 2022). However, research grants continue to be issued to this day in order to foster continuous

¹Ofgem stands for Office of Gas and Electricity Markets (Ofgem, 2013).

technological developments. Namely, the Engineering and Physical Sciences Research Council (EPSRC) (UK Research and Innovation, 2024c) started providing funding for offshore wind projects since the 1980s, while the TSB has been active since the late 2000s (Kern et al., 2014).

Foster collaboration: Some government initiatives created collaboration opportunities including academic and research institutes. For instance, the Offshore Wind Cost Reduction Task Force, established in 2011, involved academic and research institutes, the industry, and the government (RenewableUK, 2012). Moreover, in 2013 ORE Catapult was created to bring together industry, academic and research institutes, government, and other stakeholders to drive innovation, research, and collaboration in the sector (ORE Catapult, 2022).

Academic & Research Institutes \rightarrow SET

Research and test: Several academic and research institutes in the UK have been active in pursuing research and tests related to offshore wind, in diverse themes. These include the University of Strathclyde (2022), the University of Sheffield (2019), and the University of Edinburgh (2019), as well as the Carbon Trust, the ETI, and the NAREC (Geels & Turnheim, 2022; Kern et al., 2014).

Trade Associations \rightarrow Adopters & Technology Firms

Foster collaboration: Although in the early phases of offshore wind the main energy firms did not show much interest, this changed over the years and one important proof was the rising number of members in trade associations such as RenewableUK (Toke, 2011). Trade associations connected stakeholders within the industry and bridged the gap between these actors and policymakers. For example, the Offshore Wind Developers Forum, was set up in 2010 for industry collaboration and to enable discussions about common problems in the industry. Furthermore, RenewableUK, the central trade body in the UK, organises yearly offshore wind conferences since 2002 (Kern et al., 2014). The Offshore Wind Cost Reduction Task Force was created by the government in 2011 in order to bring the cost of offshore wind down. This was a collaborative project between government, knowledge institutes and industry and the Task Force was chaired by RenewableUK, that due to its influence was able to foster collaboration and discussion on the topic (RenewableUK, 2012). For instance, even though energy-related companies started seeing offshore wind as a better business investment after 2009, they were still cautious due to the market's strong dependency on subsidies. As a consequence, companies lobbied for better policies through the Offshore Wind Developers Forum and the Offshore Wind Cost Reduction Task Force (Kern et al., 2014). These were not the only trade associations involved in fostering collaboration, however. The Offshore Wind Industry Council (OWIC, 2021), the Global Underwater Hub (2024), the Sottish Renewables (2024), and the Renewable Energy Association (2023) are examples of trade associations that work to promote the growth of the offshore wind sector in the UK.

$\textbf{NGOs} \rightarrow \textbf{Policymakers}$

Lobbying: Environmental NGOs have shown enthusiastic support for offshore wind for, helping to create a window of opportunity for the technology. This includes campaigns in favour of large offshore wind projects from Greenpeace, Friends of the Earth (FoE), the Royal Society for the Protection of Birds (RSPB), and the World Wildlife Fund for Nature (WWF) (Toke, 2011).

$\textbf{NGOs} \rightarrow \textbf{Wider Public}$

Public awareness campaigns: NGOs such as Greenpeace, FoE, RSPB, and WWF spearheaded campaigns in support of offshore wind (Toke, 2011). For instance, the FoE published a booklet in 2012 clarifying frequently asked questions concerning offshore wind energy, aiming at raising awareness (Friends of the Earth, 2012). Besides, renewable energy lobby groups mobilised public concern about the perceived energy dependency to the point that, in 2008, 19% of UK citizens ranked energy as one of their top two concerns (Toke, 2011).

6.2. Wind Energy in Portugal (2000 - 2007)

Portugal does not have oil or natural gas reserves of its own but has historically relied substantially on fossil fuels. This resulted in a strong dependency on imports that challenged the country's security of supply. Portugal does have, nevertheless, strong renewable energy resources. To gain more energy independence and commit to a sustainable development, in the 1990s and early 2000s the Portuguese government started promoting renewable energy sources and the liberalisation of its energy market (IEA, 2004b; IRENA, 2013b). Wind energy, in specific, was strongly advocated for. Due to a very deep seabed in most of Portugal's sea area, fixed offshore wind turbines were either not technically suitable or too expensive to install and so the focus was put on onshore wind parks (IEA, 2004b, 2005, 2006a).

In large part due to a conducive policy environment, the adoption of wind turbines in Portugal saw a significant increase after 2004 (J. R. Ferreira & Martins, 2009; IEA, 2005, 2009a; Nunes, 2018), as observed in Figure 6.3. Between 2004 and 2009, more than 500 MW of wind capacity was installed every year (IRENA, 2013b). According to the definition used here, in 2004 a positive tipping point was then crossed. In this section, the period from 2000 until 2007 is analysed, with 2000-2004 being representative of the enabling phase and 2004-2007 of the beginning of the accelerating phase.



Figure 6.3: Wind energy in Portugal from 2000 to 2021 (Costa et al., 2021; IRENA, 2013b). The detailed data is given in Table A.1

One of the most interesting aspects of the diffusion curve of wind energy in Portugal over the last two decades is its S-shaped curve (see Figure 6.3), which resembles the curves found in tipping point literature, namely Figure 2.1. The reason for the flattening out of the curve after 2011 is in this case, however, a consequence of the country's economic crisis which was followed by regulatory instability

and less favourable support schemes in the wind energy market (Frontier Economics, 2013; IRENA, 2013b). In different economic conditions, perhaps the total installed capacity of wind energy would have stabilised at a higher value. Nevertheless, despite of the unfavourable conditions of the 2010s, in the year of 2022 renewable energy sources contributed to 62% of electricity generation (IEA, 2024a) and wind energy in specific generated 28% of all electricity (IEA, 2024b).

From 2000 until 2007, all eight actor groups identified in Section 4.3.1 were active in the development and roll-out of wind energy in Portugal. Figure 6.4 portrays how these actors interacted with each other and with the technology. While the grey blocks and black labels correspond to the elements already presented in Geels and Ayoub (2023), (see Figure 4.3), the orange blocks and labels are interactions that were expected based on general literature (see Figure 4.4). The blue labels are additions made solely according to the case study findings. An explanation of each arrow label is provided below, where the colour of the titles follow the convention of the arrow labels in Figure 6.4.



Figure 6.4: Improved feedback loop model based on the win energy in Portugal case study for the period of 2000 - 2007

Feedback 1 & Feedback 2: Technology Firms \leftrightarrow SET \leftrightarrow Adopters

Investment in R&D and production: During the period analysed, there were tower and electrical equipment manufacturers but there were no wind turbine manufacturing facilities in Portugal, despite a strong interest from the manufacturers side to establish production units in the country (IEA, 2004b, 2006a, 2006b). The main wind turbine manufacturers were Enercon, Vestas, Gamesa, Nordex, GEWE, Izar Bonus, Repower, Neg Micon, Ecotecnia, Fuhrlaender, Mitsubishi Heavy Industries, Winworld, Nordtank, Bornay, and WinWinD (IEA, 2003, 2005, 2006a). In 2007, Enercon established a factory in the north of Portugal as a result of the industrial cluster promoted by the government (J. R. Ferreira & Martins, 2009; Herman, 2013; IEA, 2008). These turbine manufacturers invested largely on R&D and production themselves, which was reflected in the new, better models released over time. For instance, Bornay, the leader of small wind turbines, offered eight models ranging from 60 to 12,000 W but was already developing higher capacity models, up until 50 kW. Also the capacity of the models produced by Enercon increased over time. Moreover, Gamesa developed new models by optimising the efficiency of their system in accordance with different site conditions (ENERCON, 2024; IEA, 2003).

Value: Over time, wind turbines and the complementary technology got better and started proposing a better value, i.e. a higher performance to cost ratio. As mentioned in the interaction above, turbines got bigger over time. From 2003 to 2004 the average wind turbine capacity of a new turbine increased by 6% (IEA, 2005). Furthermore, during 2003 and 2004 a wind turbine's unit cost reduced steadily (IEA, 2004b, 2005).

Purchases; Revenues and profit: Evidence of increased interest in wind turbines and their associated components were the 7,000 MW of new wind capacity applications received in 2002, which were installed in the following years (IEA, 2004b). For instance, the manufacturer Izar Bonus installed 60.3 MW of new wind capacity in 2002. In the first quarter of 2003 they sold another 52 MW, expecting to install 277 MW during the same year (IEA, 2003). Besides, in 2004 the Finish manufacturer WinWinD started selling beyond Finland, exporting to France, Portugal, and Sweden (IEA, 2005). This shows increased interest and purchases by adopters, which is translated into higher revenues and profits for technology firms.

Feedback 3: Policymakers \leftrightarrow Technology Firms

Lobbying: As confirmed by interviewee 2, lobbying from national technology firms was not a relevant interaction during this period due to the lack of national firms in the sector. However, there was lobbying at the European level that ultimately affect the landscape in which wind develops in Portugal. The Renewables Directive is cited as a key example of the influence of interest groups on EU environmental legislation. The EU Commission made significant alterations to its final draft proposal, particularly concerning the types of support mechanisms member states could use to increase their renewable energy production (Ydersbond, 2012).

Loans, capital grants and R&D subsidies: During the period analysed Portugal mainly relied on imports for its onshore wind industry (J. R. Ferreira & Martins, 2009; Herman, 2013; IEA, 2008). The main turbine manufacturers exporting to Portugal had manufacturing facilities in Germany (Enercon, Nordex, GEWE, Repower), Denmark (Vestas), Spain (Nordex, Gamesa, GEWE, Izar Bonus, Bornay) (ENERCON, 2024; IEA, 2002, 2006a, 2008). Thus, these technology firms' developments were shaped by loans and capital grants provided by the country/countries where they had manufacturing facilities in, most of them provided even before the year of 2000. Germany

for instance, offered loans and grants to support research, development, and deployment of wind energy technology (BMWi, 2020). In Denmark, for another, grants, subsidies, and low-interest loans were provided by government agencies such as the Danish Energy Agency and the Danish Ministry of Energy. These funds were aimed at promoting innovation, research, and deployment of onshore wind energy technology (IRENA, 2013a). At last, in Spain one significant initiative was the Renewable Energy Promotion Plan, which provided subsidies, grants, and incentives to renewable energy projects, including wind farms (MITECO, 2005).

Foster collaboration: In 2005, as part of the tendering process, a wind energy industrial cluster was created with the aim of bringing together the project developers and local manufacturers, as well as expanding the production capabilities of the country. This industrial cluster collaborated in the establishment of rotor blades, concrete towers, and mechatronics factories. Industry members involved include Enercon, EDP, Finerge, Generge, SIIF and Térmica Portuguesa (J. R. Ferreira & Martins, 2009; IRENA, 2013b; Pinto, 2005). This shows that collaboration was not only fostered among technology firms but also among these and adopters (or project developers).

Feedback 4: Policymakers \leftrightarrow SET

Policies, goals, and coalitions adjustments: For instance, average feed-in tariffs went from a twelve-year support of 0.068 €/kWh in 1999 to 0.085 €/kWh in 2001 for an indefinite length of time. However, the government soon realised that the latter was not economically sustainable and in 2005 reduced the feed-in tariff to a fifteen-year support of 0.074 €/kWh. Therefore, not only the feed-in tariff was adjusted but also the state guarantee for this financial support (P. Ferreira, Araújo, & O'Kelly, 2006; IRENA, 2013b). Besides, the government started seeing wind energy more as a priority in the country's strategy after 2005 (P. Ferreira et al., 2006; IEA, 2006a). For instance, while in 2003 the government set the goal of having 3,750 MW of wind energy by 2010, in 2005 the goal of having 5,100 MW of capacity by 2013 was set (IEA, 2004b, 2006a). Coalitions also changed during this period. For one, in 2003 Portugal joined the IEA R&D Wind team (IEA, 2004b), while in 2005 the country established the creation of a wind energy cluster (IEA, 2006a).

Infrastructure investment: In Portugal, the wind potential is not uniformly distributed, many times being the highest in remote areas of the country without good access to the electricity grid (Barata & Quadrado, 2007). Consequently, in order to facilitate the integration of renewable energy generation, substantial investments were made in the transmission and distribution grid (IEA, 2009a), with the government providing financial support to project developers for the reinforcement of the network (Agência LUSA, 2004; IEA, 2005). From 2001 to 2003, the licence-granting process for grid access was clarified and the administrative procedures were simplified (IRENA, 2013b). Besides, the 2005 tendering process specified the available power injection capacity and reception locations for each grid zone in accordance with the National grid's projected future evolution such that the grid could be developed efficiently (P. Ferreira et al., 2006; IRENA, 2013b). Furthermore, the addition of capacity was facilitated by the fact that in Portugal the responsibility of allocating connection capacity lies only on one transmission and one distribution company (IEA, 2009a).

Regulations and standards: The national action plan aimed at combating climate change and giving top priority to renewable energy sources (IEA, 2004a). Feed-in tariffs and other financial incentives stimulated renewable energy sources, including wind energy (IRENA, 2013b). Despite the efforts to shift to renewable energy, however, one of the main obstacles that projects faced was the long and bureaucratic authorisation procedure to obtain the necessary permits. This could

imply that between the application of the project and the start of its installation four to five years would have passed (IEA, 2004b). In 2004 it was decided to liberalise the electricity market, which meant opening it to every consumer (IEA, 2004a). With the market's liberalisation came also the further simplification of the administrative and licensing processes, which reduced the deployment time to only two years (IEA, 2005).

Feedback 5: Adopters \leftrightarrow Wider Public

Further adoption; Expanded technology's visibility: As wind parks gained popularity, mainly as a consequence of the attractive financial support available, the technology gained visibility. This played a big role in legitimising the technology, but also resulted in a social contagion effect towards energy firms that were still not (as) involved in producing electricity from wind energy. For instance, the development of wind energy in Portugal was largely led by Enersis and Enernova, with the latter being part of the EDP group (Climate Chance, 2018; L. Ferreira, 2008). The deep commitment of these energy firms gave a signal to the industry that, among other things, the technology had reached a decent maturity level. While in 2004 Enersis and Enernova had 50% of the market share of wind installed capacity (IEA, 2005), this number dropped to 40% by 2006, demonstrating the growing market share of other, smaller companies (IEA, 2006b).

Feedback 6: SET \leftrightarrow Wider Public

Solution to societal problems; Legitimacy: Wind turbines contribute to the reduction of detrimental emission and residuals (J. R. Ferreira & Martins, 2009; P. Ferreira et al., 2006). Wind energy contributes to the country's security of supply, which at the time was very much dependent on fossil fuel imports (J. R. Ferreira & Martins, 2009). Besides, wind energy promised to contribute to Portugal's economic growth and job creation. A study found that wind energy resulted in five times more jobs for each euro invested compared to other energy generation technologies (J. R. Ferreira & Martins, 2009). These advantages combined with a long pipeline of projects, and the redirection of some of the revenues of wind energy projects towards local municipalities played a big role in increasing the technology's legitimacy and public acceptance (IRENA, 2013b).

Feedback 7: Policymakers \leftrightarrow Wider Public

Public opinion; Strong policy plans and accompanying narratives: The Portuguese wider public had in general very positive views about onshore wind energy, with low local opposition numbers (IEA, 2009a). Apart from considering wind and solar power to be more environmentally friendly than biomass and hydropower, the wider public also view wind projects as an important source of local development (Ribeiro, Ferreira, Araújo, & Braga, 2014). One of the reasons for this, apart form job creation, might have been the government's policy establishing that 2.5% of a project's revenues were to be paid to the municipality where the farm was installed (IEA, 2009a). Perhaps also associated with the latter was the fact that the acceptance of new wind farms was larger among citizens of municipalities that had wind farms, in contrast to municipalities where wind farms had not been installed yet (Ribeiro et al., 2014).

Feedback 8: Technology Firms \leftrightarrow Academic & Research Institutes

Funding: During this period there were barely any technology firms established in Portugal, which meant that the R&D financing flowing from these firms to national academic and research institutes was limited. Nonetheless, a few technology firms belonging to the wind industry have provided R&D financing to academic and research institutes through consultancy contracts (IEA, 2008).

Knowledge transfer: The National Institute of Engineering, Technology and Innovation (INETI) created a wind atlas of Mainland Portugal as well as a simplified calculation sheet to compute a project's economical feasibility as a function of the investment. This resulted in more sales for technology firms after 2001 since project developers were able to secure financing more easily (Barata & Quadrado, 2007; IEA, 2004b; IRENA, 2013b). This programme allowed, for instance, to provide REN, the Portuguese Transmission System Operator (TSO), with wind energy production forecasts (IEA, 2004b). Besides, the software *VENTOS* was developed by the Engineering Faculty of the Porto University (FEUP), the Mechanics Engineering and Industrial Management Institute (INESC) and the research centre for Wind Energy and Atmospheric Flows as a tool to simulate the "wind draining behaviour on complex soils with or without arborisation" (Barata & Quadrado, 2007). Finally, well-educated professionals graduated from Portuguese engineering universities and integrated technology and energy firms. Some of the most relevant universities in this field were Higher Technical Institute (IST), FEUP, and Porto Higher Engineering Institute (ISEP). For instance, during the period analysed IST started offering a wind energy course (IST, 2024), while ISEP established a sustainable energy Masters programme (ISEP, 2024).

Feedback 9: Policymakers \leftrightarrow Trade Associations

Consultation; Foster collaboration: Policymakers approach trade associations to gauge the industry's current state and that way be able to release effective policies (Delicado et al., 2013; Sintrão, 2005). In return, trade associations represent their members next to policymakers, making sure their concerns are heard. This is one way of fostering collaboration between their members and policymakers. Nonetheless, collaboration exists at other levels and is a two-way street, accompanied by continuous dialogue. For instance, APREN, the Portuguese Association for Renewable Energies, helps national and international policymakers draft energy policies for Portugal (APREN, 2024c). WindEurope, on the other hand, is involved in coordinating international policy, as well as in developing policy positions related to the industry's concerns. Through their lobbying efforts, a favourable legal framework is created for their members (WindEurope, 2016a).

Feedback 10: Technology Firms \leftrightarrow Financial Institutions

Long-term certainty; Financing solutions: Governmental support AND an extensive projects pipeline improved the industry's long-term market stability over time. The consistent nature of the financial incentives enabled investors to anticipate returns on investments reliably. Furthermore, the comprehensive wind resource assessment conducted by INETI that produced a national wind atlas helped in computing a project's returns. These elements collectively facilitated greater access to financing for wind initiatives, sparking an increase in new projects after 2000. This was the case until the economic crisis of 2010-2014 (IRENA, 2013b). For instance, the banking syndicate composed by BBVA, BES Investment, Caixa BI, and Millennium bcp Investment financed Enersis wind energy projects in €806 million (TVI Notícias, 2006). During this period wind energy projects in Portugal were also financed by the European Investment Bank (EIB, 2007).

$\textbf{Policymakers} \rightarrow \textbf{Adopters}$

Purchase subsidies: One of the financial schemes that supported renewable projects between 2001 and 2005 was the Incentive Scheme for Rational Use of Energy (IRENA, 2013b). Furthermore, in 2007 the government released the PRIME/MAPE subsidy programme which funded more than 100 projects (IEA, 2008).

Feed-in tariffs: Between 2000 and 2007 the government used a feed-in tariff system as a means of promoting renewable electricity generation. This mechanism has the same effect as subsidies (IEA, 2004a). In 1999 the feed-in tariff started being calculated according to a formula that took into account the inflation rate, the avoided costs of investing, operating and maintaining a conventional power plant, and the avoided CO₂ emissions costs (IRENA, 2013b). The average feed-in tariff corresponded to 0.068 €/kWh and was paid during a maximum period of 12 years (P. Ferreira et al., 2006). In 2001, the feed-in tariff started differentiating between sustainable energy technologies, with wind receiving 0.085 €/kWh indefinitely (P. Ferreira et al., 2006; IRENA, 2013b). This was followed by another revision in 2005, according to which the tariff was reduced to 0.074 €/kWh during a maximum period of 15 years (P. Ferreira et al., 2006; IEA, 2009a; IRENA, 2013b).

Tenders: In 2005, the government released a winder energy tender for the allocation of 1,800 MW in specific points of the grid. The tendering was structured in three phases. During the first phase, occurring in 2006, the ENEOP consortium built 1,200 MW. The Ventiveste consortium then built 400 MW in 2007, while other small projects were responsible for the installation of 200 MW during the next phase (P. Ferreira et al., 2006; IEA, 2006b, 2009a; IRENA, 2013b). This instrument was one of the main drivers of wind energy in Portugal after 2005 (IRENA, 2013b).

(Binding) targets: Between 2000 and 2007, Portugal had targets not only concerning the percentage of electricity generated through renewable energy sources, but also wind energy in specific. In 2001, the E4 Programme (Energy Efficiency and Endogenous Energies) was launched. This programme set the goal of producing 39% of electricity via renewables as of 2010, which was later raised to 45% (IEA, 2004a; IRENA, 2013b). Besides, in 2003, the government's Cabinet Resolution set the goal of having 3,750 MW of installed wind energy by 2010 (IEA, 2004b). This was followed by a new target set in 2005 of having 5,100 MW of wind capacity by 2013 (IEA, 2006a).

Foster collaboration: The wind energy industrial cluster created in 2005 as part of the tendering process fostered collaboration between the project developers and local manufacturers (IRENA, 2013b). The cluster's members included Enercon, EDP, Finerge, Generge, SIIF and Térmica Portuguesa (J. R. Ferreira & Martins, 2009; Pinto, 2005), which demonstrates that several adopters (or project developers) were involved. Furthermore, the government also promoted cooperation between the municipalities where the wind turbines were installed and power producers. This was a result of the fact that part of the revenue of the wind projects was paid to local municipalities, increasing the acceptance of the projects (IRENA, 2013b).

$\textbf{Policymakers} \rightarrow \textbf{Financial Institutions}$

Public financing departments or businesses: As mentioned in *Feedback 10*, financial institutions provided financing for wind energy in Portugal during the period being analysed. Some of these institutions were public, thus created by national or international policymakers with the intention of financing the adoption of the technology. One of these was Caixa BI, while the other one was the European Investment Bank (EIB, 2007; TVI Notícias, 2006).

$\textbf{Policymakers} \rightarrow \textbf{Academic \& Research Institutes}$

R&D subsidies: In Portugal, R&D was mainly conducted at and financed by academic and research institutes (IEA, 2008), with public R&D funding being the lowest of all IEA members as a percentage of the country's GDP (IEA, 2009b). Even though during this period there was no governmental programme specifically funding R&D in the wind energy sector, as a part of the Ministry of Economy and Innovation, INETI was partially sponsored by the government (IEA, 2004b). INETI has been one of the main wind energy researchers, but other names include the FEUP and the Research Centre for Wind Energy and Atmospheric Flows (IEA, 2008). Some wind-related research was, nonetheless, funded by the government through mechanical, electrical, and energy engineering grants (IEA, 2005).

Foster collaboration: Policymakers fostered collaboration among academic and research institutes. One significant example was Portugal's participation on the IEA R&D Wind Executive Committee. Portugal became a member in 2003, being represented by INETI (IEA, 2004b). Portugal was especially active on a multi-national programme on creating dynamic models of wind farms for studies about the power system, which involved sharing both measurement data and model descriptions with other members (IEA, 2004b). However, other tasks include for instance reporting on the R&D and roll-out figures of wind energy in Portugal (IEA, 2002).

Academic & Research Institutes \rightarrow SET

Research and test: As highlighted in *feedback 8*, INETI, FEUP, INESC and the research centre for Wind Energy and Atmospheric Flows were active in performing wind-related research. INETI created a wind atlas of the country, while the remaining three worked on a software called *VENTOS* (Barata & Quadrado, 2007). Other research initiatives that contributed to the further development of the technology and the industry related to reducing the losses and improving the voltage profiles through reactive power control, as well as related to the stability of the power system (IEA, 2005).

Trade Associations \rightarrow Adopters & Technology Firms

Foster collaboration: For one, APREN, founded in 1988, is the largest trade association in Portugal for renewable energies, having to date 34 Portuguese members that are wind energy producers, despite having non-producer members too. Overall, APREN represents more than 90% of the installed renewable electricity capacity in the country. The association has the goal to coordinate and represent their members within the Portuguese electricity sector, namely next to the government and respective ministries (APREN, 2024b, 2024c). As collaboration and knowledge sharing activities for their members, APREN organises summits, webinars, and debates (APREN, 2024a). For another, the European wind trade association WindEurope has not only large technology firms and adopters such as ENERCON, Vestas, EDP and Galp as members, but also 10 industry members that only operate in Portugal (WindEurope, 2016b). WindEurope represents and lobby for a variety of members, to whom they offer learning and networking opportunities. For instance, they organise workshops, seminars, exhibitions, and conferences to exchange expertise on different topics (WindEurope, 2016a) *Lobbying:* Despite advocating for renewable energy, environmental NGOs ensured new projects complied with high environmental standards by maintaining a continuous dialogue with national and European policymakers (Delicado et al., 2013; Quercus, 2024). The stringent requirements set by these organisations could result in up to five years to issue a permit for a wind energy project, mainly due to lack of trustworthy data to analyse the project's environmental impacts. This was in fact one of the main contributors to the long and bureaucratic process of obtaining the necessary installation and operation permits for wind farms (IEA, 2004b, 2005). Some of the NGOs involved include Quercus, the Nature and Biodiversity Conservation Institute, the Regional Hydrographic Administrations, and the League for the Protection of Nature (Delicado et al., 2013). Additionally, the Portuguese Society for the Study of Birds, highlighted in 2005 that Portugal did not have any strategy concerning the Environmental Impact Assessments for new wind parks, which ultimately negatively affected the environment and the project developers (Travassos et al., 2005).

$\textbf{NGOs} \rightarrow \textbf{Wider Public}$

Public awareness campaigns: Apart from representing the wider publics next to the policymakers, NGOs raise the citizens' awareness both concerning the positive aspects of wind energy and their potential negative consequences to e.g., life quality (Delicado et al., 2013). Quercus, for instance, is dedicated to raising awareness among and providing information to the citizens. In their various projects and campaigns, they utilise dissemination tools and rely on several partnerships with relevant market actors (Quercus, 2024).

6.3. Electric Vehicles in the UK (2015 - 2021)

The adoption of EVs in the UK significantly increased in the previous years, as observed in Figure 6.5. In January 2024 the millionth BEV was driven on British roads, implying that despite their greater initial cost, EVs are becoming more common and attractive. In fact, 14.7% of all vehicles sold during that month were totally electric (CAR Magazine, 2024). Figure 6.6, shows that while initially PHEVs were more famous among adopters, BEVs gained popularity in the previous years. However, it should be noted that as of 2022 EVs had only 2.8% of cars' market share in the UK (IEA, 2022).

After a clear change in actor perspectives and attitudes after 2015, a marked acceleration can be seen after 2019, marking a socio-technical positive tipping point (Geels & Ayoub, 2023). In order to understand the dynamics that led to this tipping point to be crossed, it is relevant to look at the enabling phase and at the start of the accelerating phase (Lenton et al., 2023). Therefore, the focus of this section lies on the period between 2015 and 2021.

Several actors were involved in the roll-out of electric vehicles in the UK. Figure 6.7 shows a representation of the most relevant actors and how these interacted with each other and with the technology being deployed. The grey blocks and black labels correspond to interactions identified by Geels and Ayoub (2023). The orange blocks and labels are interactions that were identified based on general literature (see Figure 4.4), and the blue labels are interactions added when analysing the case study.



Figure 6.5: Stock of electric vehicles (BEVs and PHEVs) in the United Kingdom from 2013 to 2022 (IEA, 2022). The detailed data is given in Table A.2



Figure 6.6: Stock of plug-in hybrid electric vehicles (PHEVs) and battery electric vehicles (BEVs) in the United Kingdom from 2013 to 2022 (IEA, 2022). The detailed data is given in Table A.2

Before going into the details of the diagram, a clarification should be given concerning two of the actor groups. For one, *technology firms* refers to manufacturers of charging points, batteries, and vehicles or vehicle components and the respective raw materials, as well as charging infrastructure providers (Wills, 2020). For another, *adopters* are both individual end-consumers and companies who wish to have a more environmentally-friendly car fleet (CAR Magazine, 2024).

A second note concerns the fact that, unlike for the diffusion of wind energy, financial institutions did not play a key role in promoting the development and adoption of electric vehicles in the UK. This conclusion was taken after consulting several sources on the topic, which only mention the several public financial incentives as an enabler. It is true, however, that private financial institutions offer incentives such as the incentives for businesses offered by Barclays Bank (Barclays Bank, 2024) or the one for individuals provided by Lloyds Bank (Lloyds Bank, 2024). Nevertheless, incentives like these did not seem to be determining for the positive tipping point to be triggered as information concerning such incentives



Figure 6.7: Feedback loop model showing the relevant actors and respective roles in the diffusion of electric vehicles in the UK for the period of 2015-2021

was not included in any documents having an overview of the main developments of EVs in the UK. Rather, information concerning private financial incentives was only found due to searching specifically for *leasing options EVs UK* and adding the name of specific banks that operate in the UK to the search.

Below, an explanation is given for the interactions in Figure 6.7, starting with the closed feedback loops and following with the interactions that are not part of a specific feedback loop. Each explanation is preceded by a small title corresponding to the arrow label in Figure 6.7, following the same colour code.

 $\textbf{Feedback 1\& Feedback 2: Technology Firms} \leftrightarrow \textbf{SET} \leftrightarrow \textbf{Adopters}$

User feedback: Automakers invested in certain technology advancements due to the received user feedback. For instance, the range anxiety concerns of (potential) adopters led automakers to invest in R&D focused on improving battery efficiency, size and capacity (Geels & Ayoub, 2023).

Investment in R&D and production: Although between 2010 and 2015 more incumbent automakers started to diversify into electric vehicles, it was after 2015 that a clear expansion and diversification of EV models was observed. This was accompanied by large investments in building new skills and factories (Geels & Turnheim, 2022). Manufacturer's focus lied greatly on battery technology improvements, but manufacturers and providers of charging devices were also involved in pushing for EVs (Wills, 2020). This change of attitude was partly driven by the Diesel scandal of 2015, which led to an EV innovation race in the years to follow (Geels & Ayoub, 2023).

Value: The automakers' investment on battery improvements resulted in a nearly 1.5 times higher average driving range of BEVs in 2019 compared to 2015, with further increases in the subsequent years. Besides, enhancements were also made concerning the charging times, which were reduced due an increase in the average charging power (Geels & Ayoub, 2023). Furthermore, learning-by-doing and scale economies allowed the price of Li-ion battery packs, the primary cause of high EV pricing to drop by 63% between 2015 and 2019 (Geels & Turnheim, 2022).

Purchases; Revenues and profit: The trend observed since 2015 of higher performance and lower cost of EVs - meaning an overall higher value - together with the expansion of public charging devices increased the attractiveness of EVs and decreased user concerns (Geels & Ayoub, 2023). This meant an increase in purchases and therefore in automakers' revenues and profits after 2015, increasing the latter's confidence and hence investment in EVs (Geels & Turnheim, 2022).

Feedback 3: Policymakers \leftrightarrow Technology Firms

Lobbying: Lobbying, among other factors, meant policies became more deployment-oriented after 2015. Nevertheless, despite the automakers' investments in EVs, car companies and their trade associations still lobbied against UK and EU climate legislation, including the phase-out of ICE vehicles (ICEVs), in an effort to slow down the transition (Geels & Ayoub, 2023).

Loans and capital grants: The 2017 Clean Growth Strategy announced subsidies and capital grants for stimulating automakers to build EV and battery manufacturing plants (Geels & Ayoub, 2023). Moreover, the 2017 Faraday Battery Challenge budgeted £541 million to be spent between 2017 and 2025 on, among other things, the manufacturing capability scale-up of batteries (The Faraday Institution, 2024; UK Research and Innovation, 2023)². Further, the Supplier Competitive-ness Improvement programme, outlined in the 2018 Automotive Sector Deal, aimed at enhancing manufacturing supply chain companies' productivity and competitiveness (SMMT, 2018).

R&D subsidies: The funds allocated to the 2017 Faraday Battery Challenge were also to be allocated to battery-related R&D activities (UK Research and Innovation, 2023). Besides, the 2018 Automotive Sector Deal also stated that the country's expenditure in R&D in the automotive sector was to be increased to 2.4% of the country's GDP by 2027 (BEIS, 2018).

Foster collaboration: The 2017 Faraday Battery Challenge involved several partners, one of which the Faraday Institution. This research centre focused on fostering collaboration between universities and industry to establish the UK as the premier location for novel electrical storage technology R&D (UK Research and Innovation, 2023). The collaboration among the industry and the government was also made clear in the 2018 Automotive Sector Deal (SMMT, 2018), which had the goal of aligning these two entities when it came to the EV transition (Geels & Ayoub, 2023).

²Note that there is an inconsistency between the information found on these official sources and the information provided in Geels and Ayoub (2023), which stated that only £330 million were allocated to this project.

(Binding) targets: To start with, EVs were seen as a crucial part of the country's industrial strategy by the 2017 Clean Growth Strategy, setting out a vision for the future of this technology in the automotive sector (Geels & Ayoub, 2023). Particularly, it announced the plans to phase-out new petrol and diesel cars and vans by 2040 (BEIS, 2017), which three years later was pushed to 2030, with all new cars needing to have zero-emissions after 2035 (Department for Transport & OZEV, 2020). These targets set a direction for the carmakers wanting to sell cars in the UK in the next decade, therefore setting out a course for an EV mass market (Geels & Ayoub, 2023).

Training: Beginning with a £64 million investment for digital and construction training, the 2018 Automotive Sector Deal established a National Retraining Scheme to assist individuals to re-skill according to the country's vision for zero-emission transportation (BEIS, 2018).

Feedback 4: Policymakers \leftrightarrow SET

Policies, goals, and coalitions adjustments: For instance, between 2015 and 2021 the funding allocated to R&D investment, infrastructure investment or to the Plug-in Car grant was updated over time, as were the targets for the phase-out of non-electric vehicles (BEIS, 2017, 2018; Geels & Turnheim, 2022). As the technology evolved the alliances among stakeholders changed too. For instance, new collaboration opportunities appeared after the 2017 Faraday Battery Challenge (UK Research and Innovation, 2023) and the 2018 Automotive Sector Deal (SMMT, 2018). Furthermore, with the technology maturing and new technology firms starting their involvement in the EV industry, trade associations saw their number of members increase, which means new alliances were formed (Automotive Council UK, 2024a; SMMT, 2018). Concerning feedback effects on goals, the outcomes and impacts of EV adoption influenced policymakers' perceptions, priorities, and long-term policy objectives. For one, between 2015 and 2017 policymakers started perceiving EVs as crucial in their transport decarbonisation strategy (Geels & Ayoub, 2023). For another, possibly due to observing an increasing EV adoption, in 2017 ambitious targets for the phase-out of petrol and diesel cars were set by policymakers (BEIS, 2017).

Infrastructure investment: In terms of direct public infrastructure investment, the On-street Residential Chargepoint Scheme, introduced in 2016, provided funding to local authorities to install on-street EV charging points in residential areas (OZEV, 2016a). Besides, £40 million of funding were allocated in 2016 to the Go Ultra Low Scheme, which promoted eight local authorities across the UK in their EV uptake, including the deployment of EV charging infrastructure (Department for Transport & OZEV, 2020; Fleet News, 2016; OLEV, 2014). The 2017 Clean Growth Strategy further set out investments and requirements for the charging infrastructure, aiming to make it one of the world's premier electric vehicle charging networks (BEIS, 2017). This was followed, in 2018, by the Automotive Sector Deal which announced an investment of £400 million to be made on the charging infrastructure (BEIS, 2018). These investments in the public charging infrastructure resulted in a rapid growth of the number of public charging devices grew by almost 27,000 from 2019 to 2023 (DfT, 2023). This was one of the main factors for the positive tipping point to be crossed, even though further improvements can be made to the charging infrastructure (interviewee 1).



Figure 6.8: Total number of public chargers in the UK from 2015 to 2023 (DfT, 2023). The detailed data is given in Table A.2

Regulations and standards: The European Commission confirmed in 2019 that manufacturers must meet the 2014-established 95 g/km target for the average emissions of their sales after 2020/21. Strict fines were also imposed on businesses that failed to reach this goal, calculated by multiplying the number of cars sold by £83 (€95) for each gram over the limit. Even though Brexit took place in January 2020, the car market is a global market and so these regulations had a large impact on automakers' strategies. Overall, the industry's confidence in EVs increased in the light of such tight climate regulations (including the target of phasing-out new petrol and diesel cars by 2030) (Geels & Turnheim, 2022). Moreover, during this period, EVs did not pay or paid a reduced amount for parking in several cities of the UK, serving as an example of a regulation imposed on car owners that intended to promote the adoption of EVs (Rietmann & Lieven, 2019). Another intervention having a similar goal was the introduction of the London Ultra Low Emission Zone (ULEZ) in 2019 (SMMT, 2019), which specified that BEVs were free from the ULEZ and congestion charges, with PHEVs having to comply to certain standards to be eligible for this benefit (Garratt, 2024). This was a big incentive to switch to an EV since the London congestion charges can reach £2000 yearly (Rietmann & Lieven, 2019). Since then, seven other clean air zones have been established in the UK (Department for Environment, Food & Rural Affairs, 2020).

Feedback 5: Adopters \leftrightarrow Wider Public

Expanded technology's visibility; Further adoption: After 2015, as adoption grew and new policies were established, public attention grew. As a result, public discussions concerning EVs became more favourable and concentrated on topics such as job creation, economic growth, and clean air. Positive discourses fuelled consumers' curiosity and accelerated the uptake of EVs, which truly took off after 2019 (Geels & Ayoub, 2023).

Feedback 6: SET \leftrightarrow Wider Public

Solution to societal problems; Legitimacy: With early discourses covering aspects such as the fact that EVs were quiet, smooth, and allowed to reduce the emissions of the transportation sector, the technology started being framed as offering a solution to lessen fossil fuel dependency

and cut on emissions (Alali, Niesten, & Gagliardi, 2022; Bunce, Harris, & Burgess, 2014). After 2015, increased EV adoption led to more positive public discourses, now also focusing on economic growth and job creation. The positive views increased the technology's legitimacy over time, which improved as the pitfalls of EVs saw technological advancements (Geels & Ayoub, 2023).

Feedback 7: Policymakers \leftrightarrow Wider Public

Public opinion; Strong policy plans and accompanying narratives: As concerns about climate change grew, public pressure contributed to policies becoming more deployment-oriented after 2015. The several policies aimed at promoting clean transportation in detriment of polluting vehicles substantially increased the attention of the wider public to the topic, which ultimately led to positive public debates that legitimised the established policies (Geels & Ayoub, 2023).

Feedback 8: Technology Firms \leftrightarrow Academic & Research Institutes

Funding: A £150 million building was built at the University of Warwick as a partnership between Jaguar Land Rover, Tata and the Warwick Manufacturing Group (interviewee 1). The goal was to create the biggest European automotive R&D centre and lead the way towards electric vehicles (Coventry Live, 2017). As another example, Ford funded a collaborative research project with Loughborough University to advance digital tools for EVs (Loughborough University, 2019).

Knowledge transfer: The UK Battery Industrialisation Centre was one of the organisations that collaborated on the 2017 Faraday Battery Challenge. This research centre is the national battery manufacturing development facility, assisting organisations in scaling up their battery technology into production. Another organisation involved, the Faraday Institution, combined expertise from academia and the industry to put the UK at the forefront of electrical storage technologies (UK Research and Innovation, 2023). These initiatives demonstrate a mutual transfer of knowledge between universities and technology firms, which is shown in Figure 6.7 as a feedback loop between these two actors. It represents that, as these collaborations become stronger and both parties see beneficial results, both the universities and the industry become more open to share more information with each other and collaborate further. Moreover, well-educated professionals from technical and higher education systems then joined different technology firms, helping push forward the EV transition in the UK (BEIS, 2018).

Feedback 9: Policymakers \leftrightarrow **Trade Associations**

Consultation; Foster collaboration: Policymakers consult trade associations to gauge the industry's and market's state. Examples include a public consultation in 2017 concerning vehicle weights, dimensions, and cleaner fuels, and one in 2019 about smart charging (Department for Transport, 2017, 2021). Trade associations such as the Society of Motor Manufacturers and Traders (SMMT) and the UK Automotive Council are involved in lobbying to assure that the industry is well represented at the government (Automotive Council UK, 2024b; SMMT, 2023). Such activities can also be used in a counterproductive way, as was the case when automakers and trade associations lobbied against legislation such as the phase-out of ICEVs, in the hope of delaying the transition (Geels & Ayoub, 2023). However, it is also an opportunity to bring the industry and the government closer, fostering an environment of mutual understanding where all parties work together in the same direction (SMMT, 2018). For example, the Automotive Council was involved in writing the 2018 Automotive Sector Deal (Automotive Council UK, 2024b; BEIS, 2018).

$\textbf{Policymakers} \rightarrow \textbf{Adopters}$

Purchase subsidies: Over the years, there were several purchase subsidies involved, with the intent of promoting demand for ultra low emission vehicles (ULEV), which included electric vehicles. Firstly, the Plug-in Car grant, established in 2011, was still in place from 2015 to 2019, even though it suffered some alterations after 2018. When implemented, this grant assured that 25% of the purchasing costs of a BEV or PHEV were covered, up until £5000. In 2018, it was lowered to £3500 for BEVs (with further reductions in 2020 and 2021) and it was eliminated for PHEVs (Geels & Turnheim, 2022). The Clean Growth Strategy of 2017 further allocated £1 billion to support ULEVs, including £50 million dedicated to the taxi programme, with which taxi drivers could be eligible to up to £7,500 (BEIS, 2017). The 2018 Automotive Sector Deal set aside for the Plug-in Grant an extra £100 million (BEIS, 2018). Furthermore, the Electric Vehicle Homecharge Scheme was implemented in 2014 and lasted until 2022, when it suffered alterations and was renamed to the EV chargepoint grant. The scheme provided up to £350 to EV owners for the installation of home charging points with the aim of making it more convenient and affordable for individuals to charge their vehicles at home (British Gas, 2024). Finally, incentives were also given in the form of a voucher through the Workplace Charging Scheme to businesses wishing to improve their charging infrastructure, ultimately lowering the up-front costs (OZEV, 2016b).

Tax rebates: Implemented in the years prior to 2015 and still in place after 2019, EVs have benefited from tax reductions or exemptions. These include the Vehicle Excise Duty (VED), the Company Car Tax, and the Fuel Excise Duty (FED) (Geels & Turnheim, 2022). Since the VED and the Company Car Tax are calculated based on a car's tailpipe CO₂ emissions, this means that BEVs have more appealing tax reductions than PHEVs. Additionally, the FED also benefits greatly BEV than PHEV owners since it corresponds to the tax imposed on road fuels. Nonetheless, it can be concluded that BEV and PHEV have strong tax benefits compared to a petrol or diesel vehicle both for individuals and businesses (Zapmap, 2023).

$Policymakers \rightarrow Academic \& Research Institutes$

R&D subsidies: The 2017 Faraday Battery Challenge allocated funding to several topics and projects, for which research organisations were eligible (UK Research and Innovation, 2023). The 2018 Automotive Sector Deal increased the total investment in R&D to 2.4% of the UK's GDP until 2027 (BEIS, 2018). Other R&D funding was also provided by Innovate UK and the UK Research and Innovation (established in 2018), of which the Engineering and Physical Sciences Research Council is part (Innovate UK, 2024; UK Research and Innovation, 2024a, 2024b).

Foster collaboration: To bring the UK to the forefront of electrical storage technologies, the 2017 Faraday Battery Challenge brought together different organisations, including the two research centres UK Battery Industrialisation Centre and Faraday Institution. The latter was responsible for combining knowledge between the industry and academia (UK Research and Innovation, 2023).

Promote education: With a drive to become a world leader in the EV transition, in 2018 the government invested £406 million to create a world class technical education system. This was meant to fight the shortage of skills in science, technology, engineering and maths through allocating the money to maths, digital and technical education (BEIS, 2018).

Academic & Research Institutes \rightarrow SET

Research and test: Academic and research institutes actively contributed to the development of electric vehicles and the surrounding technology. The Warwick Manufacturing Group aimed at being the largest R&D centre for EVs, with Loughborough University focusing on digital tools (Loughborough University, 2019; WMG, 2024). Moreover, the UK Battery Industrialisation Centre is the national battery manufacturing development facility, with the Faraday Institution focusing on electrical storage technologies (UK Research and Innovation, 2023).

Trade Associations \rightarrow Adopters & Technology Firms

Foster collaboration: Trade associations such as the SMMT, the Automotive Council, and Recharge UK have several members, including technology firms (both automakers and supply chain firms), government, and research institutes (Automotive Council UK, 2024a; Recharge UK, 2024; SMMT, 2018). Such associations have been very active in promoting a clean future for UK's automotive sector. SMMT has been actively involved in assisting businesses who want to increase the competitiveness of their products and services by developing their workforce. Additionally, they have assisted in matching purchasing organisations with UK skills and have encouraged the sector to adopt a uniform strategy. Such advantages and collaboration opportunities have led, over the years, more parties to join trade associations such as SMMT, furthering fostering collaboration and discussion (SMMT, 2018). Moreover, Recharge UK is the biggest forum for electric vehicles in the UK, connecting over 100 organisations dedicated to improving the future of EVs in the country (Recharge UK, 2024). Furthermore, some associations such as the Electric Vehicle Association England represent England's current and future EV drivers, involving the adopters in the process (EVA, 2024).

$\textbf{NGOs} \rightarrow \textbf{Policymakers}$

Lobbying: NGOs have been very supportive of electric vehicles, actively advocating for them and stating that they play a key role in fighting climate change (Campaign for Better Transport, 2022). For instance, Greenpeace UK wrote that the phase out of petrol and diesel cars by 2030 consisted of a big win (Greenpeace, 2020). This enthusiastic support is also seen through the different campaigns, publications released, and reports written, which advocate for certain policies in the interest of the industry and citizens that could accelerate the adoption of EVs (T&E, 2019).

$\textbf{NGOs} \rightarrow \textbf{Wider Public}$

Public awareness campaigns: Several NGOs have campaigned for cleaner vehicles in the UK, covering topics such as climate change and air pollution. These include Campaign for Better Transport, Greenpeace UK, and Transport & Environment. Focusing on promoting the adoption of electric vehicles, these NGOs have been involved in organising campaigns and spreading petitions. For instance, Greenpeace UK blocked the entrance to the Volkswagen office, demanding they solely sell electric cars and offering advice to citizens (Greenpeace, 2018). Transport & Environment has also been very active in campaigning for electric vehicles in Europe, but also in the UK. For example, they publish an annual report where a campaign is always directed towards promoting electric vehicles. Besides, they have been involved in promoting petitions and organising street rallies, as well as advocating for new emissions controls (T&E, 2019).

6.4. Electric Vehicles in the Netherlands (2015 - 2021)

As observed in Figure 6.9, there was a significant surge in EV adoption in the Netherlands after 2018, marking the positive tipping point. After some years in which EV diffusion seemed to have stalled due to fluctuating tax rates (Hall, Wappelhorst, Mock, & Lutsey, 2020), adoption markedly picked up after 2018, especially that of BEVs, as observed in Figure 6.10. Compared to 2018, in 2022 there were 3.7 times more EVs on Dutch roads, corresponding to an increment of 270%.



Figure 6.9: Stock of electric vehicles (BEVs and PHEVs) in the Netherlands from 2013 to 2022 (IEA, 2022). The detailed data is given in Table A.2



Figure 6.10: Stock of plug-in hybrid electric vehicles (PHEVs) and battery electric vehicles (BEVs) in the Netherlands from 2013 to 2022 (IEA, 2022). The detailed data is given in Table A.2

Despite EV's having a market share of only 5.6% in 2022 (IEA, 2022) in the Netherlands, the country is a key player in the European EV market, consistently ranking among the top countries for EV sales shares (IEA, 2023b). In particular, the share of BEVs in total new car sales rose from 5.4% in 2018 to 20.3% in 2020 (Paradies, Usmani, Lamboo, & van den Brink, 2023). The Dutch government's ambitious goals

and supportive policies (including generous subsidies) significantly contributed to this EV revolution and to the Netherlands maintaining its front runner status within Europe (Transport & Environment, 2016).

Cities like Amsterdam played a pivotal role, surpassing national average EV penetration rates (ICCT, 2023). The Dutch dynamic EV landscape, characterised by shifting market shares between BEVs and PHEVs, reflects both consumer preferences and evolving policy landscapes (Hall et al., 2020). The transformation witnessed underscores a collective commitment towards cleaner, sustainable transportation systems, setting the stage for continued innovation and progress in the years to come.

Different actors were active in developing, promoting, and adopting EVs in the Netherlands. Figure 6.11 illustrates the most relevant actor groups and their roles during the period analysed (2015 to 2021). The grey blocks and black labels correspond to elements already identified by Geels and Ayoub (2023). The orange blocks and labels were found on the case study but were already expected based on literature (see Section 4.3). The blue labels are additions made solely based on the case study findings. For a definition of the actor groups *technology firms* and *adopters* consult Section 6.3.



Figure 6.11: Feedback loop model showing the relevant actors and respective roles in the diffusion of electric vehicles in the Netherlands for the period of 2015-2021.

The period from 2015 to 2021 was studied. In 2015 the government showed a deep commitment to selling solely zero-emission cars in the near future (Transport & Environment, 2016) and large charging infrastructure investments were made after 2015 (IEA, 2023b). The Paris Agreement was also signed
in 2015, setting GHG reduction targets (UNFCCC, 2022). For another, in 2021 subsidies suffered updates (Nederland Elektrisch, 2024b) and zero-emission zones were introduced (ICCT, 2023).

Like the diffusion of EVs in the UK, financial institutions did not play a key role in the uptake of EVs in the Netherlands. Financial options from financial institutions existed but this information was not found in any documents having an overview of the main developments of EVs in the country. Rather, it was only found after searching specifically for *leasing options* + *electric vehicles* + *Netherlands* and adding the name of specific banks that operate in the Netherlands to the search. Therefore, this actor group was left out of Figure 6.11. Each interaction depicted in Figure 6.11 is explained below, starting with the feedback loops and then the interactions outside specific loops. Each explanation is introduced by a small title corresponding to the arrow label in Figure 6.11, in the corresponding colour.

$\textbf{Feedback 1\& Feedback 2: Technology Firms} \leftrightarrow \textbf{SET} \leftrightarrow \textbf{Adopters}$

User feedback: Most automakers are active in many countries, so the improvements they invested in due to user feedback were probably a result of observations made in different markets. It is not easy to understand which developments were due to an event happening in the Netherlands. Thus, this interaction is in essence the same as for EVs in the UK (see Section 6.3). Feedback received from (potential) EV adopters was used by automakers in a constructive way, such that improvements were made to the technology. As an example, concerns about range anxiety were addressed through R&D by improving battery efficiency, size and capacity (Geels & Ayoub, 2023).

Investment in R&D, production, *and infrastructure*: Investment in EV-related R&D and production by technology firms relates to what was observed in the UK due to the fact that car companies act in an international context (Geels & Ayoub, 2023). The Diesel scandal of 2015 led to a change in attitudes, which resulted in an EV innovation race in the years to follow (Geels & Ayoub, 2023). After 2015, the number of EV models available increased and the offer became more diversified. Besides, new skills and factories were developed through large investments made by technology firms (Geels & Turnheim, 2022). Investments in technology advancements were made not only by car and battery manufacturers (on e.g., battery improvements) but also by charging infrastructure providers (Wills, 2020). In the Netherlands, some companies also invested in expanding the country's charging infrastructure (interviewee 3). For instance, since 2012 Fastned has been expanding its network of fast chargers (Fastned, 2019). Besides, Tesla has partnered with the Van der Valk Hotels and installed fast chargers in their premises (Van der Valk Hotels, 2024).

Value: From 2015 to 2021, the value proposition of EVs experienced improvements, driven by advancements in technology and market dynamics. Automakers enhanced battery capacity, size, efficiency, and range. The charging power also increased during this period and more charging options became available. All together, it meant that EVs were more practical and convenient to own (Geels & Ayoub, 2023; Milieu Centraal, 2024). Moreover, cost reductions in Li-ion battery packs contributed to making EVs more affordable and thus attractive (Geels & Turnheim, 2022).

Purchases; Revenues and profit: With the performance of EVs and their infrastructure increasing and their cost reducing, adopters' concerns were decreased and adoption became more appealing (Geels & Ayoub, 2023). As a result, purchases rose and the number if EVs on Dutch roads went from 87,400 in 2015 to 390,000 in 2021 (IEA, 2022; Milieu Centraal, 2024). Because of this trend, over time technology firms had higher revenues and higher profit from their EV segment, which increased their confidence and investment in the technology (Geels & Ayoub, 2023).

Feedback 3: Policymakers \leftrightarrow Technology Firms

Lobbying: Technology firms mainly lobbied via trade associations such as RAI, BOVAG, VNA, EVO, TLN, and VNO (VNA, 2024; VZR, 2022) (see feedback 9). As an example, RAI, BOVAG, and VNA lobbied against the system to determine a car's additional tax as one more or one less gram of CO₂ could make the difference between a higher or lower car sales number (VZR, 2022).

Loans, capital grants, and R&D subsidies: The Dutch government and the European Commission provided several financial incentives to promote innovation and develop the market for electric vehicles, which technology firms could apply to. This included both research and demonstration projects (RVO, 2017a). For instance, the Research and Development Promotion Act, which exists since 1994, offers a tax reduction to R&D projects (RVO, 2015). Since 2014 there was also the innovation incentive Region and Top Sectors for small and medium enterprises (SMEs) (RVO, 2023b) and the Demonstration of Energy and Climate Innovation scheme (RVO, 2023a), and since 2017 the Subsidy Scheme for Demonstration of Climate Technologies and Innovations in Transport (RVO, 2022a). During this period there were also other European-wide support schemes. Namely, the Interreg programme has existed since 1990 (European Commission, 2020; RVO, 2017b) and the LIFE subsidy programme since 1992 (European Commission, 2022; RVO, 2017c). Moreover, the Eurostarts programme initiated in 2007 and targets SMEs in specific (RVO, 2011), while the Connecting Europe Facility Transport scheme, existent since 2014, has the goal of improving the transport network (RVO, 2014).

Foster collaboration: Dutch policymakers strategically fostered collaboration among technology firms and between these and other actors to drive advancements in EVs and their charging infrastructure. A pivotal initiative was the Formula E-Team, established in 2016, forming a publicprivate partnership involving the technology firms, research institutes, and the government itself (Nederland Elektrisch, 2024a). This collaboration extended to various working groups focused on different aspects of EV adoption, such as the consumer market, light EVs, PHEVs, the charging infrastructure, and communication, among others (Nederland Elektrisch, 2019). Furthermore, the government has actively engaged in other public-private partnerships with local governments, firms, trade associations, and research institutes, as evidenced by the 2019 Climate Agreement and the Dutch National Charging Infrastructure Agenda (Nationale Agenda Laadinfrastructuur, 2022). These partnerships highlight a concerted effort to achieve climate goals and promote sustainable mobility through collaborative innovation and policy frameworks.

(Binding) targets: Policymakers have set ambitious and binding targets aimed at technology firms concerning EVs and their adoption. One of the most significant targets is that the Dutch government aims for all new vehicles sold in the country to be zero-emission vehicles by 2030 (Deuten, Vilchez, & Thiel, 2020; Hall et al., 2020; Klimaatakkoord, 2021; Nationale Agenda Laad-infrastructuur, 2022; Paradies et al., 2023; van der Kam, Meelen, van Sark, & Alkemade, 2018). Some cities have, however, even more ambitious targets. For instance, Amsterdam aims at only having emissions-free transportation in the city by 2030 (Hall et al., 2020). Additionally, national and European initiatives like the 2019 Climate Agreement and the 2021 Fit for 55 package set specific goals set for public transport and passenger cars. The EU has set in as early as 2015 CO₂ emission targets for new passenger car fleets (Agency, 2016; Paradies et al., 2023), while the Fit for 55 package set the goal of reducing by 55% the country's GHG emissions compared to 1990 (Nationale Agenda Laadinfrastructuur, 2022). Further, goals for the development of the charging infrastructure were set by the National Charging Infrastructure Agenda such that all electric vehicles had access to charging devices (Nederland Elektrisch, 2022).

Feedback 4: Policymakers \leftrightarrow SET

Policies, goals, and coalitions adjustments: After policies are implemented, their effectiveness is evaluated and, at times, the policy suffers adjustments (Geels & Ayoub, 2023). For instance, at first the incentives given by the Dutch government promoted PHEVs more than BEVs (Ashkrof, de Almeida Correia, & van Arem, 2020), leading to the adoption pattern observed in Figure 6.10. However, upon realising that PHEV owners only used their vehicles' electrical power 35% of the time, the purchase incentives given to PHEVs were reduced (Transport & Environment, 2016). Continuous adjustments were also performed to the taxation of PHEVs and BEVs, and to the purchase subsidies (Deuten et al., 2020; Hall et al., 2020). Besides, as the technology evolved, goals evolved too. Initially the focus was on increasing the market share of PHEVs. Over time it shifted to actually phasing out PHEVs. This was a consequence of their limited capacity to lower emission, associated with the fact that the battery capacity of BEVs was improving (Ashkrof et al., 2020). At the same time, this created more confidence in the technology and led to for instance the 2017 target of having 100% sale of zero-emission vehicles by 2030 (Deuten et al., 2020).

Infrastructure investment: The Netherlands has been one (if not the) leader when it comes to their charging infrastructure in terms of public chargers per capita, per electric vehicle, and in absolute terms, even though their public infrastructure consists of mostly slow chargers (Hall et al., 2020; ICCT, 2023; IEA, 2023b; Paradies et al., 2023). Indeed, as observed in Figure 6.12, the number of public recharging points raised from less than 10,000 in 2015 to nearly 75,000 in 2022 (Statista, 2023), with Amsterdam accounting most public chargers and new EV registrations (ICCT, 2023). This fast growth was due to effective policies and public investment (ICCT, 2023). Both the 2016 Green Deal and the 2019 Climate Agreement set goals and plans to expand and improve the country's EV charging infrastructure (Klimaatakkoord, 2021; Nederland Elektrisch, 2019; Rijksoverheid, 2020). As measures, for instance, the government and different regions allocated 15 million euros each to construct the national network (Nederland Elektrisch, 2020). Besides, in several Dutch municipalities EV owners can request that a public charger is installed for free in their neighbourhood. (ICCT, 2023; Nederland Elektrisch, 2024b).



Figure 6.12: Total number of public chargers in the UK from 2015 to 2023 (Statista, 2023). The detailed data is given in Table A.2

Regulations and standards: In 2019 the European Commission confirmed that after 2020/21 manufacturers must have an average emission level of 95 g/km in their sales portfolio. If businesses failed to meet this target, they were subjected to stiff fines of £83 (€95) per car sold for each gram over the limit (Geels & Turnheim, 2022). Further, EVs started paying parking fees in many Dutch municipalities after 2018, even when at public charging points (Yellowbrick, 2018). Additionally, LEZs were implemented by some Dutch municipalities and significantly promoted EVs. Since 2018, Amsterdam has an environmental zone that does not allow old diesel buses or taxis into the city, promoting the use of EVs by taxi drivers. Arnhem went one step further in 2019 by banning any diesel car older than 2005 to enter the city. Furthermore, in Utrecht diesel cars with engines older than 2001 are also not allowed to drive in the city centre since 2015, and the same thing goes for Rotterdam since 2016 (Auto Siero, 2021; IEA, 2023b).

Feedback 5: Adopters \leftrightarrow Wider Public

Expanded technology's visibility; Further adoption: As the adoption of electric vehicles increases, the technology's visibility and acceptance seemed to increase too. In 2017, only 17% of the Dutch population had the intention of purchasing an EV in the coming five years. In contrast, in 2021 this percentage raised to 25% (ANWB, 2021).

Feedback 6: SET \leftrightarrow Wider Public

Solution to societal problems; Legitimacy: As the technology evolved, it started being framed more as a solution to some societal problems. Among them, environmental concerns listed high since EVs produce around 35 to 55% less CO₂ during their lifetime compared to an ICEV (ANWB, 2019, 2021; KiM, 2022). Another advantage are the lower operating costs of EVs compared to petrol cars. This is a result of lower charging costs compared to refuelling, lower maintenance requirements, and road tax exemption (ANWB, 2021; KiM, 2022). Concerning a vehicle's total cost of ownership, EVs actually had a lower value on average compared to petrol vehicles (ANWB, 2021). Besides, EVs are not as noisy as their counterparts and are odourless, making them a cleaner solution than fossil-fuel-based cars (ANWB, 2018; KiM, 2022). The better value offered over the years (see *Feedback 1*) and by tackling some car-related societal problems, the legitimacy of EVs increased, leading to higher acceptance of the technology (ANWB, 2019, 2021).

Feedback 7: Policymakers \leftrightarrow Wider Public

Public opinion; Strong policy plans and accompanying narratives: Dutch people are considerate of environmental problems and have been adapting their behaviour towards more sustainable practices such as purchasing EVs. Such concerns and preferences are heard by policymakers, who then create more ambitious climate policies (ANWB, 2019; European Investment Bank, 2022). This feedback between policymakers and the wider public is also visible in other aspects. For instance, the purchase subsidies introduced in 2020 targeted the public concern about high EV prices (ANWB, 2019). Moreover, being low vehicle range another concern, policymakers provided several R&D subsidies to help advance the technology (ANWB, 2019; RVO, 2017a). Lastly, even though the Netherlands has been a global leader in the charging infrastructure, in 2021 27% of the Dutch population still considered it to be too limited. The Dutch government has, however, been making huge investments in the public charging infrastructure (see *Feedback 4*) with the aim of promoting the technology and quieting these uncertainties (ANWB, 2021; ICCT, 2023).

Feedback 8: Technology Firms \leftrightarrow Academic & Research Institutes

Funding: Technology firms also fund academic and research institutes. For instance, the Formula Student Team of TU Delft develops and builds electric racing cars with the help of several business partners, who invest in their project (DUT, 2024). In 2018, the company Lithium Werks announced they would build an R&D centre in Enschede and collaborate with University of Twente to develop the new generation of batteries and flexible energy storage solutions. The project costed more than €100 million, which was funded by different sources (University of Twente, 2018).

Knowledge transfer: Technology firms and academic and research institutes collaborated in several projects, transferring knowledge and expertise among each other. The Formula E-Team was created in 2016 as a public-private partnership that brought together the government, knowledge institutions, and businesses. These parties collaborated on projects concerning knowledge development and sharing about electric transportation and the charging infrastructure (Nederland Elektrisch, 2022). One of these knowledge institutions was ElaadNL, a centre that investigates and tests EV charging solutions (ElaadNL, 2022a). National Charging Infrastructure Knowledge Platform Foundation (NKL), which is dedicated to lowering the public charging infrastructure costs by optimising its installation (Nederland Elektrisch, 2023). Furthermore, the four technical universities (4TU) of the Netherlands were one of the partners of the Formula E-Team, having therefore contributed to knowledge transfer among member (RVO, 2018). Through their EV-dedicated programmes, the 4TU - Delft, Eindhoven, Twente, and Wageningen - are committed to creating expert engineers and designers that can lead the industry (Nederland Elektrisch, 2019).

Feedback 9: Policymakers \leftrightarrow Trade Associations

Consultation; Foster collaboration: Policymakers involved trade associations in collaborative projects. The 2016 Formula E-Team, a public-private partnership to promote electric transportation included trade associations such as the Interprovincial Consultation, the Grid Operators Netherlands (*Netbeheer Nederland*) and the Association of Electric Drivers (IPO, 2024; Nederland Elektrisch, 2019; Netbeheer Nederland, 2024; VER, 2023). Trade associations also collaborated with the Dutch government and other actors, such as firms and academic and research institutes in e.g., the 2019 Climate Agreement (Nationale Agenda Laadinfrastructuur, 2022). Both instances show that the government actively engaged and consulted trade associations for their knowledge on the industry and market situation. Furthermore, trade associations played a big role in bringing the concerns of the industry and (potential) adopters to the policymakers, such that collaboratively a better solution could be found. These include the VZR, ANWB, RAI, BOVAG, VNA, EVO, TLN, and VNO. For instance, BOVAG, RAI, and VNA lobbied against the system of determining the additional tax of a car. Currently based on the vehicle's CO₂ emissions, one more or one less gram of CO₂ might mean the car pays 14% or 20% of additional tax (VZR, 2022).

$\textbf{Policymakers} \rightarrow \textbf{Adopters}$

Purchase subsidies: At first, direct purchase subsidies and reduced registration costs mainly targeted PHEVs (Deuten et al., 2020; Transport & Environment, 2016). However, after 2017 BEVs started increasing at a high rate and in 2022 BEVs comprised 65% of Dutch EVs (IEA, 2022). Prospective owners of EVs have been benefiting since 2020 from €2,950 to buy or lease a new EV, while this subsidy lowers to €2,000 for second-hand cars (Milieu Centraal, 2024; RVO, 2017a, 2022b). Nonetheless, these premiums have been reducing as the industry develops (IEA, 2023b).

Tax rebates: Tax reductions and exemptions have also been serving as incentives to purchase an EV (Klimaatakkoord, 2021). After 2011 EVs were exempt from circulation and registration taxes, although the exemption for PHEVs was dependent on emission standards and ended in 2016 (ACEA, 2014; Transport & Environment, 2016). BEVs also enjoyed a tax advantage with a reduced addition to taxable income for company-owned cars of 4% from 2014 onwards. Similarly, PHEVs had reduced additions to taxable income over the years, starting at 7% in 2014 and increasing to 22% in 2017 (Deuten et al., 2020). Over time, tax benefits extended to acquisition, ownership, and company cars, favouring zero-emission vehicles. (Klimaatakkoord, 2021).

$\textbf{Policymakers} \rightarrow \textbf{Academic \& Research Institutes}$

R&D subsidies: Between 2015 and 2021 academic and research institutes could apply to some of the financial support provided by the Dutch government and the European Commission aimed at promoting the development of electric vehicles and their market (RVO, 2017a). One example was the Subsidy Scheme for Demonstration of Climate Technologies and Innovations in Transport, available since 2017 (RVO, 2017a, 2022a). Other European-wide support schemes include the Interreg programme, which was launched in 1990 (European Commission, 2020; RVO, 2017b) and the LIFE subsidy scheme, which exists since 1992 (European Commission, 2022; RVO, 2017c). Moreover the Connecting Europe Facility Transport scheme, created in 2014, has the goal of improving the transport network (RVO, 2014).

Foster collaboration: As mentioned in Feedback 3, Dutch policymakers fostered collaboration among different actors. This included knowledge institutions, which can be interpreted as academic and research institutes. One of the initiatives created by the Green Deal was the Formula E-Team, established in 2016 by the government, which consisted in a public-private partnership involving technology firms, knowledge institutions, and policymakers (Nederland Elektrisch, 2019, 2024a). One of these institutions is the ElaadNL which investigates and tests charging solutions for EVs (ElaadNL, 2022b). The 2019 Climate Agreement and the Dutch National Charging Infrastructure Agenda also highlight other public-private partnerships between local governments, firms, trade associations, and knowledge institutions (Nationale Agenda Laadinfrastructuur, 2022).

Promote education: As part of the Green Deal, the four technical universities (4TU), i.e. Delft, Eindhoven, Twente, and Wageningen, developed educational programmes on electric transportation to prepare future professionals for the transition (Nederland Elektrisch, 2019). Other secondary vocational education (MBO) and university (HBO) programmes also give EV related courses (Tweede Kamer der Staten-Generaal, 2016).

Academic & Research Institutes \rightarrow SET

Research and test: In the Netherlands, several academic and research institutes research and test electric vehicle technology. ElaadNL, for instance, focuses on sustainable and smart EV charging for cars, buses, and trucks (ElaadNL, 2022a). Besides, NKL focuses on the optimisation of the public charging infrastructure (Nederland Elektrisch, 2023). The 4TU have also been committed to electric-vehicle-related education and research (Nederland Elektrisch, 2019).

Trade Associations \rightarrow Adopters & Technology Firms

Foster collaboration: Trade associations were active in lobbying at the government and ministries in the name of their members, to whom they also fostered collaboration opportunities. Examples of trade associations are the RAI, BOVAG, VNA, EVO, TLN, and VNO (VNA, 2024; VZR, 2022). For instance, the RAI Association represents manufacturers and importers of several vehicle types (RAI Vereniging, 2024), while BOVAG was founded by car companies but now includes also other companies whose business is associated with vehicles. BOVAG offers professional support and knowledge sharing opportunities (BOVAG, 2023). Additionally, the VNA (Association of Dutch Car Leasing Companies) offers insights on future regulations and policies to the government to favour the car leasing industry. The VNA encourages their members to share their views on different topics. Besides, the VNA seeks collaboration with other trade associations, such as BOVAG, RAI Association, Evovenedex, ANWB, among others. VNA also represents private lease drivers (VNA, 2024). Other trade associations representing adopters include VER, which represents all electric drivers (VER, 2023), while the VZR is the voice of the business drivers (VZR, 2024). Moreover, ANWB stands up for the interests of mobility and tourism parties, for instance by offering retail, insurance, travel, and roadside assistance (ANWB, 2020).

$NGOs \rightarrow Policymakers$

Lobbying: Environmental NGOs have shown enthusiastic support for electric vehicles in different ways. For instance, the *Natuur en Milieu* (Foundation for Nature Conservation and Environmental Protection) was one of the partners of the Formula E-Team, showing in this way their strong support for the technology (RVO, 2018). *Milieudefensie* (known in English as Friends of the Earth Netherlands) has also been engaged in environmental protection activities since 1971, including promoting clean mobility solutions such as electric vehicles (Knol, 2018). Transport & Environment is another NGO that acts at the European level and has been active in advocating for electric transportation (T&E, 2019). Nonetheless, it should be noted that NGOs also point out potential technological improvements to achieve a higher product sustainability. For example, *Natuur en Milieu* wrote about the sustainability of batteries (Penders & de Lange, 2022).

$\textbf{NGOs} \rightarrow \textbf{Wider Public}$

Public awareness campaigns: NGOs share important information about different technologies such that they can raise awareness among the general public. This can be done through articles, campaigns, or petitions. For instance, *Natuur en Milieu* wrote about sustainable car batteries and the advantages of having an electric car (de Jongh, 2022; Penders & de Lange, 2022). A campaign meant to promote electric vehicles among Uber drivers was named the 'True Cost of Uber' and was launched by environmental NGOs from different countries, including the Netherlands. The campaign raised awareness through petitions in cities such as Amsterdam (T&E, 2019).

In this chapter, the four case studies performed are extensively described, providing insights into the tipping point dynamics of each case. Each case's main policies are summarised in Section C.2. Chapter 7 puts the findings into perspective, discussing them.

7 | Discussion

In this chapter, the results are discussed. In Section 7.1 the expansion of the framework developed by Geels and Ayoub (2023) is presented. This section covers which actors and actor roles are deemed relevant in triggering a positive tipping point in the adoption of wind energy and electric vehicles in specific, and of sustainable energy technologies in general. Then, Section 7.2 discusses the main outcomes of the validation process. Finally, some general remarks are provided in Section 7.3.

7.1. Model Expansion

The expansion of the framework of Geels and Ayoub (2023) is presented in Section 7.1.1, answering sub-question 5. Then, Section 7.1.2 and Section 7.1.3 elaborate upon the relevant actor groups and their roles, covering sub-questions 2 and 3, respectively. In Section 7.1.4, the specific role of policy-makers is discussed and sub-question 4 is answered¹.

7.1.1. Final Feedback Loop Models

Two final feedback loop models were created, one representing the wind energy case studies, and the other the EV ones. These are shown in Figure 7.1, and Figure 7.2, respectively. The figures were constructed based on the case-specific feedback loop models (shown in Chapter 6), by overlapping the interactions found in the two wind energy and in the two EV case studies². The grey blocks and black labels can be found in Figure 4.3, the refined version of Geels and Ayoub (2023), and are common among both final models. The green and yellow blocks and labels are additions made based on this thesis' work. Two models are proposed because the main actors and actor roles differed in some aspects between the wind energy and the EV cases. This is discussed in Section 7.1.2 and Section 7.1.3. To make these differences clearer, in Figure 7.1, and Figure 7.2 green is used for elements that belong to both models, while yellow shows elements that are particular to either model.

In comparison, Geels and Ayoub (2023) presented a single feedback loop model. It is argued that this results in an inaccurate and too generic representation of the deployment of technologies that are fundamentally different, at least when it comes to their target audience and to the scale in which they are adopted. The differences between the wind energy and EV case studies suggest that there are differences between the tipping point dynamics of B2B and B2C technologies. Figure 7.1 and Figure 7.2 can thus be used as a *reference* for the socio-technical dynamics of positive tipping points in the adoption of B2B and B2C technologies, respectively. Nevertheless, it should be kept in mind that the results are based solely on four case studies that only focus on two technologies. Actors and actor roles might show variations when looking at other technologies and/or contexts.

¹Sub-question 1 has been answered in Section 4.2.

²There are two exceptions, however. The interactions *promote education* and *training* were found in at least one of the two EV case studies and yet were left out of Figure 7.2. The reasoning for this is given at the beginning of Section 7.1.3.



Figure 7.1: Feedback loop model showing the relevant actors and respective roles in triggering a positive tipping point in the adoption of wind energy. This model might be used as a reference for positive tipping points in the uptake of business-to-business sustainable energy technologies, i.e. mainly adopted in large numbers by businesses

Sub-question 5 - How can the feedback loop model of Geels and Ayoub (2023) be expanded to include a wider set of actors and interactions between these while maintaining its readability? - is therefore answered by the two feedback loop models illustrated in Figure 7.1, and Figure 7.2. Concerning maintaining the readability of the models, it was attempted to only include data that had been relevant in triggering the tipping point or maintain the subsequent accelerated growth (see Section 5.2.2).

Furthermore, Figure 7.1 and Figure 7.2 show several interactions between actor groups that are not part of a feedback loop. In contrast, Geels and Ayoub (2023) only included one of these 'loose' interactions in their model, the *purchase subsidies* given by policymakers to adopters. However, these interactions are important as they might be responsible for triggering or strengthening reinforcing feedback loops. For instance, financing incentives and (binding) targets helps set in motion and maintain feedback loops 1 and 2. As another example, public awareness campaigns of NGOs raise awareness among the wider public and shapes their opinion, strengthening feedback loops 5, 6, and 7.



Figure 7.2: Feedback loop model showing the relevant actors and respective roles in triggering a positive tipping point in the adoption of electric vehicles. This model might be used as a reference for positive tipping points in the uptake of business-to-consumer sustainable energy technologies, i.e. mainly adopted in small numbers by individual buyers

As noted by interviewee 4, apart from the nine or ten feedback loops identified in Figure 7.1 and Figure 7.2, there might be other larger, hidden feedback loops occurring. For instance, in Figure 7.1 *investment in R&D and production* results in higher product *value* which allows potential adopters to give *long-term certainty* to financial institutions, resulting in more *financing solutions* available. This leads to more *purchases* and therefore higher *revenue and profit* for technology firms, which in turn use part of that money for more *investment in R&D and production*, repeating the cycle.

These larger feedback loops are important since they encompass more than two actor groups, truly showing how the socio-technical system is connected and the importance of having such an holistic view. Nonetheless, this thesis focused on particular interactions and feedback loops among only two actor groups. Consequently it is now hard to identify these larger feedback loops in Figure 7.1 and Figure 7.2. If they had been kept in mind from the beginning, perhaps the information had been presented differently in the feedback loop models or additional links would have been added. This is therefore one of the limitations of the current work, and could be an avenue for future research (see Section 8.3).

Section 7.1.2 discusses the actors present in Figure 7.1 and Figure 7.2, while Section 7.1.3 does so for the interactions and actor roles.

7.1.2. Actors

Table 7.1 provides a summary of the actor groups in Geels and Ayoub (2023), the wind energy case studies, and the EV ones, and how these compare to the actors found in the initial literature (see Section 4.3). The literature initially analysed suggested the relevance of eight actor groups in triggering tipping points in SET adoption: policymakers, technology firms, adopters, wider publics, academic and research institutions, trade associations, NGOs, and financial institutions. Geels and Ayoub (2023) only included the first four in their model. In contrast, as observed in Figure 7.1 and Figure 7.2, all eight actors were found important in the wind energy case studies, while only financial institutions did not appear relevant in the deployment of EVs. While public financial incentives proved essential for EV adoption, the influence of private financial institutions appeared to be limited. The results are in agreement with Muratori et al. (2021). Conversely, documentation on wind energy developments frequently cited financial institutions as a catalyst.

The answer to sub-question 2 - "Apart from the actor groups selected by Geels and Ayoub (2023), which other actors are relevant in promoting positive tipping points in the deployment of sustainable energy technologies?" - therefore has two parts. On the one hand, for B2B technologies like wind energy all eight aforementioned actor groups seem to be relevant. On the other hand, in the uptake of B2C technologies such as EVs, the same actor groups appear important with the exception of financial institutions. This difference was one of the reasons for having two final feedback loop models.

Actor groups expected (Section 4.3)	Geels and Ayoub (2023) (Figure 4.3)	Wind energy (Figure 7.1)	Electric vehicles (Figure 7.2)
Policymakers	1	1	1
Technology Firms	1	1	✓
Adopters	1	1	✓
Wider Public	1	1	1
Academic & Research Institutes	×	1	1
Trade Associations	×	1	1
NGOs	×	1	✓
Financial Institutions	×	1	×

Table 7.1: Actor groups found on the feedback loop model of Geels and Ayoub (2023), and those based on the wind energy and EV case studies. The eight actor groups are the ones that were expected based on the literature presented in Section 4.3

Generalising the case studies findings to other (B2B and B2C) SETs should be done carefully. For one, this thesis focused solely on four case studies. Analysing different cases might lead to different conclusions. For another, only two technology types were analysed. More research, namely involving other technology types, should be conducted to improve the robustness of these generalisations.

A relevant observation is that, during the periods analysed, incumbents (both technology firms and adopters) were very committed to the transition, contrary to what was expected from literature (Geels, 2019; Lenton et al., 2022; Roberts et al., 2018). However, it might be that for many years incumbents fought change, with the periods analysed coinciding with the moment in which they eventually committed to it. This was partially suggested by interviewee 1 but also by Geels and Ayoub (2023). In their EV case study it is observed that, while initially Tesla was the biggest player, after the 2015 Diesel scandal all incumbents followed suit. The tipping point then occurred a few years after in the UK, the Netherlands, and globally. If this is true for other case studies too, the results might suggest that the involvement of incumbents is essential to trigger tipping points. However, performing case studies that cover more years prior to the positive tipping point is necessary to prove this hypothesis.

Additionally, as discussed with interviewee 1 and 2, some actors can fit in different actor categories depending on the situation. This is the case for consulting firms, which perform many activities. While they fit the best within *technology firms* they perform certain roles at times that align with other actor groups. Furthermore, project developers might fit in *technology firms* or in *adopters* depending on whether they are involved in the installation of the project or only on getting it to a ready-to-build phase.

Finally, despite useful, the concept of actors can be limiting if certain functions or actions are seen as pertaining solely to that actor group. It might be that a function can be performed by another actor. It is hence beneficial to think in terms of actor roles, as discussed in Section 7.1.3.

7.1.3. Actor Roles

Actor roles refer to the functions that the actors perform in the socio-technical system. This is a useful concept as it specifies roles that need to be fulfilled, rather than who fulfils it. They are shown in the feedback loop model as the interactions between actors or between an actor and the technology, which are represented by the arrows and respective labels. All the interactions included in the feedback loop models presented throughout this report are summarised in Appendix B.

Although in this thesis actor roles relate to interactions *between* actors, important system functions can be performed by a single actor group, independently. For instance, academic and research institutes might conduct research within their own organisation, without collaborating with other parties. In that case, despite relevant, this function is not captured in the feedback loop model. That is one of the limitations of Geels and Ayoub (2023) but also of this thesis, as mentioned in Section 8.3.

Upon an initial inspection of relevant literature (see Section 4.3), thirteen overarching actor roles were formulated. As shown in Table 7.2, twelve of these actor roles were identified in the wind energy case studies and eleven on the EV cases. The role of *promoting education* was not deemed relevant in the deployment of neither wind turbines nor EVs. Additionally, the role of *establishing market-based mechanisms* was not present in the EV case studies. In contrast, the model of Geels and Ayoub (2023) only includes seven actor roles, suggesting the authors left important aspects out of their model.

Actor roles expected (Section 4.3)	Geels and Ayoub (2023) (Figure 4.3)	Wind energy (Figure 7.1)	Electric vehicles (Figure 7.2)
Technology development	1	1	1
Technology adoption	1	1	1
Legitimising technology	1	1	1
Providing financing incentives/options	1	1	1
Investing in infrastructure	1	1	1
Establishing legislation	1	1	1
Lobbying	1	1	1
Raising awareness	×	1	1
Fostering collaboration	×	1	1
Transferring knowledge	×	1	1
Giving & receiving feedback	×	1	1
Establishing market-based mechanisms	×	1	×
Promoting education	×	×	×

 Table 7.2: Summary of the relevant overarching actor roles in achieving a positive tipping point and whether they were included in the feedback loop model based on Geels and Ayoub (2023), the wind energy case studies, and the EV case studies. These actor roles were expected based on the general literature presented in Section 4.3

These results can be used to answer sub-question 3: *"What are the main actor roles in promoting a positive tipping point in the deployment of sustainable energy technologies?"*. The findings suggest that (at least) the eleven actor roles that were found both in the initial literature and in the four case studies should be fulfilled to trigger a tipping point in the deployment of SETs. As for *establishing market-based mechanisms*, only present in the wind energy case studies, the results indicate that they are probably only relevant for B2B technologies. Nevertheless, it should be taken into account that the context can play an important role and should always be considered carefully. Furthermore, despite the consistency of the results obtained, more case studies should be conducted to verify them.

The only actor role not included in any of the final feedback loop models, *promoting education*, was actually found on the EV case studies. In the UK, policymakers created a retraining programme for technology firms to acquire new skills aligned with a zero-emission mobility future. It is possible that in the other case studies retraining happened within the firms themselves, not involving a special government programme, and therefore did not appear as an interaction *between* actors. Besides, in both the UK and the Netherlands policymakers promoted and invested in education programmes at academic institutes. However, educational programmes take time to be improved or developed, and it takes time for the students to graduate from these programmes and join technology firms or start producing relevant research. In the UK, large investments in education occurred only one year before the tipping point. In the Netherlands, the tipping point occurred two years after EV-specific courses and programmes were strongly developed. Upon interviewee 4 agreeing that the promotion of education by policymakers was not a relevant factor for either tipping point, this interaction was excluded from the final models.

Despite the high match between the actor roles expected from the initial literature and the case studies, not all interactions in Figure 4.4 are found in Figure 7.1 and Figure 7.2. This showcases that an actor role might be fulfilled in distinct ways by distinct actors. However, the fact that not all the exact interactions in Figure 4.4 were found in the case studies can be due to several reasons. Firstly, the literature concerned a socio-technical context but not necessarily related to *sustainable energy* technologies. Secondly, it is possible that actor roles differ in different stages of the technology's development and deployment and that perhaps some of the literature was specific of a particular phase. Thirdly, if more case studies were researched perhaps more of the interactions in Figure 4.4 would have been identified in them. Below, an explanation is given of the twelve actor roles found in the case studies.

Technology development

Technology development entails improving the technology's performance and reducing its cost, hence increasing its value proposition. Two main actor groups were involved in this process. As identified by Geels and Ayoub (2023), technology firms invest in R&D and production to improve the quality of their products and the efficiency of their processes. For another, academic and research institutes conduct research and tests, especially relevant for novel technologies. Table 7.3 summarises the interactions found on this report's feedback loop models related to technological development.

Table 7.3: Interactions related to	technology development, as wel	I as the actor groups involved
------------------------------------	--------------------------------	--------------------------------

Interaction label	Source	\rightarrow	Recipient
Investment in R&D and production Value	Technology Firms		SET Adopters
Research and test	Academic & Research Institutes		SET

Technology adoption

As a technology's performance raises and its cost reduces, the role of *technology adoption* is stimulated. The feedback loop model interactions that relate to the role of technology adoption are listed in Table 7.4. Higher adoption levels increase the technology's visibility, making it more desirable to the wider public through social contagion and leading to more adoption. Higher purchases allow for higher revenues and profits for technology firms, allowing them to further invest in increase a technology's value proposition.

Table 7.4: Interactions related to technology adoption, as well as the actor groups involved

Interaction label	\mid Source \rightarrow	Recipient
Purchases	Adopters	SET
Revenue and profit	Adopters	Technology Firms
Further adoption	Wider Public	Adopters
Expanded technology's visibility	Adopters	Wider Public

Legitimising technology

As public acceptance increases, the technology is legitimised. This is important because public opinion can strongly influence a technology's adoption, as shown by nuclear energy developments in several countries (Ming et al., 2016). A positive public opinion can create a conducive environment for a technology's development and adoption. In the case studies it was observed that two main factors contribute to shaping public opinion. The first one is to frame the technology as offering a solution to societal concerns, being them environmental or of another nature. Secondly, policymakers can also have a role in shaping public opinion through their policy plans and accompanying narratives. The feedback loop model interactions that are associated with legitimising the technology are summarised in Table 7.5.

Table 7.5: Interactions related to legitimising the technology, as well the parties involved

Interaction label	\mid Source $ ightarrow$	Recipient
Policy plans and accompanying narratives	Policymakers	Wider Public
Solution to societal problems	SET	Wider Public
Legitimacy	Wider Public	SET
Public opinion	Wider Public	Policymakers

Providing financing incentives/options

Financing has proved essential in promoting the research, development, and adoption of the technologies analysed. In the model of Geels and Ayoub (2023) policymakers are the only ones providing financing options. However, the literature initially analysed (see Section 4.3) and the case studies showed that technology firms and financial institutions can also be important actors in this respect. Table 7.6 provides an overview of the feedback loop model interactions related to this actor role.

As included in Geels and Ayoub (2023), policymakers can give loans, capital grants, and R&D subsidies to technology firms to promote the development of new technologies or the improvement of existent ones. They can also give purchase subsidies to adopters, making technologies more affordable. Nonetheless, these are not the only financial incentives given by policymakers. For instance, in all case studies policymakers provide R&D funding and grants to academic and research institutes. Besides, feed-in tariffs were observed in both wind energy case studies, while tax rebates played an important role in the EV cases. Although this difference might be related to whether it is a B2B or B2C technology, it might also be a consequence of whether the technology is used to generate energy/electricity or not.

Furthermore, it was found on three of the four case studies that technology firms provide funding to academic and research institutes for research to be conducted (collectively or not) on topics that are relevant to the firm. This was confirmed and emphasised by interviewee 1. This interaction was most likely not found on the wind in Portugal case study because during the period analysed the country relied mostly on imports (interviewee 2). Additionally, financial institutions³ provided financing support to the adopters of wind energy projects both in the UK and in Portugal. This was true especially after projects started providing long-term certainty. Project finance can at times be the distinguishing factor between the realisation or not of projects like wind farms, and should therefore not be dismissed.

Table 7.6: Interactions related to providing financing options, as included in the feedback loop model, as well as the parties			
involved			

Interaction label	\mid Source $ ightarrow$	Recipient
Loans	Policymakers	Technology Firms
Capital grants	Policymakers	Technology Firms
R&D subsidies	Policymakers	Technology Firms
		Academic & Research Institutes
Purchase subsidies	Policymakers	Adopters
Tax rebates	Policymakers	Adopters
Feed-in tariffs	Policymakers	Adopters
Funding	Technology Firms	Academic & Research Institutes
Long-term certainty	Adopters	Financial Institutions
Financing solutions	Financial Institutions	Adopters
Public financing departments or businesses	Policymakers	Financial Institutions

Investing in infrastructure

Table 7.7 shows the feedback loop interactions that correspond to this actor role. In the UK civil engineering contractors contributed to developing the transmission infrastructure for wind energy. Besides, in the Netherlands private firms invested in expanding the EV charging infrastructure. Hence, although infrastructure investment is usually a public endeavour, it can also be conducted by technology firms.

Table 7.7: Interactions related to investing in SET-related infrastructure, as well as the parties involved

Interaction label	Source	\rightarrow	Recipient
Investing in infrastructure	Policymakers		SET
	Technology Firms		SET

Establishing legislation

Several interactions relate to estabilishing legislation, as summarised in Table 7.8. Regulations and standards can be effective in guiding development and adoption choices. For instance, the Renewable Obligation scheme in the UK mandated utility firms to have a specific percentage of renewable electricity in their portfolio. In Portugal, the installation of wind energy projects was promoted by liberalising the

³Some of the financial institutions were state-backed or state-owned.

energy market and simplifying the licence granting process. When it comes to EVs, the creation of clean air zones within cities was one of the EV promoters both the UK and the Netherlands.

Furthermore, setting ambitious but attainable targets, being them binding or not, was found to be a policy instrument capable of steering the industry and the market in the right direction. An important difference was found, nonetheless, between the wind energy and EV case studies, suggesting differences between B2B and B2C technologies. For one, in wind energy the targets were aimed at the adopters (e.g., energy or utility firms) as they concerned adoption figures for wind energy or renewable energy capacity. For another, in the EV case studies targets were directed at technology firms (e.g., automakers). These include the phase-out of polluting cars and CO_2 targets for the cars sold.

Finally, this actor role also relates to evaluating and adjusting policies, goals, and coalitions such that they are effective. This interactions was already included in Geels and Ayoub (2023) and was found to have been relevant in all four case studies. In fact, it was found in all cases that adoption accelerated after a certain policy or goal was adjusted. This is covered in more detail in Section 7.1.4.

Table 7.8: Interactions related to setting legislation, as well as the parties involved

Interaction label	Source	\rightarrow	Recipient
Regulations and standards	Policymakers		SET
(Binding) targets	Policymakers		Technology Firms (wind energy case studies)
			Adopters (EV case studies)
Policies, goals, and coalitions adjustments	SET		Policymakers

Lobbying

The interactions and actor groups involved in lobbying are summed up in Table 7.9. Trade associations were involved in lobbying in the name of their members (both technology firms and adopters), leading to a more collaborative environment. Additionally, lobbying was also performed directly by technology firms and NGOs, with respect to for instance the industry's direction and environmental concerns. While lobbying can be performed to fight change, there can also be 'positive' lobbying, in which actors enthusiastically advocate for a certain technology. Both situations were observed. Furthermore, although lobbying was defined as a separate actor role, it closely relates to legitimising (or not) a technology.

Interaction label	Source	\rightarrow	Recipient
Lobbying	Technology Firms NGOs		Policymakers Policymakers
Foster collaboration Public opinion	Trade Associations Wider Public		Policymakers Policymakers

The seven actor roles described above were included in Geels and Ayoub (2023). However, five others were identified in the initial literature and in the case studies. These are described below.

Raising awareness

Raising awareness can be accomplished via public awareness campaigns, which were observed to be typically performed by NGOs. Trade associations also raise awareness through campaigns and publications, usually targeted at adopters. Both these interactions are summarised in Table 7.10.

Table 7.10: Interactions related to promoting public awareness, as well as the parties involved

Interaction label	Source	\rightarrow	Recipient
Public awareness campaigns	NGOs		Adopters
Foster collaboration	Trade Associations		Adopters

Fostering collaboration

Collaboration played a key role in all case studies. The main actors fostering collaboration were policymakers and trade associations. Furthermore, the actor groups involved in the collaboration opportunities include policymakers, academic and research institutes, technology firms, trade associations, and adopters. This information is summarised in Table 7.11.

The involvement of adopters played a bigger role in the wind energy case studies than the EV ones due to the different nature of adopters of B2B and B2C technologies. In the wind energy cases the adopters were mainly a few well-established energy firms, facilitating collaboration. Conversely, most EV adopters are individual citizens, although companies with electric car fleets also fall under the adopters category. It is hence more impractical to involve EV adopters in collaborative opportunities. Nonetheless, both in the UK and in the Netherlands there were trade associations who represented EV drivers, fostering a collaborative environment between adopters, policymakers and technology firms.

The role of trade associations in fostering collaboration encompassed collaborative projects but also the representation of and lobbying for their members. By advocating for their members interests, trade associations fostered a cooperative environment in which the views of different stakeholders were valued. Lobbying in the name of technology firms led to more realistic and effective policies. Furthermore, voicing adopters concerns might result in technology firms improving certain aspects of a technology they had not yet considered, or to release stronger policies to help adoption uptake.

Interaction label	Source -	\rightarrow	Recipient
Foster collaboration	Policymakers		Technology Firms Academic & Research Institutes
	Trade Associations		Policymakers Technology Firms Adopters

Transferring knowledge

In the case studies, knowledge was transferred between academic and research institutes and technology firms. This mutual knowledge transfer relationship is summed up in Table 7.12. Academic and research institutes conduct relevant research that might help advance technologies. Knowledge transfer from academic and research institutes to technology firms can occur through journal and conference publications or through collaborative projects. When the latter applies, there is also knowledge shared in the opposite direction. Additionally, technology firms can provide training to academic and research institutes, as was found on the wind energy in the UK case study (Kern et al., 2014). Finally, many well-educated people leave academic and research institutes to join technology firms. There, they help advance the firm's technologies, transferring knowledge in this way.

Table 7.12: Interactions related to transferring knowledge, as well as the parties involved

Interaction label	Source	\rightarrow	Recipient
Knowledge transfer	Academic & Research Institutes Technology Firms		Technology Firms Academic & Research Institutes

Giving & receiving feedback

The interactions related to giving and receiving feedback are summarised in Table 7.13. Firstly, policymakers approached trade associations to gauge the industry's and the market's status. This can be referred to as (policy) consultation and helps policymakers draft and adjust their policies such that they are as effective as possible. This interaction was mentioned by interviewee 1 and was verified in every case study by different sources. Secondly, user feedback plays a role in the development of any technology since design is an iterative process that tries to accommodate for the preferences of all stakeholders. User feedback was, however, found to play a bigger role in the EV case studies.

The latter could be due to a number of reasons. For one, during the period analysed for the wind energy developments in Portugal (2000 - 2007), onshore wind turbines had already been widely implemented in other countries and therefore the technology had already gone through several iterative stages beforehand. For another, for the deployment of offshore wind in the UK (2002 - 2014), the technology was adapted from onshore wind turbines, which were well-established by then. At the same time, offshore wind turbines were also not that recent themselves, with the first commercial offshore wind farm being installed in Denmark in 1991 (Ørsted, 2019). In contrast, in the period analysed for the EV case studies (2015 - 2021) the technology underwent several improvements, for instance concerned with the battery's specific capacity (energy storage capacity per unit weight) and the car's range, with the global positive tipping point occurring only around 2020 (Geels & Ayoub, 2023; IEA, 2023a).

Interaction label	Source	\rightarrow	Recipient
Consultation	Policymakers		Trade Associations
User feedback	Adopters		Technology Firms

Table 7.13: Interactions related to giving feedback, as well as the parties involved

Establishing market-based mechanisms

Another instrument that can be used by policymakers to drive adoption besides financial incentives are market-based mechanisms such as competitive tenders. Tenders are a commonly used instrument to deploy B2B SETs, with the potential to be more effective than feed-in tariffs (Bento, Borello, & Gianfrate, 2020). Tenders played an important role in both wind energy case studies, being associated with the attainment of both tipping points. In Portugal, tenders specified locations with wind energy potential that

were located in grid-accessible places or places where it was relatively easy to expand the grid. It was also seen that in the UK the tender process became particularly effective after round 3, when the zones with more potential for exploration were identified by the Crown Estate. These results suggest that for B2B technologies tenders can be an effective instruments, especially if the zones with the highest potential are identified by the tender providers. Tenders are, nevertheless, not a common instrument in the deployment of technologies adopted at an individual level. Accordingly, they did not play a role in the EV case studies. Table 7.14 shows which actor groups were involved in the tendering processes.

Table 7.14: Interactions related to establishing market-based mechanisms as well as the parties involved

Interaction label	Source	\rightarrow	Recipient
Tenders	Policymakers		Adopters

Having explained the different actor roles identified, Section 7.1.4 focuses on the specific role of policymakers, allowing sub-question 4 to be answered.

7.1.4. Role of Policymakers

Often policies are analysed in isolation. However, policies emerge from the interaction between policymakers and societal actors. They are part of a broader set of formal and informal rules that actors have to comply with. Hence, it is not appropriate to analyse policies isolated from the context they belong to (Edmondson et al., 2019; Fesenfeld et al., 2022; Roberts et al., 2018). Therefore, in this thesis policies were interpreted as an actor role and represented as only one element of Figure 7.1 and Figure 7.2, which illustrate how the policies interact with the actors and vice-versa.

As expected from literature (Edmondson et al., 2019; Fesenfeld et al., 2022; Geels & Ayoub, 2023; Roberts et al., 2018), policymakers were instrumental in achieving each case's tipping point, being involved in eight of the twelve actor roles identified in the case studies. These were legitimising technology, providing financing incentives/options, investing in infrastructure, establishing legislation, lobbying, fostering collaboration, giving and receiving feedback, and establishing market-based mechanisms.

Despite the importance of all these roles, financial incentives, infrastructure investment, legislation, and market-based mechanisms seemed to have the largest impact. Firstly, financial incentives substantially helped make the technologies more affordable, while market-based mechanisms (in the wind energy cases) promoted the strategic allocation of resources. In the wind energy case studies, purchase subsidies, feed-in tariffs and tenders were used to promote adoption. For another, the uptake of EVs both in the UK and in the Netherlands was fostered by purchase subsidies and tax rebates.

Secondly, in the four case studies policymakers were instrumental in developing the infrastructure. In the wind energy cases, they streamlined the connection processes and reinforced the grid, while in the EV case studies policymakers installed more public chargers. Despite the different types of infrastructure investment, the case studies show the relevance of it being mostly a public investment. For one, the national grid is most often publicly owned so public bodies are usually responsible for its reinforcement and expansion. For another, when it comes to EV charging points, not only public investment can help reduce range anxiety at an initial deployment stage (Patil, 2019), but also policymakers are usually the best equipped to optimally place charging devices within a city (Lamontagne et al., 2023).

Thirdly, in all case studies legislation steered the industry in the right direction, particularly the targets established. For wind energy, these targets were directed at the adopters and concerned the share of renewable sources in electricity and energy production, as well as wind installed capacity. For EVs, the targets were aimed at the technology firms and related to the phase-out of polluting vehicles. Despite directed at different actor groups, targets proved invaluable in all cases.

Therefore, in the four case studies the most influential policies were deployment-oriented, even though policies related to R&D and technology development activities were still observed. This is in agreement with Jennings, Tipper, Daglish, Grubb, and Drummond (2020), who states that as a technology matures the policies gradually shift from being technology-push to being demand-pull. In other words, from focusing on the development of technical knowledge to focusing on developing the market.

Although the tipping points analysed were a consequence of the interventions of several actors, their timing appeared to be correlated with the implementation or adjustment of a policy. The UK's offshore wind energy experienced a positive tipping point in 2009, the same year of the Renewable Obligation amendment, which doubled the financial support for offshore wind. In Portugal, the installed capacity of onshore wind accelerated after the feed-in tariffs were increased in 2001, leading to a spike in applications in 2002. However, this high interest was only reflected in adoption numbers after 2004 due to the high number of applications and the long licensing processes. In the UK, the number of EV adopters and public charging points increased after 2019, with plans to phase-out petrol and diesel cars made in 2017 and 2018. In the Netherlands, public chargers and EV adoption increased after 2018, with plans to sell only zero-emission vehicles after 2030 occurring in 2017. These results suggest that when a policy is effective its consequences are observed relatively fast. From this follows the importance of evaluating and adjusting policies and goals (part of the actor role of *establishing legislation*).

Each case study policies are listed in Section C.2. Table 7.15 summarises this information by showing which types of economic, regulatory and information instruments (see Section 4.3.1) were found on each case study, versus which ones were included in Geels and Ayoub (2023). In all case studies financial, regulatory and information instruments were employed. This suggests that, although some instruments might be more effective than others as discussed above, attaining a tipping point in SET adoption depends on a combination of policies, or policy mixes, as argued by Edmondson et al. (2019).

Policy instrument	Geels and Ayoub (2023)	Win UK	d Energy Portugal	EI UK	ectric Vehicles The Netherlands
Economic instruments					
Financial Incentives Market-based mechanisms R&D support	۲ ۲ ۲	1 1 1	5 5 5	✓ × ✓	√ × √
Regulatory instruments					
Standards Strategic planning Market facilitation	۲ ۲ ۲	\ \ \ \	5 5 5	\$ \$ \$	J J J
Information instruments					
Knowledge transfer Collaboration & networking Outreach	× × ×	✓ ✓ ✓ ×	× √ ×	✓ ✓ ×	√ √ ×

 Table 7.15: Summary of the relevant policies on each case study, as well as whether those policy types were also found on Geels and Ayoub (2023). For a detailed explanation of the categories used see Section C.1

Sub-question 4 can be answered with the help of Table 7.15, which reads as follows: Apart from the policy types identified by Geels and Ayoub (2023), which other policies are relevant in promoting positive tipping points in the deployment of sustainable energy technologies? Geels and Ayoub (2023) included financial incentives and R&D support as economic instruments, as well as standards and market facilitation as regulatory instruments. It was found that market-based mechanisms can also be a relevant economic instrument for wind energy in particular and B2B SETs in general. (Binding) targets were also another important regulatory instrument in all case studies. Furthermore, while Geels and Ayoub (2023) did not consider any information instruments, the case studies suggest that policymakers actively promoted knowledge transfer, collaboration and networking opportunities.

Some remarks should, however, be made. First, policymakers were not found to promote knowledge transfer in the uptake of onshore wind energy in Portugal. This could be because during the period analysed there was almost no national wind energy industry, with the country relying mostly on imports. Second, market-based mechanisms were only observed in the wind energy case studies, suggesting they might be more relevant for B2B technologies. Third, in none of the case studies were there outreach policy instruments. Nevertheless, this role was not left unfulfilled. In all case studies NGOs and trade associations raised awareness among and shared information with the wider public and the adopters. This shows that certain actor roles are interchangeable.

Lastly, looking back at the framework developed by Fesenfeld et al. (2022) (see Figure 3.3), in the case studies analysed the policies were substitution-focused as technological change was very relevant in both cases, taking precedence to behavioural changes. Even though a transition to more sustainable technologies is incredibly valuable, a next step in the direction of further fighting climate change might relate not only to substitution but also to sufficiency practices, requiring high behavioural changes too. To exemplify sufficiency, future policies might promote more strongly the use of public transportation versus a private electric car. These thoughts could be explored further in future research.

Having looked at how the model proposed by Geels and Ayoub (2023) was expanded based on this thesis findings, Section 7.2 presents the main points of the validation conducted.

7.2. Validation

To validate the case studies findings, and the final feedback loop models, four semi-structured interviews were conducted. A summary of the interviews can be found in Appendix D. Below, the main points are presented. First, some general comments are covered. This is followed by the validation of the case studies in Section 7.2.1, and of the final feedback loop models in Section 7.2.2.

For one, interviewee 1 agreed that consultancy firms fit nicely within the category of technology firms. However, they pointed out that such firms can fit in different boxes. This highlights that the actor groups included in the model can be fluid, being it possible for one stakeholder to belong to different categories.

For another, interviewee 1 believed that typically small and medium enterprises invest in innovations and trigger their tipping points, with incumbents only following suit once the technology is at a more developed stage. However, as discussed in Section 7.1.2, the results did not support this hypothesis.

7.2.1. Case Studies Validation

The validation of the case studies was a result of three interviews. The main points are here discussed, in the same order as the case studies were presented in Chapter 6.

Wind Energy in the UK

Interviewee 1 argued that, apart from the arrow reading *foster collaboration* from trade associations to policymakers, there should also be an arrow going back. This should represent input and feedback sought by policymakers from trade associations on the industry or market's state to inform the development of future policies and regulations (DECC, 2009; Offshore Energies UK, 2024). In the model, this was added as *consultation*, forming feedback loop 9. The other case studies were also checked for it.

Interviewee 1 also pointed out that academic and research institutes interact directly with the sustainable energy technology block through *research and test*. Furthermore, the interviewee argued that technology firms also provided *funding* to academic and research institutes and their projects. These two interactions were not being represented. After consulting other sources, they were included in the model. The other case studies were also checked for them.

In general, interviewee 1 agreed with the specific actors presented, but mentioned that ORE Catapult also performed relevant research, and that the Renewable Energy Association (REA) was also an influential trade association. These actors were therefore added to the description of the case study.

Lastly, interviewee 1 clarified that the Crown Estate is an independent private organisation set up by the government whose profit goes to the country's Treasury. It is a private firm with the mission of bringing prosperity to the nation (The Crown Estate, 2023). Therefore, it was interpreted as a financial institution (established by policymakers) and as a policymaker, due to their proximity to the UK's government.

Wind Energy in Portugal

The first comment made by interviewee 2 that is worth more attention relates to the interviewee's remark on the stabilisation phase post-2010. According to them, it was not only due to the economic crisis but also due to grid capacity constraints. Accordingly, the introduction of the case study was adjusted.

They mentioned that in projects involving wind or solar farms it is common for project developers to develop the whole project until the ready-to-build phase, selling it to the final buyer afterwards. They argued that in this case the project developer would not be an *adopter* but would fit better under *technology firms*. This was commented on in Section 7.1.2. The interviewee also highlighted the relevance of consulting firms in providing reassurance to adopters and validating the projects. These firms, while performing these competencies could also be integrated in the category of *technology firms* (as mentioned by interviewee 1). This discussion highlighted that the group *technology firms* might currently embrace a large set of actors, which can be seen as a limitation of this thesis (covered in Section 8.3). Nevertheless, it was decided not to create extra actor categories to limit the model's complexity.

The interviewee's commented that the numbers provided for the feed-in tariffs should be referred to as average numbers. Besides, they mentioned that the duration of this support differed per project as it depended on the park's energy production. The description of the case study was adapted accordingly.

Lastly, the interviewee said that according to their perception, while over time the turbines' power and rotor diameters increased, their cost per MW did not particularly decrease. This was contradicted by reliable sources (IEA, 2004b, 2005) and thus the description of feedbacks 1 and 2 was not modified.

Electric Vehicles in the UK

Similarly to the wind energy in the UK case study, ijnterviewee 1 argued that technology firms supported academic and research institutes financially. After consulting other sources, *funding* was added as an interaction between these two actors. Moreover, at the time of the validation meeting the interaction between trade associations and adopters of *fostering collaboration* was not part of the model. Interviewee 1 agreed that this made sense since adopters were mostly individual buyers. However, afterwards associations that directly represent adopter groups, such as EVA, were found and therefore this interaction was added to the case study's model (EVA, 2024).

Furthermore, the interviewee highlighted the importance of infrastructure investments. They considered that while EVs have been affordable for a long time, the lack of an extensive charging infrastructure was one of the main impediments to the technology's adoption. The interviewee also mentioned that the charging infrastructure still needs to be further expanded and that grid capacity needs to be increased. These comments were incorporated into the case study's description.

Electric Vehicles in the Netherlands

Interviewee 3 also highlighted the infrastructure's relevance to the EV deployment in the Netherlands. At the same time, they mentioned that the environmental zones within Dutch cities were effective in promoting the transition. These points were already discussed in the case study.

They mentioned that, additionally to what had been presented, there was an auction for fast chargers next to highways, which was seized by Fastned. Besides, they pointed out that Tesla also invested in expanding the charging infrastructure, namely in partnership with the Van der Valk hotels. This information was added to feedback loop 2, backed by other sources.

The interviewee commented that, apart from other advantages of EVs, it was "cool" to own one. This closely relates to enabling condition 6, desirability/symbolism, specified by Lenton et al. (2022). Despite a technology's price or performance it might attract potential adopters due to the higher social status it brings. Nevertheless, based on the data analysed it seemed that the increasing performance of EVs and the more extensive charging infrastructure were bigger factors in fostering adoption than status.

Finally, interviewee 3 pointed out that, although currently wind turbines and EVs seem like 'easy-toimplement' technologies, at first they faced several challenges. The difficulty is in kick-starting the industry, establishing the infrastructure, and choosing where to focus. This comment relates to this research's importance for the tipping points of other SETs and is followed up in Section 7.3 and 8.2.

7.2.2. Final Feedback Loop Models Validation

In general, interviewee 4 agreed with what was presented. However, they made clear that it was not straightforward to discuss some of the points without having knowledge of the case studies. Nonetheless, in the 1h validation interview it was not possible to provide deep details about all case studies.

Interviewee 4 mentioned that literature on social acceptance typically distinguishes between the general public and communities where the technology is (to be) located, each having a different level of acceptance. They suggested that in the feedback loop models the wider public could be split into two actor groups to include this nuance. However, that was not pursued since it would add unnecessary complexity. Rather, the influence of public opinion was explained in the descriptions of the case studies.

Concerning the role of financial institutions, the interviewee commented that they might still play a role in the EV case studies, even if it is not as straightforward. They mentioned that many people might take loans to purchase an electric vehicle. However, they mentioned that including this actor or not would depend on the case study findings. For one, it might be that these people would require a loan to buy another (non-electric) car. For another, after performing extra research, private financing did not appear critical in the EV case studies and was thus not included in the respective models.

Interviewee 4 also pointed out that, although the role of policymakers is typically the most visible, it is not necessarily the most important. Instead, groups such as global communities and wider public can also play a big role in shaping the system. This was kept in mind when writing Section 7.1.4.

Lastly, the interviewee questioned whether there were not any bigger feedback loops present in the model, apart form the nine or ten included in Figure 7.1 and Figure 7.2. This comment was very pertinent and was incorporated in Section 7.1.1 and as a limitation of the work in Section 8.3).

7.3. General Remarks

Some final, general remarks can be done about the results. If the focus is to be solely on the actor roles and not the actors, a causal loop diagram (CLD) could have been more suitable. CLDs focus on relationships and feedback loops between variables, leaving out the concept of actors (Sterman, 2000). This can be an advantage since some roles can be performed by different actor groups, as discussed in Section 7.1.3. Although this avenue was briefly pursued, it was decided to utilise a similar layout as that used by Geels and Ayoub (2023) as the main goal of this thesis was to expand on their work.

Furthermore, it was deemed that the concept of actor groups was still important. Actors evolved over time to be very specialised in certain tasks, making them probably the most suitable to fulfil a particular actor role. Besides, not all roles are interchangeable. For instance, it is unlikely that academic and research institutes are going to give financial incentives to adopters. Consequently, the use of feedback loop models as those presented in Section 7.1.1 (rather than CLDs) was considered to be an adequate way of portraying the dynamics involved in triggering positive tipping points in the adoption of SETs.

Thirdly, the uptake of wind energy in Portugal might have entered the stabilisation phase too early due to the economic crisis and grid capacity issues. This might mean that there is remaining wind energy potential and that, with the right set of policies or industry effort, another positive tipping point might be observed. Despite speculative, this could mean that it is possible for multiple positive tipping points to occur. This inspires a possible avenue for future research as discussed in Section 8.3.

Finally, the case studies demonstrate that the challenge is in kick-starting the industry, setting-up the infrastructure, and deciding where to focus (interviewee 3). While currently these innovations might seem like a given, both wind turbines and EVs faced significant challenges at the beginning. Studies like this one can shed some light on how to proceed with other SETs, helping them gain traction. Keeping this in mind, Chapter 8 covers conclusions and recommendations.

8 | Conclusions & Recommendations

The answers to this research's questions are summarised in Section 8.1. The scientific contributions are considered in Section 8.2. Then, the limitations and recommendations for future research are presented in Section 8.3. Finally, Section 8.4 covers the practical applicability of the work done.

8.1. Answering the Research Questions

This thesis answered the following research question: "How can the feedback loop model of Geels and Ayoub (2023) be refined and expanded to understand how positive tipping points can be triggered in the adoption of sustainable energy technologies?". This question was answered by answering five sub-questions. Insights were gathered from general literature on the deployment of (sustainable energy) technologies, as well as four case studies where a positive tipping point in adoption had been identified. The case studies covered the uptake of offshore wind energy in the UK, onshore wind energy in Portugal, as well as electric vehicles in the UK and in the Netherlands.

Sub-question 1 read as: "How can the feedback loop model of Geels and Ayoub (2023) be refined to improve its consistency and readability?". In answering it, a refined version of the authors' model was proposed in Section 4.2. It can be visualised in Figure 4.3.

Sub-question 2 asked: "Apart from the actor groups selected by Geels and Ayoub (2023), which other actors are relevant in promoting positive tipping points in the deployment of sustainable energy technologies?". Geels and Ayoub (2023) included in their model policymakers, technology firms, adopters, and the wider public. The initial investigation indicated that four other actor groups were relevant too. These were academic and research institutes, trade associations, NGOs, and financial institutions. As presented in Section 7.1.2, in the wind energy case studies all eight actor groups appeared to have a significant role, while financial institutions were not found to be a relevant actor in the EV cases. This suggests that financial institutions might have a bigger role in the uptake of B2B technologies.

Then, sub-question 3 read as follows: *"What are the main actor roles in promoting a positive tipping point in the deployment of sustainable energy technologies?"*. It can be said that Geels and Ayoub (2023) cover seven overarching actor roles: 1) technology development, 2) technology adoption, 3) legitimising technology, 4) providing financing incentives/options, 5) investing in infrastructure, 6) establishing legislation, and 7) lobbying. In the literature initially analysed, six other actor roles were identified: 8) raising awareness, 9) fostering collaboration, 10) transferring knowledge, 11) giving and receiving feedback 12) establishing market-based mechanisms, and 13) promoting education. In the wind energy case studies only actor role 13 was not found relevant, while in the EV case studies, actor roles 12 and 13 were not identified (see Section 7.1.3). The results thus suggest that market-based mechanisms (actor role 12), observed in the form of tenders, are more suitable to promote B2B technologies.

Sub-question 4 is as follows: "Apart from the policy types identified by Geels and Ayoub (2023), which

other policies are relevant in promoting positive tipping points in the deployment of sustainable energy technologies?". In their feedback loop model, Geels and Ayoub (2023) only covered policy instruments falling under financial incentives, R&D support, the establishment of standards, and market facilitation. This thesis' results suggest that a wider set of policies are involved in achieving positive tipping points, as presented in Section 4.3.1 and discussed in Section 7.1.4. Other important policies involve market-based mechanisms (in the case of the wind energy case studies), setting (binding) targets, promoting knowledge transfer, and creating collaboration and networking opportunities. Outreach instruments were expected based on the literature initially investigated, but were not found in any of the case studies. Nonetheless, other actors appeared to take on this role, namely NGOs and trade associations.

At last, sub-question 5 was: "How can the feedback loop model of Geels and Ayoub (2023) be expanded to include a wider set of actors and interactions between these while maintaining its readability?". As discussed in Section 7.1.1, due to the differences found between the wind energy and EV case studies, this question was answered using two separate feedback loop models, presented in Figure 7.1 and Figure 7.2. The former relates to the wind energy case studies and can be used as a baseline for B2B technologies, while the latter was based on the EV case studies and can be used as a reference for B2C technologies. Figure 7.1 includes three new feedback loops, while Figure 7.2 has two new ones. Besides, in contrast with the model presented by Geels and Ayoub (2023) both models include several interactions that do not directly relate to a particular feedback loop.

8.2. Scientific Contribution

This thesis aimed to tackle relevant knowledge gaps found in literature concerning positive tipping points, as covered in Section 3.2. Concerning the lack of knowledge on how to deliberately trigger socio-technical tipping points, this researched touched upon 1) which actors are relevant to trigger a socio-technical tipping point, 2) how different actors promote or hinder a tipping point through their interventions and interactions with other actors, 3) how do the socio-technical tipping point dynamics differ in different contexts, 4) the deliberate cultivation of reinforcing feedback loops, and 5) how to identify a positive tipping point. Furthermore, when it comes to the lack of knowledge with respect to the role of policies in triggering socio-technical tipping points, this thesis partially looked at how do policies lead to behavioural and technological changes.

Despite there being many possibilities to expand on these and other knowledge gaps, this investigation focused on how to trigger positive tipping points in the adoption of SETs, a domain that remains relatively understudied (Geels & Ayoub, 2023; Lenton et al., 2022). This was done by contributing to the creation of a visual framework showing reinforcing feedback loops between actors. The conceptual framework utilised corresponded to the feedback loop model of Geels and Ayoub (2023), which offers insights into socio-technical tipping points. This research's dedication to enhancing and expanding upon this model promoted the development of a pertinent theoretical framework in the field.

The four case studies conducted contributed to the empirical knowledge on the dynamics of positive tipping points in the adoption of SETs, as recommended by Fesenfeld et al. (2022), Geels and Ayoub (2023), and Roberts et al. (2018). This is relevant as empirical evidence from previously crossed positive tipping points may allow future ones to be intentionally triggered. The prospects of doing so brings hope in realising a successful energy transition (Fesenfeld et al., 2022; Frumkin, 2022), thereby preventing the Earth climate system from degrading to an irreversible state. Additionally, conducting these four case studies and testing them against the model of Geels and Ayoub (2023) contributed to validat-

ing the authors work. Indeed, all the interactions identified by Geels and Ayoub (2023) were found in the case studies analysed. Nonetheless, this research also contributed to the scientific community by expanding on Geels and Ayoub (2023) work, as appealed by the authors themselves. This was done by adding four new actors and five new actor roles to the feedback loop model.

By investigating which actors were important, this thesis contributes to having a better understanding of which actor groups are relevant in enabling and triggering a positive tipping point in SET adoption. Furthermore, the use of feedback loop models helped explore the interaction between innovations and the socio-technical system (Geels et al., 2017). By introducing the concept of *actor roles* this thesis focused on the specific *functions* that actors play in the socio-technical system (Lenton et al., 2022). This allows for a better understanding of the actor roles needed to create the necessary conditions to trigger positive tipping points, independently of the actor that performs it. It highlights the need to intervene in different places simultaneously (Lenton et al., 2022). Indeed, socio-technical systems are complex and their developments depend on the actions of many actors and the interactions among these. Without an in-depth analysis of these roles, it might seem like all the conditions are there to enable a positive tipping point, when in fact a very relevant role might have been neglected.

One specific actor group that was found to be very relevant in all four case studies were policymakers. By analysing their role and that of different policies, this thesis took one step further in understanding how policies can effect technological and behavioural changes, a recommendation made by Fesenfeld et al. (2022). The findings of this thesis are therefore relevant both for the scientific community, as well as for policymakers and industry players (see Section 8.4).

In accomplishing technological and behavioural changes through policies the use of a feedback loop model is relevant as it allows to visualise the contextual factors and complexities present, rather than looking at policies in isolation. These models can help understand how policy mixes affect sociotechnical transformation and, in turn, how modifications to the socio-technical system influence the evolution of the policy mix, contributing to the work of Edmondson et al. (2019). The framework used in this thesis thus supports a more effective and sensible comparative policy analysis and design.

This thesis' findings relate to tipping points in the adoption of wind energy and EVs. However, the results might be extendable to other SETs. For one, the wind energy findings might relate to the deployment of other B2B technologies, mainly adopted in large numbers by businesses. For another, the results for electric vehicles might be representative of the uptake of other B2C technologies, typically adopted in small numbers by individual buyers. Nonetheless, it is important to keep in mind that different technologies might require additional actors or actor roles, since the context influences a technology's development and adoption. Additionally, since these actor groups are commonly involved in other socio-technical systems, these findings might be transverse to sectors not associated with SETs. However, more research would be needed to understand whether this is the case.

To sum up, this research tackles important knowledge gaps about how to intentionally trigger positive tipping points in the adoption of SETs, advancing understanding of the relevant actors and their roles in tipping point dynamics. It did so by refining and expanding a relevant model, that of Geels and Ayoub (2023). The results add significantly to the current discussion on socio-technical tipping points and have implications for theoretical frameworks as well as practical interventions.

8.3. Limitations & Recommendations

Several limitations can be found in the work conducted. This section covers this work's limitations and proposes recommendations on how to tackle them. These research possibilities contribute to some of the knowledge gaps identified in Section 3.2, as well as to the larger scientific scope. Section 8.3.1 discusses limitations and recommendations related to the research scope, while Section 8.3.2 cover the ones related to the analysis of the actors and actor roles. Section 8.3.3 focuses on the use of feedback loop models. Then, Section 8.3.4 considers future research avenues related to combining the direction taken by Geels and Ayoub (2023) with other frameworks. Finally, limitations and recommendations related to the operationalisation of positive tipping points are exposed in Section 8.3.5, while Section 8.3.6 concerns the relative importance and the temporal dynamics of interactions.

8.3.1. Research Scope

In terms of the scope of this research, a limitation follows from having only analysed two technology types. This inhibits the generalisability of the findings to other SETs. Future case studies should focus on other technologies such as solar photovoltaic, which offer both a B2B and B2C business model.

Additionally, all case studies concerned European countries. This was a deliberate choice such that the comparability of the case studies was enhanced. However, it might mean that the findings cannot be smoothly translated to other countries, where the context might significantly differ. This applies especially to countries with a notably different government model and developing countries. Coincidentally, triggering positive tipping points in countries such as China, India or Indonesia might have a large impact in tackling climate change (Climate Watch, 2021). Future research should therefore also analyse case studies in which a tipping point occurred in other countries, including non-European countries.

Conducting more case studies covering other technologies and countries would enlarge the empirical knowledge on tipping point dynamics in different contexts, as recommended by Fesenfeld et al. (2022), Geels and Ayoub (2023), and Roberts et al. (2018), and stated in knowledge gap 1.3. It would also help understand whether other interactions and feedback loop models emerge or whether Figure 7.1 and Figure 7.2 are indeed representative of other B2B and B2C technologies in general.

8.3.2. Actors & Actor Roles

Although the models presented are more comprehensive than that of Geels and Ayoub (2023), some actor groups still encompass a large set of actors. For instance, *technology firms* comprise the whole supply chain, including designers, manufacturers, infrastructure providers, consultancy firms, among others. Further, if project developers only get the project to the ready-to-build phase, they too fall under this actor group (interviewee 2). Future research could add additional actor groups to the model, providing a more nuanced understanding and contributing to knowledge gap 1.1 and 1.2. However, that increases the model's complexity, possibly compromising readability.

Besides, this thesis focused on the tipping point enablers, which are typically very prominent close to a tipping point. However, the barriers that had to be overcome can be as important to understand. These usually standout more at an initial stage. When conducting other case studies it can be useful to analyse more years prior to the tipping point, allowing to further understand how the tipping point dynamics relate

to actors' interventions (knowledge gap 1.2), which might promote or hinder change. For instance, incumbents are typically seen as fighting change (Geels, 2019; Lenton et al., 2022; Roberts et al., 2018). Nonetheless, all case studies suggested that incumbents were very committed to the uptake of each SET. However, it might be that before the time period analysed incumbents resisted change and that the tipping point was only enabled because their attitudes shifted. Further researching how actors' attitudes are related to the attainment of tipping points can also prove relevant in the field of adoption of innovations (Ortt & Kamp, 2022; Ortt, Langley, & Pals, 2013; Rogers, 2003).

8.3.3. Feedback Loop Models

Another limitation related to having focused on tipping point enablers is that the feedback loop models only include reinforcing feedback loops, like the model of Geels and Ayoub (2023). However, dampening feedbacks prove relevant too as they relate to the barriers found at an initial stage. Future work could integrate dampening feedback loops into the models, to improve their completeness, contributing to knowledge gaps 1.2 and 1.4. A dampening feedback loop could occur for instance when individuals resist purchasing EVs due to a perceived lack of charging stations. This slow adoption reduces demand for charging infrastructure investment, exacerbates range anxiety concerns, and further delays the expansion of charging networks. By including dampening feedback loops in the models, the main barriers to technology adoption could be linked to the enablers, guiding the strategies of several actor groups (Edmondson et al., 2019; Geels et al., 2017). For instance, technology firms could use these insights to tailor their niche introduction strategies (Ortt & Kamp, 2022; Ortt et al., 2013). Using causal loop diagrams (CLDs) could facilitate the integration of dampening feedback loops as these models identify whether a change in one variable will be amplified or dampened over time (Sterman, 2000).

Moreover, this thesis focused on 'small' feedback loops between two actor groups, while there might be 'larger' feedbacks involving several interactions in the model. In the models these larger feedback loops are not apparent, as touched upon in Section 7.1.1. This concern was raised by interviewee 4, but due to time constraints, larger feedbacks were not integrated into the model. Doing so might have revealed important, hidden system dynamics, that can help understand how tipping points are enabled. This would have significance for knowledge gaps 1.3 and 1.4. Showing the larger feedback loops could have also been easier by using CLDs since these focus on the interactions in a system and not on the actors themselves (Sterman, 2000). The latter breaks the larger feedbacks into different segments.

A further advantage of using CLDs instead is their ease of conversion into computational models, a recommendation made by Geels et al. (2017) and acknowledged in knowledge gap 1.6 in Section 3.2. This is particularly relevant in the field of computational system dynamics, where numerical models are used to interpret the system dynamics and simulate different scenarios (Crielaard et al., 2022).

Lastly, this thesis focused on interactions *between* actors. However, a lot happens *within* an actor group. There might be feedback loops occurring inside an actor group without involving other actors, which were not captured in the models. These inner-group dynamics may influence the feedback loops between actor groups, strengthening or weakening important reinforcing feedbacks. Additionally, doing so would possibly reveal additional actor roles to the ones identified in this thesis, further contributing to knowledge gap 1.2. These dynamics within a single actor group could have been represented more easily by using a CLD since such models represent the relationships between system variables, rather than focusing on the actors themselves (Sterman, 2000). By converting the CLD into a numerical model, the interactions between inner- and outer-group dynamics could be modelled automatically.

8.3.4. Application of Other Frameworks

Geels (2002) introduced the Multi-Level Perspective, widely used in research on socio-technical systems. Recently, Geels has also entered the field of positive tipping points (Geels & Ayoub, 2023). When it comes to (positive) tipping points, Lenton et al. (2022, 2023, 2008) is one of the biggest names in the field. It would be interesting to combine the work of both authors for instance by applying the framework of Lenton et al. (2022) to feedback loop model analyses like the one performed by Geels and Ayoub (2023) and in this thesis. This could help understand how the dynamics observed link to specific enabling conditions, reinforcing feedbacks, and interventions (see Figure 3.2). It is possible that a pattern is found (perhaps between B2B and B2C technologies), which could give actionable insights on which enabling conditions, feedbacks, and interventions should be pursued to effectively trigger a tipping point. Such a research avenue could also result in additions to the framework of Lenton et al. (2022) and it would contribute to knowledge gaps 1.1 through 1.5.

Another possibility for future research is to analyse these and other case studies through the lens of Fesenfeld et al. (2022) to understand how not only substitution (this thesis' focus) but also sufficiency policies can drive positive tipping points. Sufficiency policies aim to reduce overall consumption or demand and typically face higher opposition as they require more behavioural adaptation. For instance, a feedback loop model analysis could be applied to case studies concerning e.g., the ban of inner-city cars, offering comprehensive understanding of policy mechanisms that accelerate sustainable technology adoption. This would tackle knowledge gap 2.1 on how policies result in behavioural and technological changes, and gap 2.2 on understanding how to overcome political opposition. Additionally, following this research avenue would bring together the work of Geels and Ayoub (2023) and Fesenfeld et al. (2022) but also be in line with recommendations made by Edmondson et al. (2019).

8.3.5. Operationalisation of Positive Tipping Points

The operationalisation of tipping points remains challenging. Currently, there is not a precise and quantitative method for identifying socio-technical tipping points, which might lead to the incorrect identification of one. What might seem like a tipping point might solely be a short-term fluctuation. This limitation is especially relevant for the EV case studies since less time elapsed since their tipping points. Future studies could establish operational metrics, such as specific percentages of market or sales share, that reliably indicate a tipping point, as suggested by Lenton et al. (2022). This could allow to identify whether a tipping point was approaching or had just been crossed, relating to knowledge gap 1.5. This would be valuable for policymakers and industry actors in anticipating and responding to the shift.

Another option is to investigate whether it is possible to have multiple positive tipping points and what this means. Typically adoption is represented as following an idealised S-curve ending in a stabilisation phase (Rogers, 2003). However, it might be that this stabilisation can be followed by another tipping point and one more stabilisation phase. For instance, onshore wind energy in Portugal might have stabilised too early due to the economic crisis and grid capacity considerations (see Section 6.2). It is hence plausible that with the right conditions in place a second tipping point can be triggered. The danger of getting a sense of completion when observing a tipping point followed by a stabilisation phase is that perhaps one does not realise that the system has not reached its full potential yet and that another tipping point could and should be enabled. It would be interesting to study multiple tipping points in SET adoption, which might have wider implications on diffusion of innovations theories.

8.3.6. Importance and Sequence of Interactions

Future research could study the relative importance of different interactions by comparing the case studies using a quantitative method. For instance, each interaction could be given a rating according to its importance to the technology's uptake. These ratings could then be compared between case studies to understand which interactions had a low, medium and high level of importance. This could allow to, for example, understand whether the same interactions were always the most or the least relevant across all case studies. Doing this for the period before and after the tipping point could allow to understand whether the relative importance of certain actors or interactions changed after the tipping point. This could be useful to understand to which extent actors promote reinforcing feedback and tipping points (knowledge gaps 1.2 and 1.4). At the same time, it could help in guiding actors' decisions throughout the transition, possibly paving the way towards the realisation of a roadmap, advocated by Lenton et al. (2022). Namely, this could contribute to the field of political science by studying the relative importance of different political factors in a technology's adoption (Roberts et al., 2018).

Besides, this thesis did not aim to compare the sequence of different events. In contrast, Geels and Ayoub (2023) analysed the temporal dynamics of their case studies, which differed between offshore wind and EVs. Performing such an analysis for the case studies presented in this report as well as others could be an avenue for future research. Indeed, it would be relevant to understand whether a pattern is also found in this respect among B2B and B2C technologies, contributing to knowledge gap 1.2, 1.3, and 1.4. By expanding the knowledge on *how* to intentionally trigger positive tipping, this research avenue could contribute to adoption of innovation theories (Rogers, 2003) and constitute one more step towards a guiding map, helpful for taking action (Lenton et al., 2022). It could also shed light on how to best determine the moment in which a tipping point is crossed, relating to knowledge gap 1.5. If a certain sequence of events was typically observed prior to a tipping point, this could help identify whether one is about to be triggered or has just been crossed, relating to Section 8.3.5.

A last avenue for research includes understanding who initiates certain feedback loops, which can be facilitated by analysing the temporal dynamics of events. Investigating the origins of these loops can reveal which actors are most proactive in triggering and sustaining interactions, contributing to knowledge gap 1.1. For instance, identifying whether government policies, technological advancements, or consumer advocacy groups are the primary initiators in the adoption of a technology can inform future strategies. Linking this with the field of behavioural economics, understanding the actors' motivations and decision-making processes can enhance the effectiveness of interventions (Rogers, 2003; Thaler & Sunstein, 2008). Knowing which actors drive these changes and how their behaviour influences the system can help design a guiding map with targeted interventions and policies, ensuring that feedback loops leading to positive tipping points are effectively triggered and maintained (Alkemade & de Coninck, 2021; Lenton et al., 2022). Thus, it would also contribute to knowledge gaps 1.2 and 1.4.

Despite the several limitations of this study and possibilities for future research, the work performed in this thesis has policy relevant and practical applicability, as explained in Section 8.4.

8.4. Policy Relevance & Practical Implications

Understanding system dynamics and identifying less obvious indicators can be instrumental in promoting the adoption of other (sustainable) technologies. Despite its limitations, this work contributes to the scientific community and holds political and practical relevance. Scientifically, as explained in Section 8.2, this investigation enhances the academic understanding of the socio-technical dynamics involved in the adoption of B2B and B2C SETs. Researchers studying innovation systems, policy analysis, and energy transitions can build upon these findings to further explore these complex interactions, as outlined in the recommendations in Section 8.3.

This work also highlights the importance of integrating knowledge from socio-technical transitions and political science, rather than examining policies in isolation (Roberts et al., 2018). It demonstrates the value of using policy mixes (Edmondson et al., 2019). By understanding the roles and interventions of policymakers in successful case studies, government agencies can design and implement effective policies, especially in wind energy and EVs. This way, policymakers can deliberately cultivate reinforcing feedback loops, driving tipping points in the adoption of SETs (Fesenfeld et al., 2022).

Finally, industry players and investors can benefit from understanding the policy landscape and its impact on adoption, contributing to strategic decision-making. This includes investment in R&D, in-frastructure, and market positioning for SET-related companies, or identifying promising projects and effectively allocating resources for investors.

Due to the complexity and dimension of socio-technical systems, policies, technology developments, and sales are often analysed in isolation. However, this research underscores the importance of considering the interactions among the actors in the system. These interactions are the enablers of effective policies, improved technologies, and increased sales. The findings from this investigation suggest that involving a particular set of actors in the development and deployment of sustainable energy technologies is crucial. If the uptake of a technology is lagging and certain actor groups are not engaged, an effective strategy might be to involve these remaining actors. Alternatively, analysing the interactions between the actors and identifying which roles are not being successfully fulfilled could ensure that all relevant actor roles are covered, potentially leading to a positive tipping point in the technology's adoption. The involvement of certain actor groups and the fulfilment of specific roles can be facilitated by different agents within the socio-technical system. For instance, policymakers, technology companies, and trade firms can seek insights and create collaboration opportunities with other actors.

In conclusion, this research provides significant contributions to understanding how positive tipping points can be intentionally triggered in the adoption of SETs. By refining and expanding the feedback loop model of Geels and Ayoub (2023), this thesis addressed key research questions through a detailed analysis of four case studies: offshore wind energy in the UK, onshore wind energy in Portugal, and EVs in the UK and the Netherlands. The study identified critical actors and their roles, extended the range of policy types involved, and proposed new feedback loop models that incorporate additional interactions and actor groups. Despite some limitations, the findings are valuable for both academic research and practical applications. By enhancing our understanding of the dynamics between various actors and their interventions, this thesis offers a theoretical framework and practical insights that can guide policymakers and industry players in accelerating the adoption of SETs. This work ultimately supports the broader goal of achieving a successful energy transition and addressing climate change.

References

- ACEA. (2014). Overview of Purchase and Tax Incentives for Electric Vehicles in the EU (Tech. Rep.). Retrieved from https://www.acea.auto/uploads/publications/Electric_vehicles _overview_2014.pdf
- Agency, E. E. (2016). Monitoring CO2 emissions from new passenger cars and vans in 2015 (Tech. Rep.). Retrieved from https://www.eea.europa.eu/publications/monitoring-co-2 -emissions-from/download
- Agência LUSA. (2004). Serra do Barroso wind farm inaugurated with investment of 20 million euros. RTP Notícias. Retrieved 2024-04-02, from https://www.rtp.pt/noticias/economia/ inaugurado-parque-eolico-da-serra-do-barroso-com-investimento-20-me_n71912
- Alali, L., Niesten, E., & Gagliardi, D. (2022). The impact of UK financial incentives on the adoption of electric fleets: The moderation effect of GDP change. *Transportation Research Part A: Policy and Practice*, *161*, 200-220. Retrieved from https://doi.org/10.1016/j.tra.2022.04.011
- Alkemade, F., & de Coninck, H. (2021). Policy mixes for sustainability transitions must embrace system dynamics. *Environmental Innovation and Societal Transitions*, *41*, 24-26. Retrieved from https://doi.org/10.1016/j.eist.2021.10.014
- ANWB. (2018). *Electric Driving Monitor 2018*. Retrieved 2024-03-24, from https://www.anwb.nl/ belangenbehartiging/duurzaam/elektrisch-rijden-monitor
- ANWB. (2019). *Electric Driving Monitor 2019*. Retrieved 2024-03-24, from https://www.anwb.nl/ belangenbehartiging/duurzaam/elektrisch-rijden-monitor-2019
- ANWB. (2020). Royal Dutch Touring Club ANWB. Retrieved 2024-03-22, from https://www.anwb.nl/ over-anwb/vereniging-en-bedrijf/organisatie/english-page
- ANWB. (2021). *Electric Driving Monitor 2021*. Retrieved 2024-03-24, from https://www.anwb.nl/ belangenbehartiging/duurzaam/elektrisch-rijden-monitor-2021
- APREN. (2024a). *Next events.* Associação Portuguese de Energias Renováveis. Retrieved 2024-04-09, from https://www.apren.pt/pt/apren/eventos-apren
- APREN. (2024b). *Promoters.* Associação Portuguese de Energias Renováveis. Retrieved 2024-04-09, from https://www.apren.pt/pt/associados/associados/1
- APREN. (2024c). *Who we are.* Associação Portuguese de Energias Renováveis. Retrieved 2024-04-09, from https://www.apren.pt/pt/apren/quem-somos
- Ashkrof, P., de Almeida Correia, G. H., & van Arem, B. (2020). Analysis of the effect of charging needs on battery electric vehicle drivers' route choice behaviour: A case study in the Netherlands. *Transportation Research Part D: Transport and Environment*, 78, 102206. Retrieved from https://doi.org/10.1016/j.trd.2019.102206
- Astuti, S. P., Day, R., & Emery, S. B. (2019). A successful fuel transition? Regulatory instruments, markets, and social acceptance in the adoption of modern LPG cooking devices in Indonesia. *Energy Research & Social Science*, 58, 101248. Retrieved from https://doi.org/10.1016/ j.erss.2019.101248
- Auto Siero. (2021). Which environmental zones are there in the Netherlands? Retrieved 2024-03-19, from https://autosiero.nl/welke-milieuzones-zijn-er-in-nederland
- Automotive Council UK. (2024a). Automotive Council Members. Retrieved 2024-02-27, from https://

www.automotivecouncil.co.uk/what-is-the-automotive-council/our-members/

- Automotive Council UK. (2024b). What is the Automotive Council? Retrieved 2024-02-27, from https://www.automotivecouncil.co.uk/what-is-the-automotive-council/
- Barata, F. A., & Quadrado, J. C. (2007, December). Portugal Wind Energy Situation. In *5th WSEAS Int. Conf. on Environemnt, Ecosystems and Development* (p. 186-190). Tenerife, Spain.
- Barclays Bank. (2024). Asset finance services: Funding business-critical assets. Retrieved 2024-03-04, from https://www.barclays.co.uk/business-banking/borrow/asset-finance/
- Becker, J. D., & Brown, C. V. (2000). Industry/Academic Partnerships in Information Systems and Technology. In AMCIS 2000 Proceedings (pp. 1761–1763). Retrieved from https:// aisel.aisnet.org/cgi/viewcontent.cgi?article=1661&context=amcis2000
- BEIS. (2017). The Clean Growth Strategy (Tech. Rep.). London, UK. Retrieved from https://assets.publishing.service.gov.uk/media/5ad5f11ded915d32a3a70c03/ clean-growth-strategy-correction-april-2018.pdf
- BEIS. (2018). Industrial Strategy: Automotive Sector Deal (Tech. Rep.). Retrieved from https://assets.publishing.service.gov.uk/media/5a82b544ed915d74e62373b0/ automotive-sector-deal-single-pages.pdf
- BEIS. (2019). Investment trends IN UK REnewable electricity Landscape (Tech. Rep.). Retrieved from https://assets.publishing.service.gov.uk/media/627e3b3ee90e0721aabad5ce/ CfD_evaluation_phase_1_scoping_phase_-_Investment_trends.pdf
- Bento, N., Borello, M., & Gianfrate, G. (2020). Market-pull policies to promote renewable energy: A quantitative assessment of tendering implementation. *Journal of Cleaner Production*, 248, 119209. Retrieved from https://doi.org/10.1016/j.jclepro.2019.119209
- Besley, T., & Persson, T. (2023). The Political Economics of Green Transitions. *The Quarterly Journal of Economics*, *138*(3), 1863–1906. Retrieved from https://doi.org/10.1093/qje/qjad006
- BIS, & DECC. (2013). Offshore Wind Industrial Strategy Business and Government Action (Tech. Rep.). Retrieved from https://assets.publishing.service.gov.uk/government/uploads/ system/uploads/attachment_data/file/243987/bis-13-1092-offshore-wind-industrial -strategy.pdf
- BMWi. (2020). Renewable energy sources in figures (Tech. Rep.). Retrieved from https://www.bmwk.de/Redaktion/EN/Publikationen/Energie/renewable-energy-sources -in-figures.pdf?__blob=publicationFile&v=1
- Boleat, M. (1996). *Trade association strategy and management*. Association of British Insurers London. Retrieved 2023-10-30, from https://www.boleat.com/materials/trade_assoc_strat _management_96.pdf
- Bossink, B., Blankesteijn, M. L., & Hasanefendic, S. (2023). Upscaling sustainable energy technology: From demonstration to transformation. *Energy Research & Social Science*, *103*, 103208. Retrieved 2023-10-31, from https://doi.org/10.1016/j.erss.2023.103208
- BOVAG. (2023). *About BOVAG*. BOnd Van Automobielhandelaren en Garagehouders. Retrieved 2024-03-21, from https://www.bovag.nl/over-bovag
- Brescia, R. H. (2019). On Tipping Points and Nudges: Review of Cass Sunstein's How Change Happens. *Notre Dame Journal of Law, Ethics and Public Policy*, *34*, 55–78. Retrieved from https://ssrn.com/abstract=3389471
- British Gas. (2024). Government EV charger grants to help you go green. Retrieved 2024-02-28, from https://www.britishgas.co.uk/smart-home/guides/electric-vehicles/EV-grants.html
- Bunce, L., Harris, M., & Burgess, M. (2014). Charge up then charge out? Drivers' perceptions and experiences of electric vehicles in the UK. *Transportation Research Part A*, 59, 278-287. Retrieved from http://dx.doi.org/10.1016/j.tra.2013.12.001

- BVG Associates. (2019). Guide to an offshore wind farm (Tech. Rep.). Retrieved from https:// www.thecrownestate.co.uk/media/2860/guide-to-offshore-wind-farm-2019.pdf
- BVG Associates. (2021). Uk offshore wind history. Catapult. Retrieved 2024-01-08, from https://guidetoanoffshorewindfarm.com/offshore-wind-history
- Campaign for Better Transport. (2022). *Pay-as-you-drive*. Retrieved 2024-03-04, from https:// bettertransport.org.uk/campaigns/pay-as-you-drive/
- CAR Magazine. (2024). The millionth electric car is sold in the UK yet EV take-up is dwindling among private buyers. Retrieved 2024-02-21, from https://www.carmagazine.co.uk/ electric/ev-sales-figures-uk/
- Chai, Q., & Zhang, X. (2010). Technologies and policies for the transition to a sustainable energy system in china. *Energy*, 35, 3995-4002. Retrieved from https://doi.org/10.1016/j.energy .2010.04.033
- Chai, S., & Shih, W. (2016). Bridging Science and Technology through Academic-Industry Partnerships. *Research Policy*, 45(1), 148-158. Retrieved from https://doi.org/10.1016/j.respol.2015 .07.007
- Climate Chance. (2018). Portugal: a blazing energy transition hampered by the resilience of coal (Tech. Rep.). Retrieved from https://www.climate-chance.org/wp-content/uploads/2019/ 03/en_fp5-portugal-electricite_def.pdf
- Climate Watch. (2021). *Historical GHG Emissions*. Retrieved 2024-05-27, from https://www .climatewatchdata.org/ghg-emissions?end_year=2020&start_year=1990
- Connected Papers. (2024). *Explore connected papers in a visual graph.* Retrieved 2024-03-04, from https://www.connectedpapers.com/
- Contestabile, M. (2019). Chapter 16 Batteries, Hydrogen Fuel Cells, and Bioethanol in Passenger Cars. In A. Basile, A. Iulianelli, F. Dalena, & T. N. Veziroğlu (Eds.), *Ethanol* (p. 407-423). Elsevier. Retrieved from https://doi.org/10.1016/B978-0-12-811458-2.00016-X
- Costa, P., Simões, T., Couto, A., & Estanqueiro, A. (2021). IEA Wind Technology Collaboration Programme Portugal 2021 (Tech. Rep.). Laboratório Nacional de Energia e Geologia (LNEG). Retrieved from https://usercontent.one/wp/iea-wind.org/wp-content/uploads/2022/12/ IEA_Wind_TCP_AR2021_Portugal.pdf
- Coventry Live. (2017). First look at £150m Jaguar Land Rover, Tata and WMG building which will create 1,000 jobs. Retrieved 2024-05-07, from https://www.coventrytelegraph.net/news/coventry-news/first-look-150m-jaguar-land-13780684
- Cox, S. (2016). Financial Incentives to Enable Clean Energy Deployment: Policy Overview and Good Practices (Tech. Rep.). National Renewable Energy Laboratory. Retrieved from https://doi .org/10.2172/1239887
- Crielaard, L., Uleman, J. F., Châtel, B. D. L., Epskamp, S., Sloot, P. M. A., & Quax, R. (2022). Refining the Causal Loop Diagram: A Tutorial for Maximizing the Contribution of Domain Expertise in Computational System Dynamics Modeling. *Psychological Methods*, 29(1). Retrieved from http://dx.doi.org/10.1037/met0000484
- De Bruycker, I., & Beyers, J. (2019). Lobbying strategies and success: Inside and outside lobbying in european union legislative politics. *European Political Science Review*, *11*(1), 57–74. Retrieved from https://doi.org/10.1017/S1755773918000218
- DECC. (2009). Offshore Energy Strategic Environmental Assessment: Post Public Consultation Report (Tech. Rep.). Retrieved from https://ctprodstorageaccountp.blob.core.windows.net/ prod-drupal-files/documents/resource/public/Policy-innovation-offshore-wind -report-2020.pdf
- DECC. (2011). UK Renewable Energy Roadmap (Tech. Rep.). Retrieved from

https://assets.publishing.service.gov.uk/media/5a79e9bae5274a18ba50fb84/ 2167-uk-renewable-energy-roadmap.pdf

- DECC. (2013). Final investment decision enabling for renewables: Updates 1, 2 and 3. GOV.UK. Retrieved 2024-03-06, from https://www.gov.uk/government/publications/ increasing-certainty-for-investors-in-renewable-electricity-final-investment -decision-enabling-for-renewables
- de Jongh, M. (2022). Discover electric driving. Natuur en Milieu. Retrieved 2024-03-26, from https://natuurenmilieu.nl/themas/vervoer-reizen/elektrisch-rijden/ontdek -elektrisch-rijden/
- Delicado, A., Silva, L., Junqueira, L., Horta, A., Fonseca, S., & Truninger, M. (2013). Environment, landscape, heritage, and economy: Conflicts surrounding wind farms in Portugal. *Revista Critica de Ciencias Sociais*, *100*(100), 11-36. Retrieved from http://dx.doi.org/10.4000/rccs.5198
- Deloitte Digital. (2024). Ten Types of Innovation: The discipline of building breakthroughs. Retrieved 2024-05-02, from https://www.deloittedigital.com/us/en/capabilities/creative -experience-design/applied-design-innovation/ten-types.html
- Department for Environment, Food & Rural Affairs. (2020). *Clean air zones*. Author. Retrieved 2024-02-29, from https://www.gov.uk/guidance/driving-in-a-clean-air-zone#cities-with-clean -air-zones
- Department for Transport. (2017). Consultation on amendments to permissible vehicle weights and dimensions, including to incentivise cleaner fuel technologies, and other associated proposals: Summary of Responses (Tech. Rep.). Retrieved from https://assets.publishing.service .gov.uk/media/5a8041e040f0b6230269271d/consultation-on-amendments-to-permissible -vehicle-weights-and-dimensions-summary-of-responses.pdf
- Department for Transport. (2021). Electric vehicle smart charging consultation: summary of responses: Executive Summary. GOV.UK. Retrieved 2024-05-07, from https://www.gov.uk/ government/consultations/electric-vehicle-smart-charging/public-feedback/ electric-vehicle-smart-charging-consultation-summary-of-responses
- Department for Transport, & OZEV. (2020). Transitioning to zero emission cars and vans: 2035 delivery
 plan (Tech. Rep.). London, UK. Retrieved from https://assets.publishing.service.gov.uk/
 media/60f9a3918fa8f5042aecd384/transitioning-to-zero-emission-cars-vans-2035
 -delivery-plan.pdf
- Deuten, S., Vilchez, J. J. G., & Thiel, C. (2020). Analysis and testing of electric car incentive scenarios in the Netherlands and Norway. *Technological Forecasting and Social Change*, *151*, 119847. Retrieved from https://doi.org/10.1016/j.techfore.2019.119847
- DfT. (2023). Electric vehicle charging device statistics: January 2023. Department for Transport. Retrieved 2024-02-22, from https://www.gov.uk/government/statistics/ electric-vehicle-charging-device-statistics-january-2023/electric-vehicle -charging-device-statistics-january-2023
- Dubois, M., & Eyckmans, J. (2014). Chapter 35 economic instruments. In E. Worrell & M. A. Reuter (Eds.), *Handbook of recycling* (p. 511-519). Boston: Elsevier. Retrieved from https://doi.org/ 10.1016/B978-0-12-396459-5.00035-0
- DUT. (2024). Supporting Us. Retrieved 2024-05-08, from https://www.fsteamdelft.nl/partners
- Eder, C., & Stadelmann-Steffen, I. (2023). Bringing the political system (back) into social tipping relevant to sustainability. *Energy Policy*, 177, 113529. Retrieved from https://doi.org/10.1016/j.enpol.2023.113529
- Edmondson, D. L., Kern, F., & Rogge, K. S. (2019). The co-evolution of policy mixes and socio-technical systems: Towards a conceptual framework of policy mix feedback in sustainability transitions.
Research Policy, *10*, 103555. Retrieved from https://doi.org/10.1016/j.respol.2018.03 .010

- EIB. (2007). 1986-2006: 20 years of EIB in Portugal (Tech. Rep.). Retrieved from https://www.eib .org/attachments/country/portugal_20years_pt.pdf
- Ekim, Z., Mattsson, P., & Bernardo, R. (2023). Assessments of users' interactions with energy-efficient solutions: A systematic review. *Building and Environment*, 242, 110522. Retrieved 2023-11-01, from https://doi.org/10.1016/j.buildenv.2023.110522
- ElaadNL. (2022a). About us. Retrieved 2024-03-12, from https://elaad.nl/over-ons/
- ElaadNL. (2022b). About us. Retrieved 2024-03-19, from https://elaad.nl/over-ons/
- Elicit. (2024). *Analyze research papers at superhuman speed.* ScienceDirect. Retrieved 2023-11-15, from https://elicit.com/
- ENERCON. (2024). The ENERCON Story. Retrieved 2024-04-10, from https://www.enercon.de/ en/company/enercon-story
- Entsalo, H., Kalimo, H., Kautto, P., & Turunen, T. (2023). Analysing regulatory instruments in sustainability transitions: A combined 'intervention points' and 'roles of law' approach to the European Union's Ecodesign framework. *Sustainable Production and Consumption*, *42*, 125-137. Retrieved from https://doi.org/10.1016/j.spc.2023.09.013
- European Commission. (2020). Interreg 30 Years. Retrieved 2024-03-20, from https://ec.europa .eu/regional_policy/policy/cooperation/european-territorial/30-years_en
- European Commission. (2021). Development of eu ets (2005-2020). Retrieved 2024-03-11, from https://climate.ec.europa.eu/eu-action/eu-emissions-trading-system-eu -ets/development-eu-ets-2005-2020_en
- European Commission. (2022). 30 years of bringing green ideas to LIFE. Retrieved 2024-03-20, from https://cinea.ec.europa.eu/programmes/life/history-life_en
- European Investment Bank. (2022). When shopping for a new car, the vast majority of Dutch people say they will opt for a hybrid or electric vehicle. Retrieved 2024-03-25, from https://www.eib.org/en/press/all/2022-050-when-shopping-for-a-new-car-the-vast -majority-of-dutch-people-say-they-will-opt-for-a-hybrid-or-electric-vehicle
- EVA. (2024). About EVA England. Retrieved 2024-05-07, from https://www.evaengland.org.uk/
- Falchetta, G., Michoud, B., Hafner, M., & Rother, M. (2022). Harnessing finance for a new era of decentralised electricity access: A review of private investment patterns and emerging business models. *Energy Research & Social Science*, 90, 102587. Retrieved 2023-11-01, from https:// doi.org/10.1016/j.erss.2022.102587
- Fang, L., Li, Y., & Govindan, K. (2024). Entry mode selection for a new entrant of the electric vehicle automaker. *European Journal of Operational Research*, 313, 270-280. Retrieved 2023-10-31, from https://doi.org/10.1016/j.ejor.2023.08.014
- Farmer, J. D., Hepburn, C., Ives, M. C., Hale, T., Wetzer, T., Mealy, P., ... Way, R. (2019). Sensitive intervention points in the post-carbon transition. *Science*, *364*(6436), 132-134. Retrieved 2023-10-30, from https://doi.org/10.1126/science.aaw7287
- Fastned. (2019). What makes us us. Retrieved 2024-05-08, from https://www.fastnedcharging .com/nl/over-ons/over-fastned
- Ferreira, J. R., & Martins, F. R. (2009, January). Winds of Change. Wind Energy in Portugal. In Cape Verde: Networks and Regional Development (pp. 388–408). Santiago Island, Cape Verde. Retrieved from https://www.researchgate.net/publication/277032583_Ventos_de_Mudanca_-_A_Energia_Eolica_em_Port
- Ferreira, L. (2008). Enercon strengthens its leadership in the wind market, and Enersis gives way to Enernova. Público. Retrieved 2024-04-09, from https://www.publico.pt/2008/

02/07/jornal/enercon-reforca-lideranca-no-mercado-eolico-e-enersis-cede-lugar-a -enernova-248076

- Ferreira, P., Araújo, M., & O'Kelly, M. E. J. (2006). An overview of the Portuguese wind power sector (Tech. Rep.). Retrieved from https://repositorium.sdum.uminho.pt/bitstream/1822/5381/ 1/ITOR_PF.pdf
- Fesenfeld, L., Schmid, N., Finger, R., Mathys, A., & Schmidt, T. S. (2022). The politics of enabling tipping points for sustainable development. *One Earth*, 5(10), 1100-1108. Retrieved from https://doi.org/10.1016/j.oneear.2022.09.004
- Fitzgerald, L. (1999). Case studies as a research tool. *Quality in Health Care*, *8*, 75. Retrieved from https://doi.org/10.1136/qshc.8.2.75
- Fleet News. (2016). Go Ultra Low cities likely to increase fleet electric vehicle adoption. Retrieved 2024-02-29, from https://www.fleetnews.co.uk/news/environment/2016/10/12/go -ultra-low-cities-likely-to-nudge-up-fleet-electric-vehicle-adoption
- Fraser, S. D., Bishop, J., Gupta, R., Mancl, D., Ramming, J. C., & Rivas, S. (2011, October). Industryacademic research partnerships. In OOPSLA '11: Proceedings of the ACM international conference companion on Object oriented programming systems languages and applications companion (pp. 175–178). Retrieved from https://doi.org/10.1145/2048147.2048194
- Friends of the Earth. (2012). Wind power: your questions answered (Tech. Rep.). Retrieved from https://friendsoftheearth.uk/sites/default/files/downloads/wind_power _your_questions.pdf
- Frontier Economics. (2013). International support for onshore wind (Tech. Rep.). Retrieved
 from https://assets.publishing.service.gov.uk/media/5a75acc140f0b67f59fcebab/
 international_support_onshore_wind_frontier.pdf
- Frumkin, H. (2022). Hope, Health, and the Climate Crisis. *The Journal of Climate Change and Health*, 5, 100115. Retrieved from https://doi.org/10.1016/j.joclim.2022.100115
- Garratt, E. (2024). Our guide to the Ultra Low Emission Zone. The Electric Car Scheme. Retrieved 2024-02-29, from https://www.electriccarscheme.com/blog/our-guide-to-the-ultra-low -emission-zone
- Geels, F. W. (2002). Technological transitions as evolutionary reconfiguration processes: a multilevel perspective and a case-study. *Research Policy*, *31*, 1257–1274. Retrieved from https:// doi.org/10.1016/S0048-7333(02)00062-8
- Geels, F. W. (2019). Socio-technical transitions to sustainability: a review of criticisms and elaborations of the multi-level perspective. *Current Opinion in Environmental Sustainability*, 39, 187-201. Retrieved from https://doi.org/10.1016/j.cosust.2019.06.009
- Geels, F. W., & Ayoub, M. (2023). A socio-technical transition perspective on positive tipping points in climate change mitigation: Analysing seven interacting feedback loops in offshore wind and electric vehicles acceleration. *Technological Forecasting and Social Change*, 193. Retrieved from https://doi.org/10.1016/j.techfore.2023.122639
- Geels, F. W., Sovacool, B. K., Schwanen, T., & Sorrell, S. (2017). Sociotechnical transitions for deep decarbonization. *Science*, 357(6357), 1242-1244. Retrieved from https://ssrn.com/abstract= 3447276
- Geels, F. W., & Turnheim, B. (2022). The Great Reconfiguration: A Socio-Technical Analysis of Low-Carbon Transitions in UK Electricity, Heat, and Mobility Systems. Cambridge: Cambridge University Press. Retrieved 2023-10-30, from http://dx.doi.org/10.1017/9781009198233
- Global Underwater Hub. (2024). About GHU. Retrieved 2024-03-08, from https://www .globalunderwaterhub.com/about/
- Global Wind Energy Council. (2023). What we do. Author. Retrieved 2023-10-26, from https://

gwec.net/global-wind-energy-council/what-is-gwec/

- Green Investment Group. (2023). Who we are measuring our impact. Author. Retrieved 2023-10-29, from https://www.greeninvestmentgroup.com/en/who-we-are/measuring-our -impact.html
- Green Investment Group. (2024). Who we are. Retrieved 2024-03-08, from https://www .greeninvestmentgroup.com/en/who-we-are/our-mission.html
- Greenpeace. (2018). Anti-diesel campaigners and medics block hundreds of volkswagen staff from head office. Retrieved 2024-03-01, from https://www.greenpeace.org.uk/news/anti-diesel -campaigners-medics-block-hundreds-volkswagen-staff-head-office/
- Greenpeace. (2020). The 2030 ban on new petrol and diesel cars, explained. Retrieved 2024-03-04, from https://www.greenpeace.org.uk/news/petrol-diesel-cars-vans-2030-ban-phase -out/
- Gregersen, B., & Johnson, B. (2010, June). Stimulating emerging sustainable energy technologies through policy learning. In *13th conference of the international joseph a. schumpeter society: Innovation, organisation, sustainability and crises* (p. 1-17). Aalborg, Denmark.
- Grubler, A., Wilson, C., & Nemet, G. (2016). Apples, oranges, and consisten comparisons of the temporal dynamics of energy transitions. *Energy Research & Social Science*, *22*, 18-25. Retrieved from https://doi.org/10.1016/j.erss.2016.08.015
- GWEC. (2023). Global Offshore Wind Report 2023 (Tech. Rep.). Global Wind Energy Council. Retrieved from https://gwec.net/wp-content/uploads/2023/08/GWEC-Global-Offshore -Wind-Report-2023.pdf
- Hall, D., Wappelhorst, S., Mock, P., & Lutsey, N. (2020). European Electric Vehicle Factbook 2019/2020 (Tech. Rep.). Retrieved from https://theicct.org/sites/default/files/publications/EV -EU-Factbook-2020.pdf
- Herman, K. S. (2013). *The Green Job Engine in Portugal* (Tech. Rep.). Retrieved from http://works.bepress.com/kyle_herman/14/
- Higgins, P., & Foley, A. (2014). The evolution of offshore wind power in the United Kingdom. *Renewable* and Sustainable Energy Reviews, 37, 599-612. Retrieved from http://dx.doi.org/10.1016/ j.rser.2014.05.058
- Hinestroza-Olascuaga, L. M., Carvalho, P. M., & de Jesus, C. M. C. (2023). Lowering risk intolerance to unlock private investments in renewable energy-based rural electrification. *Energy for Sustainable Development*, 74, 258-268. Retrieved 2023-11-01, from https://doi.org/10.1016/j.esd.2023 .02.011
- HM Government. (2014). Overview of Support for the Offshore Wind Industry (Tech. Rep.). Retrieved from https://assets.publishing.service.gov.uk/media/5a7dc7d3ed915d2ac884d82c/ bis-14-880-support-for-the-offshore-wind-industry-overview.pdf
- HM Revenue & Customs. (2018). Carbon price floor (Tech. Rep.). Retrieved from https://assets.publishing.service.gov.uk/media/5a7ab68fed915d670dd7e120/ carbon_price_floor.pdf.pdf
- Huergo, E., & Moreno, L. (2014). National or international public funding? Subsidies or loans? Evaluating the innovation impact of R&D support programmes (Tech. Rep.). Universidad Complutense de Madrid. Retrieved from https://mpra.ub.uni-muenchen.de/54218/1/MPRA_paper_54218.pdf
- ICCT. (2014). Eu co2 emission standards for passenger cars and light-commercial vehicles (Tech. Rep.). Retrieved from https://theicct.org/sites/default/files/publications/ ICCTupdate_EU-95gram_jan2014.pdf
- ICCT. (2023). Battery electric and plug-in hybrid vehicle uptake in European cities (Tech. Rep.). Retrieved from https://theicct.org/wp-content/uploads/2023/02/bev-phev

-european-cities-mar23.pdf

- IEA. (2002). IEA Wind Energy Annual Report 2001 (Tech. Rep.). Retrieved from https://iea-wind .org/wp-content/uploads/2023/02/IEAWIND-AR2001.pdf
- IEA. (2003). IEA Wind Energy Annual Report 2002 (Tech. Rep.). Retrieved from https://iea-wind .org/wp-content/uploads/2022/12/2002-IEA-Annual-Report.pdf
- IEA. (2004a). Energy Policies of IEA Countries: Portugal 2004 Review (Tech. Rep.). Retrieved from https://iea.blob.core.windows.net/assets/0ae791e2-81e1-477c-9b56-80e1755cf444/ EnergyPoliciesofIEACountriesPortugal2004.pdf
- IEA. (2004b). IEA Wind Energy Annual Report 2003 (Tech. Rep.). Retrieved from https://iea-wind .org/wp-content/uploads/2023/02/IEAWIND-AR2003.pdf
- IEA. (2005). IEA Wind Energy Annual Report 2004 (Tech. Rep.). Retrieved from https://iea-wind .org/wp-content/uploads/2023/02/IEAWIND-AR2004.pdf
- IEA. (2006a). IEA Wind Energy Annual Report 2005 (Tech. Rep.). Retrieved from https://iea-wind .org/wp-content/uploads/2022/12/2005_IEA-Annual-Report.pdf
- IEA. (2006b). IEA Wind Energy Annual Report 2006 (Tech. Rep.). Retrieved from https://iea-wind .org/wp-content/uploads/2022/12/2006-IEA-Annual-Report.pdf
- IEA. (2008). IEA Wind Energy Annual Report 2007 (Tech. Rep.). Retrieved from https://iea-wind .org/wp-content/uploads/2022/12/2007-IEA-Annual-Report.pdf
- IEA. (2009a). Energy Policies of IEA Countries: Portugal 2009 Review (Tech. Rep.). Retrieved from https://iea.blob.core.windows.net/assets/557b4675-6044-4c75-b060-2315a7d89c19/ Portugal2009.pdf
- IEA. (2009b). IEA commends Portugal on effective implementation of energy policy but cautions that further challenges remain. International Energy Agency. Retrieved 2024-03-28, from https://www.iea.org/news/iea-commends-portugal-on-effective -implementation-of-energy-policy-but-cautions-that-further-challenges-remain
- IEA. (2014). Offshore wind capital grants scheme. Retrieved 2024-03-11, from https://www.iea.org/ policies/3791-offshore-wind-capital-grants-scheme
- IEA. (2022). Global EV Data Explorer: EV stock, cars, United Kingdom, 2010-2022. Internation Energy Agency. Retrieved 2024-01-08, from https://www.iea.org/data-and-statistics/ data-tools/global-ev-data-explorer
- IEA. (2023a). Global electric car stock, 2010-2022. International Energy Agency. Retrieved 2024-04-14, from https://www.iea.org/data-and-statistics/charts/global-electric-car-stock -2010-2022
- IEA. (2023b). Global EV Outlook 2023: Catching up with climate ambitions (Tech. Rep.). Internation Energy Agency. Retrieved from https://iea.blob.core.windows.net/assets/dacf14d2-eabc -498a-8263-9f97fd5dc327/GEV02023.pdf
- IEA. (2023c). Tracking clean energy progress 2023. IEA. Retrieved 2023-09-27, from https://
 www.iea.org/reports/tracking-clean-energy-progress-2023
- IEA. (2024a). *Portugal: Energy Mix.* International Energy Agency. Retrieved 2024-03-28, from https://www.iea.org/countries/portugal/energy-mix
- IEA. (2024b). *Portugal: Overview.* International Energy Agency. Retrieved 2024-03-28, from https://www.iea.org/countries/portugal
- Innovate UK. (2024). Low emission vehicles. Retrieved 2024-03-04, from https://iuk.ktn-uk.org/ transport/low-emission-vehicles/
- IPCC. (2023a). Longer Report. Geneva, Switzerland: IPCC. Retrieved from https://doi.org/ 10.59327/IPCC/AR6-9789291691647.002
- IPCC. (2023b). Summary for Policymakers. Geneva, Switzerland: IPCC. Retrieved from https://

doi.org/10.59327/IPCC/AR6-9789291691647.001

- IPO. (2024). About the IPO. Interprovinciaal Overleg. Retrieved 2024-03-12, from https://www.ipo .nl/over-het-ipo/
- IRENA. (2013a). 30 Years of Policies for Wind Energy: Lessons from Denmark (Tech. Rep.). Retrieved from https://www.irena.org/-/media/Files/IRENA/Agency/Publication/ 2013/GWEC/GWEC_Denmark.pdf?la=en&hash=C14BEEC4FFEEBA20B2B1928582AA23931F092F48
- IRENA. (2013b). 30 Years of Policies for Wind Energy: Lessons from Portugal (Tech. Rep.). Retrieved from https://www.irena.org/-/media/Files/IRENA/Agency/Publication/ 2013/GWEC/GWEC_Portugal.pdf?la=en&hash=9AC1FCF27CBEF69846F1FDDBCE5434E2DD79267F
- ISEP. (2024). *Master's in Sustainable Energies*. Instituto Superior de Engenharia do Porto. Retrieved 2024-04-09, from https://www.isep.ipp.pt/Course/Course/52
- ISO. (2024). *Standards*. International Organization for Standardization. Retrieved 2024-05-28, from https://www.iso.org/standards.html
- IST. (2024). Bologna Master's in Engineering and Energy Management: Wind Energy. Instituto Superior Técnico. Retrieved 2024-04-09, from https://fenix.tecnico.ulisboa.pt/cursos/mege/ disciplina-curricular/845953938490326
- Jaffe, A. B., & Stavins, R. N. (1994). Energy-Efficiency Investments and Public Policy. *The Energy Journal*, *15*(2), 43-66. Retrieved from http://dx.doi.org/10.5547/ISSN0195-6574-EJ-Vol15 -No2-3
- Jennings, T., Tipper, H. A., Daglish, J., Grubb, M., & Drummond, P. (2020). Policy, innovation and cost reduction in UK offshore wind (Tech. Rep.). Retrieved from https://ctprodstorageaccountp.blob.core.windows.net/prod-drupal-files/documents/ resource/public/Policy-innovation-offshore-wind-report-2020.pdf
- Johansson, M., Langlet, D., Larsson, O., L²ofgren, A., Harring, N., & Jagers, S. (2021). A risk framework for optimising policies for deep decarbonisation technologies. *Energy Research & Social Science*, 82, 102297. Retrieved from https://doi.org/10.1016/j.erss.2021.102297
- Joskow, P. L., & Schmalensee, R. (1998). The Political Economy of Market⊡Based Environmental Policy: the U.S. Acid Rain Program. *The Journal of Law & Economics*, *41*(1), 37-84. Retrieved from https://doi.org/10.1086/467384
- Junk, W. M. (2016). Two logics of ngo advocacy: understanding inside and outside lobbying on eu environmental policies. *Journal of European Public Policy*, *23*(2), 236-254. Retrieved from http://dx.doi.org/10.1080/13501763.2015.1041416
- Kamata, A. S., Khosla, R., & Narayanamurti, V. (2020). Illuminating homes with leds in india: Rapid market creation towards low-carbon technology transition in a developing country. *Energy Research & Social Science*, 66, 101488. Retrieved 2023-10-11, from https://doi.org/10.1016/ j.erss.2020.101488
- Karttunen, E., Tsytsyna, E., Lintukangas, K., Rintala, A., Abdulkareem, M., Havukainen, J., & Nuortila-Jokinen, J. (2021). Toward environmental innovation in the cement industry: A multiple-case study of incumbents and new entrants. *Journal of Cleaner Production*, *314*, 127981. Retrieved 2023-10-31, from https://doi.org/10.1016/j.jclepro.2021.127981
- Kekäle, T., & Helo, P. (2014). The Tipping Points of Technology Development. *Quality Innovation prosperity*, *18*(1), 1–14. Retrieved from https://doi.org/10.12776/QIP.V18I1.211
- Kern, F., Smith, A., Shaw, C., Raven, R., & Verhees, B. (2014). From laggard to leader: Explaining offshore wind developments in the UK. *Energy Policy*, 69, 635-646. Retrieved 2024-01-15, from https://dx.doi.org/10.1016/j.enpol.2014.02.031
- KiM. (2022). Explanations for the differences in car ownership among Dutch households (Tech. Rep.). Retrieved from https://shorturl.at/o3vCr

- Kirby, A. J. (1988). Trade associations as information exchange mechanisms. The RAND Journal of Economics, 19(1), 138-146. Retrieved 2023-10-30, from https://www.jstor.org/stable/ 2555403
- Klimaatakkoord. (2021). Agreements for Mobility. Retrieved 2024-03-12, from https://www .klimaatakkoord.nl/mobiliteit
- Knol, A. (2018). Electric cars are only part of the solution. Milieu Defensie. Retrieved 2024-03-25, from https://milieudefensie.nl/actueel/elektrische-auto-is-maar -een-deel-van-de-oplossing
- Kochenkova, A., Grimaldi, R., & Munari, F. (2015). Public policy measures in support of knowledge transfer activities: a review of academic literature. *The Journal of Technology Transfer*, *41*, 407-429. Retrieved from https://doi.org/10.1007/s10961-015-9416-9
- Lamontagne, S., Carvalho, M., Frejinger, E., Gendron, B., Anjos, M. F., & Atallah, R. (2023). Optimising Electric Vehicle Charging Station Placement Using Advanced Discrete Choice Models. *INFORMS journal on computing*, 35(5), 1195-1213. Retrieved from https://doi.org/10.1287/ijoc.2022 .0185
- Lehmann, M. (2006). Government–Business Relationships Through Partnerships for Sustainable Development: the Green Network in Denmark. *Journal of Environmental Policy & Planning*, 8(3), 235-257. Retrieved from http://dx.doi.org/10.1080/15239080600915592
- Lenton, T. M., Benson, S., Smith, T., Ewer, T., Lanel, V., Petykowski, E., ... Sharpe, S. (2022). Operationalising positive tipping points towards global sustainability. *Global Sustainability*, *5*, 1-16. Retrieved from http://dx.doi.org/10.1017/sus.2021.30
- Lenton, T. M., McKay, D. I. A., Loriani, S., Abrams, J. F., Lade, S. J., Donges, J. F., ... Laybourn, L. (2023). *Global Tipping Points Report 2023* (Tech. Rep.). Exeter, UK. Retrieved from https:// global-tipping-points.org/
- Lenton, T. M., Rockström, J., Gaffney, O., Rahmstorf, S., Richardson, K., Steffen, W., & Schellnhuber, H. J. (2008). Tipping elements in the earth's climate system. *PNAS*, *105*(6), 1786-1793. Retrieved from http://dx.doi.org/10.1073/pnas.0705414105
- Lloyds Bank. (2024). *Electric vehicles*. Retrieved 2024-03-04, from https://www.lloydsbank.com/ car-finance/electric-vehicles.html
- Loughborough University. (2019). Loughborough University is working on a collaborative industry research project to develop digital engineering tools that will help bring the next generation of electrified commercial vehicles to the market much faster. Retrieved 2024-05-07, from https://www.lboro.ac.uk/news-events/news/2019/october/vivd -project-looking-at-next-gen-electric-vehicles/
- Mazur, C., Contestabile, M., Offer, G. J., & Brandon, N. P. (2015). Assessing and comparing german and uk transition policies for electric mobility. *Environmental Innovation and Societal Transitions*, *14*, 84-100. Retrieved 2023-10-11, from https://doi.org/10.1016/j.eist.2014.04.005
- McNally, P. (2022). An Efficient Energy Transition: Lessons From the UK's Offshore Wind Rollout (Tech. Rep.). Tony Blair Institute for Global Change. Retrieved from https://assets.ctfassets .net/75ila1cntaeh/3Po8s5i1nntFokfzc6loSz/d8337ecf0cf72381a879c6dbad648d98/ An-Efficient-Energy-Transition-Lessons-From-the-UK-s-Offshore-Wind-Rollout.pdf
- Mercure, J.-F., Sharpe, S., Vinuales, J. E., Ives, M., Grubb, M., Lam, A., ... Nijsse, F. J. M. M. (2021). Risk-opportunity analysis for transformative policy design and appraisal. *Global Environmental Change*, 70, 102359. Retrieved from https://doi.org/10.1016/j.gloenvcha.2021.102359
- Metcalfe, S., & Sasse, T. (2023). The development of the UK's offshore wind sector 2010–16 (Tech. Rep.). Institute for Government. Retrieved from https://www.instituteforgovernment.org .uk/sites/default/files/2023-10/Case-study-offshore-wind-2010-2016.pdf

- Milieu Centraal. (2024). *Everything about electric cars*. Retrieved 2024-03-12, from https://www.milieucentraal.nl/duurzaam-vervoer/elektrische-auto/alles-over -elektrische-auto/
- Ming, Z., Yingxin, L., Shaojie, O., Hui, S., & Chunxue, L. (2016). Nuclear energy in the Post-Fukushima Era: Research on the developments of the Chinese and worldwide nuclear power industries. *Renewable and Sustainable Energy Reviews*, 58, 147-156. Retrieved from https://doi.org/ 10.1016/j.rser.2015.12.165
- MITECO. (2005). Summary of the Renewable Energy Plan in Spain 2005-2010 (Tech. Rep.). Retrieved from https://www.miteco.gob.es/content/dam/miteco/es/energia/files -1/desarrollo/EnergiaRenovable/Plan/Documents/ResumenPlanEnergiasRenov.pdf
- Muratori, M., Alexander, M., Arent, D., Bazilian, M., Cazzola, P., Dede, E. M., ... Ward, J. (2021). The rise of electric vehicles 2020 status and future expectations. *Progress in Energy*, *3*(2). Retrieved from https://doi.org/10.1088/2516-1083/abe0ad
- Nationale Agenda Laadinfrastructuur. (2022). Dutch National Charging Infrastructure Agenda (Tech. Rep.). Retrieved from https://nederlandelektrisch.nl/u/images/nal-english-may-2023 .pdf
- Nederland Elektrisch. (2019). Evaluation report Green Deal Electric Transport (Tech. Rep.). Retrieved from https://evkenniscentrum.nl/Uploads/Editor/Evaluatie%20Green%20Deal% 20Elektrisch%20Vervoer%202016-2020.pdf
- Nederland Elektrisch. (2020). The national government is investing in the construction of charging stations with 6 regions: 213 charging points will be needed per working day next year. Retrieved 2024-03-11, from https://nederlandelektrisch.nl/actueel/nieuwsoverzicht/ i1493/rijksoverheid-investeert-met-6-regio-s-in-aanleg-laadpalen-komend-jaar-213 -laadpunten-per-werkdag-nodig
- Nederland Elektrisch. (2022). Formula E-Team: businesses, knowledge institutions and the government are working together on electric transport. Retrieved 2024-03-11, from https:// nederlandelektrisch.nl/formule-e-team
- Nederland Elektrisch. (2023). Charging infrastructure. Retrieved 2024-03-11, from https:// nederlandelektrisch.nl/charging-infrastructure
- Nederland Elektrisch. (2024a). National policy on electric driving. Retrieved 2024-03-12, from mhttps://nederlandelektrisch.nl/e-rijden-in-mijn-gemeente/rijksbeleid -elektrisch-rijden
- Nederland Elektrisch. (2024b). Subsidies and Schemes. Retrieved 2024-03-11, from https:// nederlandelektrisch.nl/subsidies-financiering/subsidies-en-regelingen
- Netbeheer Nederland. (2024). *Working together with the energy system*. Retrieved 2024-03-12, from https://www.netbeheernederland.nl/
- Newell, P., & Simms, A. (2021). How Did We Do That? Histories and Political Economies of Rapid and Just Transitions. New Political Economy, 26(6), 907-922. Retrieved from http://dx.doi.org/ 10.1080/13563467.2020.1810216
- Newell, P., Twena, M., & Daley, F. (2021). Scaling behaviour change for a 1.5-degree world: challenges and opportunities. *Global Sustainability*, *4*(e22), 1–13. Retrieved from https://doi.org/10 .1017/sus.2021.23
- Nunes, A. N. (2018). Energy changes in Portugal. An Overview of the Last Century. Journal of Mediterranean geography, 130, 1-19. Retrieved from https://doi.org/10.4000/mediterranee .10113
- Offshore Energies UK. (2024). What we do. Retrieved 2024-05-07, from https://oeuk.org.uk/ who-we-are/what-we-do/

- Ofgem. (2013). Monitoring the 'connect and manage' electricity grid access regime (Tech. Rep.). Retrieved from https://www.ofgem.gov.uk/sites/default/files/docs/2013/12/connect_and _manage_ar_2013_final_051213.pdf
- OLEV. (2014). £35m Go Ultra Low City Scheme (Tech. Rep.). London, UK. Retrieved from https://assets.publishing.service.gov.uk/media/5a74e1a240f0b65c0e8453c9/ 141211_City_Scheme_Guidance_Final.pdf
- ORE Catapult. (2022). About ORE Catapult. Retrieved 2024-05-07, from https://ore.catapult.org .uk/about/
- Orset, C. (2021). Is information a good policy instrument to influence the energy behaviour of households? *Energy Economics*, 102, 105451. Retrieved from https://doi.org/10.1016/ j.eneco.2021.105451
- Ortt, J. R., & Kamp, L. M. (2022). A technological innovation system framework to formulate niche introduction strategies for companies prior to large-scale diffusion. *Technological Forecasting* & Social Change, 180, 121671. Retrieved from https://doi.org/10.1016/j.techfore.2022 .121671
- Ortt, J. R., Langley, D. J., & Pals, N. (2013). Ten niche strategies to commercialize new high-tech products. In 2013 International Conference on Engineering, Technology and Innovation (ICE) & IEEE International Technology Management Conference (p. 1-12). Retrieved from https:// doi.org/10.1109/ITMC.2013.7352687
- Otto, I. M., Donges, J. F., Cremades, R., Bhowmik, A., Hewitt, R. J., Lucht, W., ... Schellnhuber, H. J. (2020). Social tipping dynamics for stabilizing Earth's climate by 2050. *PNAS*, *117*(5), 2354–2365. Retrieved from https://doi.org/10.1073/pnas.1900577117
- OWIC. (2021). *About owic*. Offshore Wind Industry Council. Retrieved 2024-03-08, from https://www.owic.org.uk/
- OZEV. (2016a). On-Street Residential Chargepoint Scheme guidance for local authorities. Office for Zero Emission Vehicles. Retrieved 2024-02-29, from https://shorturl.at/eGus6
- OZEV. (2016b). Workplace Charging Scheme: grant claim form for installing charging equipment. Office for Zero Emission Vehicles. Retrieved 2024-02-29, from https://www.gov.uk/government/ publications/workplace-charging-scheme-grant-claim-form-for-installing-charging -equipment
- Panayotou, T. (1995). Economic instruments for environmental management and sustainable development (Tech. Rep.). International Environment Program Harvard Institute for International Development Harvard University. Retrieved from https://www.cbd.int/doc/nbsap/finance/ Panayotou1994EconInstEnvMgSusDev_199EcInsEnvMgSusDev.pdf
- Paradies, G. L., Usmani, O. A., Lamboo, S., & van den Brink, R. W. (2023). Falling short in 2030: Simulating battery-electric vehicle adoption behaviour in the Netherlands. *Energy Research & Social Science*, 97, 102968. Retrieved from https://doi.org/10.1016/j.erss.2023.102968
- Patil, P. (2019). Electric Vehicle Charging Infrastructure: Current Status, Challenges, and Future Developments. *International Journal of Intelligent Automation and Computing*, 2(1), 1-12. Retrieved from https://research.tensorgate.org/index.php/IJIAC/article/view/25/23
- Penders, D., & de Lange, W. (2022). Electric cars and batteries: how sustainable are they? Natuur en Milieu. Retrieved 2024-03-26, from https://natuurenmilieu.nl/publicatie/elektrische -autos-en-accus-dit-is-wat-je-moet-weten/
- Pinto, L. (2005). Wind cluster with investment of 1200 million euros. Público. Retrieved 2024-03-29, from https://www.publico.pt/2005/10/21/jornal/cluster-eolico-com-investimento -de-1200-milhoes-de-euros-44909
- Quercus. (2024). Energy and Climate Change. Retrieved 2024-04-09, from https://quercus.pt/

energia-e-alteracoes-climaticas/

- RAI Vereniging. (2024). We are RAI Association. Retrieved 2024-03-21, from https://www .raivereniging.nl/
- Recharge UK. (2024). *Electric vehicles*. Retrieved 2024-03-04, from https://www.r-e-a.net/our -members-forum/electric-vehicles/
- Renewable Energy Association. (2023). *Wind*. Retrieved 2024-05-07, from https://www.r-e-a.net/ technologies/wind/
- RenewableUK. (2012). Offshore Wind Cost Reduction Task Force Report (Tech. Rep.). Retrieved
 from https://assets.publishing.service.gov.uk/government/uploads/system/uploads/
 attachment_data/file/66776/5584-offshore-wind-cost-reduction-task-force-report
 .pdf
- RenewableUK. (2023). *New renewable electricity generation records confirmed in government's energy bible.* Author. Retrieved 2024-01-10, from https://shorturl.at/bg2iP
- RenewableUK. (2024). *Wind energy statistics*. Author. Retrieved 2024-01-10, from https://www .renewableuk.com/page/UKWEDhome
- Ribeiro, F., Ferreira, P., Araújo, M., & Braga, A. C. (2014). Public opinion on renewable energy technologies in Portugal. *Energy*, 69, 39-50. Retrieved from https://doi.org/10.1016/j.energy.2013.10.074
- Richardson, K., Steffen, W., Lucht, W., Bendtsen, J., Cornell, S. E., Donges, J. F., ... Rockström, J. (2023). Earth beyond six of nine planetary boundaries. *Science Advances*, 9(37), eadh2458. Retrieved from https://doi.org/10.1126/sciadv.adh2458
- Rietmann, N., & Lieven, T. (2019). How policy measures succeeded to promote electric mobility e Worldwide review and outlook. *Journal of Cleaner Production*, 206, 66-75. Retrieved from https://doi.org/10.1016/j.jclepro.2018.09.121
- Rijksoverheid. (2020). Climate Agreement measures per sector. Retrieved 2024-03-12, from https://www.rijksoverheid.nl/onderwerpen/klimaatverandering/klimaatakkoord/ maatregelen-klimaatakkoord-per-sector
- Roberts, C., Geels, F. W., Lockwood, M., Newell, P., Schmitz, H., Turnheim, B., & Jordan, A. (2018). The politics of accelerating low-carbon transitions: towards a new research agenda. *Energy Research & Social Science*, 44, 304-311. Retrieved from https://doi.org/10.1016/j.erss .2018.06.001
- Rogers, E. M. (2003). *Diffusion of Innovations* (5th ed.). New York, United States of America: Free Press.
- Rogge, K. S., & Reichardt, K. (2016). Policy mixes for sustainability transitions: an extended concept and framework for analysis. *Research Policy*, *45*, 1620-1635. Retrieved from https://doi.org/ 10.1016/j.respol.2016.04.004
- Roseland, M. (1996). Economic Instruments for Sustainable Community Development. *Local Environment*, *1*(2), 197-210. Retrieved from https://doi.org/10.1080/13549839608725492
- Rosenbloom, D., Berton, H., & Meadowcroft, J. (2016). Framing the sun: A discursive approach to understanding multi-dimensional interactions within socio-technical transitions through the case of solar electricity in ontario, canada. *Research Policy*, *45*, 1275-1290. Retrieved 2023-10-11, from http://dx.doi.org/10.1016/j.respol.2016.03.012
- RVO. (2011). Eurostars: subsidy for international market-oriented R&D. Rijksdienst voor Ondernemend Nederland. Retrieved 2024-03-20, from https://www.rvo.nl/subsidies-financiering/ eurostars
- RVO. (2014). Connecting Europe Facility (CEF) Transport. Rijksdienst voor Ondernemend Nederland. Retrieved 2024-03-20, from https://www.rvo.nl/subsidies-financiering/cef-transport

- RVO. (2015). WBSO: tax scheme for research and development. Rijksdienst voor Ondernemend Nederland. Retrieved 2024-03-20, from https://www.rvo.nl/subsidies-financiering/wbso
- RVO. (2017a). Financial support for electric driving. Rijksdienst voor Ondernemend Nederland. Retrieved 2024-03-18, from https://www.rvo.nl/onderwerpen/elektrisch-vervoer/ financiele-ondersteuning
- RVO. (2017b). *Interreg 2021-2027*. Rijksdienst voor Ondernemend Nederland. Retrieved 2024-03-20, from https://www.rvo.nl/subsidies-financiering/interreg
- RVO. (2017c). *LIFE: European nature, environment and climate projects.* Rijksdienst voor Ondernemend Nederland. Retrieved 2024-03-20, from https://www.rvo.nl/subsidies-financiering/ life
- RVO. (2018). The Formula E-Team and the rise of e-mobility in the Netherlands (Tech. Rep.). Retrieved from https://english.rvo.nl/sites/default/files/2023-07/The_Formula_E_Team _and_the_rise_of_e_mobility_in_the_Netherlands.PDF
- RVO. (2022a). Demonstration of climate technologies and innovations in transport (DKTI-transport). Rijksdienst voor Ondernemend Nederland. Retrieved 2024-03-20, from https://www.rvo.nl/ subsidies-financiering/dkti-transport
- RVO. (2022b). Subsidy Scheme for Electric Passenger Cars for Private Individuals (SEPP). Rijksdienst voor Ondernemend Nederland. Retrieved 2024-03-18, from https://www.rvo.nl/subsidies -financiering/sepp
- RVO. (2023a). *Demonstration Energy and Climate Innovation (DEI+)*. Rijksdienst voor Ondernemend Nederland. Retrieved 2024-03-20, from https://www.rvo.nl/subsidies-financiering/dei
- RVO. (2023b). MIT: SME innovation stimulation Region and Top Sectors. Rijksdienst voor Ondernemend Nederland. Retrieved 2024-03-20, from https://www.rvo.nl/subsidies-financiering/ mit
- ScienceDirect. (2024). Search for peer-reviewed journal articles and book chapters (including open access content). Author. Retrieved 2023-09-15, from https://www.sciencedirect.com/
- SEA, O. E. (2009). Appendix 5 Regulatory Controls (Tech. Rep.). Retrieved from https://assets .publishing.service.gov.uk/media/5a7a388ced915d1a6421bea2/0ES_A5_Controls.pdf
- Shahbaz, M., Siddiqui, A., Siddiqui, M., Jiao, Z., & Kautish, P. (2023). Exploring the growth of sustainable energy technologies: A review. Sustainable Energy Technologies and Assessments, 57, 103157. Retrieved 2023-10-31, from https://doi.org/10.1016/j.seta.2023.103157
- Sharpe, S., & Lenton, T. M. (2021). Upward-scaling tipping cascades to meet climate goals: plausible grounds for hope. *Climate Policy*, *21*, 421-433. Retrieved from https://doi.org/10.1080/ 14693062.2020.1870097
- Siemens. (2023). *Electric times.* Siemens. Retrieved 2023-10-26, from https://www.siemens.com/global/en/company/about/history/specials/siemens-energy-history.html
- Sintrão, C. (2005). Public Consultation Report: Environmental Impact Assessment (Tech. Rep.). Retrieved from https://siaia.apambiente.pt/AIADOC/AIA1430/relatoriocp1430201532617318 .pdf
- Smil, V. (2016). Examining energy transitions: A dozen insights based on performance. Energy Research & Social Science, 22, 194-197. Retrieved from http://dx.doi.org/10.1016/j.erss .2016.08.017
- SMMT. (2018). 2018 UK Automotive Sustainability Report (Tech. Rep.). London, UK. Retrieved from https://www.smmt.co.uk/industry-topics/sustainability/report-archive/
- SMMT. (2019). 2019 UK Automotive Sustainability Report (Tech. Rep.). London, UK. Retrieved from https://www.smmt.co.uk/industry-topics/sustainability/report-archive/
- SMMT. (2023). SMMT History. Retrieved 2024-02-21, from https://www.smmt.co.uk/about/

history/

- Sottish Renewables. (2024). About. Retrieved 2024-03-08, from https://www.scottishrenewables .com/about
- Sovacool, B. K. (2016). How long will it take? Conceptualizing the temporal dynamics of energy transitions. *Energy Research & Social Science*, *13*, 202-215. Retrieved from https://doi.org/10.1016/j.erss.2015.12.020
- Stadelmann-Steffen, I., Eder, C., Harring, N., Spilker, G., & Katsanidou, A. (2021). A framework for social tipping in climate change mitigation: What we can learn about social tipping dynamics from the chlorofluorocarbons phase-out. *Energy Research & Social Science*, 82, 102307. Retrieved from https://doi.org/10.1016/j.erss.2021.102307
- Statista. (2024). Cumulative installed capacity of offshore wind in the United Kingdom (UK) from 2009 to 2022. Retrieved 2024-01-08, from https://www.statista.com/statistics/792374/ cumulative-offshore-wind-capacity-united-kingdom/
- Steffen, W., Rockstrom, J., Richardson, K., Lenton, T. M., Folke, C., Liverman, D., ... Schellnhuber, H. J. (2018). Trajectories of the Earth System in the Anthropocene. *PNAS*, *115*(33), 8252-8259.
 Retrieved from https://doi.org/10.1073/pnas.1810141115
- Steinbach, S. A., & Bunk, N. (2024). The future European hydrogen market: Market design and policy recommendations to support market development and commodity trading. *International Journal* of Hydrogen Energy, 70, 29-38. Retrieved from https://doi.org/10.1016/j.ijhydene.2024 .05.107
- Sterman, J. (2000). *Business dynamics: Systems thinking and modelling for a complex world* (1. ed.). Jeffrey J. Shelstad.
- Sultana, R., Dwivedi, A., & Moktadir, M. A. (2023). Investigating the role of consumers, producers, and policymakers: A case of leather supply chain towards sustainable chemistry. *Current Opinion in Green and Sustainable Chemistry*, 39, 100724. Retrieved 2023-10-31, from https://doi.org/ 10.1016/j.cogsc.2022.100724
- T&E. (2019). Annual report 2019: Cars. Transport & Environment. Retrieved 2024-03-01, from https://www.transportenvironment.org/annual-report-2019/campaigns/cars.html
- Tesla. (2023). About us. Tesla. Retrieved 2023-10-26, from https://www.tesla.com/about
- Thaler, R. H., & Sunstein, C. R. (2008). *Nudge: Improving Decisions About Health, Wealth, and Happiness* (1st ed.). United States of America: Yale University Press.
- The Crown Estate. (2023). About Us: Lasting and shared prosperity for the nation. Retrieved 2024-05-07, from https://www.thecrownestate.co.uk/about-us
- The Faraday Institution. (2024). Faraday Battery Challenge. Retrieved 2024-02-23, from https://www.faraday.ac.uk/the-faraday-battery-challenge/
- The World Bank. (2023). Climate Policies with Real-World Results. Author. Retrieved 2023-11-15, from https://www.worldbank.org/en/news/feature/2023/09/19/climate-policies -with-real-world-results
- Toke, D. (2011). The UK offshore wind power programme: A sea-change in UK energy policy? *Energy Policy*, 39, 526-534. Retrieved 2024-01-19, from https://doi.org/10.1016/j.enpol.2010.08.043
- Transport & Environment. (2016). *Electric Vehicles in Europe* (Tech. Rep.). Retrieved from https://www.transportenvironment.org/wp-content/uploads/2021/07/TE%20EV%

20Report%202016%20FINAL.pdf

- Travassos, P., Costa, H. M., Saraiva, T., Tomé, R., Armelin, M., Ramírez, F. I., & Neves, J. (2005). Wind energy and the conservation of birdlife in Portugal (Tech. Rep.). Lisbon. Retrieved from https://escoladaenergia.abaae.pt/investiga/docs/pdf/ EnergiaEolica_ConservacaoAvifaunaPortugal.pdf
- TVI Notícias. (2006). Banks finance €806 million for a Portuguese wind project. Retrieved 2024-04-11, from https://tvi.iol.pt/noticias/portugal/europa/bancos-financiam-806-milhoes -para-projecto-eolico-portugues
- Tweede Kamer der Staten-Generaal. (2016). *Parliamentary document 31305, nr. 216.* Overheid.nl. Retrieved 2024-03-25, from https://zoek.officielebekendmakingen.nl/kst-31305-216.html
- UK Government. (2018). *The Road to Zero* (Tech. Rep.). London, UK. Retrieved from https://assets .publishing.service.gov.uk/media/5b968e3ee5274a13859deed2/road-to-zero.pdf
- UK Infrastructure Bank. (2022). About us. Retrieved 2024-05-06, from https://www.ukib.org.uk/ about-us
- UK Research and Innovation. (2023). *Faraday battery challenge*. Retrieved 2024-02-23, from https://www.ukri.org/what-we-do/browse-our-areas-of-investment-and-support/ faraday-battery-challenge/
- UK Research and Innovation. (2024a). About uk research and innovation. Retrieved 2024-03-04, from https://www.ukri.org/who-we-are/about-uk-research-and-innovation/
- UK Research and Innovation. (2024b). *Electrical motors and drives and electromagnetics.* Retrieved 2024-03-04, from https://www.ukri.org/what-we-do/browse-our-areas-of -investment-and-support/electrical-motors-and-drives-and-electromagnetics/
- UK Research and Innovation. (2024c). Search results for 'wind'. Author. Retrieved 2024-01-12, from https://www.ukri.org/?s=wind
- UNFCCC. (2022). *The Paris Agreement*. Retrieved 2024-03-19, from https://unfccc.int/process -and-meetings/the-paris-agreement
- University of Edinburgh. (2019). *Wind Energy.* Retrieved 2024-05-07, from https://www.eng.ed.ac .uk/research/themes/wind-energy
- University of Sheffield. (2019). Sheffield-Siemens Wind Power Research Centre (S2WP) opened. Retrieved 2024-05-07, from https://www.sheffield.ac.uk/eee/news/sheffield-siemens-wind -power-research-centre-s2wp-opened
- University of Strathclyde. (2022). Scottish renewable energy collaboration between industry and the University extended into tenth year. Retrieved 2024-05-07, from https://shorturl.at/mXtix
- University of Twente. (2018). Lithium Werks and UT Join Forces in Energy Transition Research. Retrieved 2024-05-08, from https://www.utwente.nl/en/news/2018/9/75551/lithium-werks -and-ut-join-forces-in-energy-transition-research
- van Calster, G. (2014). Chapter 37 regulatory instruments: Sustainable materials management, recycling, and the law. In E. Worrell & M. A. Reuter (Eds.), *Handbook of recycling* (p. 527-535). Boston: Elsevier. Retrieved from https://doi.org/10.1016/B978-0-12-396459-5.00037-4
- Van der Valk Hotels. (2024). *Tesla Charging Points*. Retrieved 2024-05-08, from https://www .valkbusiness.com/nl/tesla-laadpunten
- van der Kam, M. J., Meelen, A. A. H., van Sark, W. G. J. H. M., & Alkemade, F. (2018). Diffusion of solar photovoltaic systems and electric vehicles among Dutch consumers: Implications for the energy transition. *Energy Research & Social Science*, 46, 68-85. Retrieved from https://doi.org/ 10.1016/j.erss.2018.06.003
- Vasseur, V., Kamp, L. M., & Negro, S. O. (2013). A comparative analysis of photovoltaic technological innovation systems including international dimensions: the cases of japan and the netherlands.

Journal of Cleaner Production, 48, 200-210. Retrieved 2023-10-11, from http://dx.doi.org/ 10.1016/j.jclepro.2013.01.017

- VER. (2023). About VER. Retrieved 2024-03-12, from https://www.evrijders.nl/over-ver
- Vivalt, E., & Coville, A. (2023). How do policymakers update their beliefs? *Journal of Development Economics*, *165*, 103121. Retrieved 2023-10-31, from https://doi.org/10.1016/j.jdeveco .2023.103121
- VNA. (2024). *About us.* Vereniging van Nederlandse Autoleasemaatschappijen. Retrieved 2024-03-21, from https://www.vna-lease.nl/over-vna/
- VZR. (2022). Lobby Car Letter 2. Vereniging Zakelijke Rijders. Retrieved 2024-03-21, from https:// www.vzr.nl/lobby/lobby-autobrief-2
- VZR. (2024). Core values. Vereniging Zakelijke Rijders. Retrieved 2024-03-21, from https://www .vzr.nl/de_vereniging/kernwaarden
- Wills, T. (2020). Briefing document The UK's transition to electric vehicles (Tech. Rep.). London, UK. Retrieved from https://www.theccc.org.uk/wp-content/uploads/2020/12/The-UKs -transition-to-electric-vehicles.pdf
- WindEurope. (2016a). About us. Retrieved 2024-04-09, from https://windeurope.org/about-us/
- WindEurope. (2016b). Meet our members. Retrieved 2024-04-09, from https://windeurope.org/ membership/meet-our-members/?sector=&country=PT&search=#search-results
- Winkelmann, R., Donges, J. F., Smith, E. K., Milkoreit, M., Eder, C., Heitzig, J., ... Lenton, T. M. (2022). Social tipping processes towards climate action: A conceptual framework. *Ecological Economics*, 192, 107242. Retrieved from https://doi.org/10.1016/j.ecolecon.2021.107242
- WMG. (2024). *Research Areas.* Warwick Manufacturing Group. Retrieved 2024-05-07, from https://warwick.ac.uk/fac/sci/wmg/research/research-areas/
- World Wildlife Fund. (2023). Energy transition. Author. Retrieved 2023-10-27, from https://wwf.panda.org/discover/our_focus/climate_and_energy_practice/what_we _do/changing_energy_use/
- Worrell, E. (2014). Chapter 36 information instruments. In E. Worrell & M. A. Reuter (Eds.), *Handbook of recycling* (p. 521-525). Boston: Elsevier. Retrieved from https://doi.org/10.1016/B978-0 -12-396459-5.00036-2
- Xu, N., Kasimov, I., & Wang, Y. (2022). Unlocking private investment as a new determinant of green finance for renewable development in china. *Renewable Energy*, *198*, 1121-1130. Retrieved 2023-11-01, from https://doi.org/10.1016/j.renene.2022.07.037
- Ydersbond, I. (2012). *Multi-level lobbying in the eu: The case of the renewables directive and the german energy industry* (Unpublished doctoral dissertation). Fridtjof Nansen Institute, Lysaker, Norway.
- Yellowbrick. (2018). Free parking or paid parking at a charging point? Retrieved 2024-03-19, from https://yellowbrick.nl/blog/gratis-parkeren-of-parkeergeld-betalen-bij -een-laadpunt/
- Zapmap. (2023). *Electric Car Tax.* Retrieved 2024-02-27, from https://www.zap-map.com/ev -guides/electric-car-tax
- Zhang, S., & He, Y. (2013). Analysis on the development and policy of solar PV power in China. Renewable and Sustainable Energy Reviews, 21, 393-401. Retrieved from https://doi.org/ 10.1016/j.rser.2013.01.002
- Ørsted. (2019). Making green energy affordable (Tech. Rep.). Retrieved from https://orsted.com/ en/insights/white-papers/making-green-energy-affordable/1991-to-2001-the-first -offshore-wind-farms

A | Data

Year	Cumulative Installed Capacity	(MW)
Tear	Offshore wind energy in the UK	Onshore wind energy in Portugal
	(BVG Associates, 2021; RenewableUK, 2024; Statista, 2024)	(Costa et al., 2021; IRENA, 2013b)
2000	4.0	99.0
2001	-	129.0
2002	-	194.5
2003	64.0	297.0
2004	-	537.5
2005	-	1,041.0
2006	-	1,594.0
2007	404.0	2,269.0
2008	-	3,003.0
2009	819.5	3,603.0
2010	1,338.0	4,012.5
2011	1,838.0	4,378.5
2012	2,834.0	4,521.0
2013	3,671.5	4,731.0
2014	4,501.0	4,953.0
2015	5,096.0	5,034.0
2016	5,293.0	5,313.0
2017	6,381.0	5,313.0
2018	8,038.5	5,379.0
2019	9,795.0	5,459.0
2020	10,397.5	5,502.0
2021	10,857.5	5,828.0
2022	13,928.0	-
2023	14,735.0	-

Table A.1: Cumulative wind installed capacity in MW in the UK and in Portugal, used to create Figure 6.1 and Figure 6.3

 Table A.2: Total stock of BEVs, PHEVs, and EVs, and total number of public charging devices in the UK and in the Netherlands. This data was used to create Figure 6.5, Figure 6.8, Figure 6.9, and Figure 6.12

			UK				Netherlar	ıds
Year	BEVs	PHEVs	EVs	Public charging devices	BEVs	PHEVs	EVs	Public charging devices
		(IEA, 2022)		(DfT, 2023)		(IEA, 2022)		(Statista, 2023)
2010	1,500	21	1,521	-	270	11	281	-
2011	2,600	28	2,628	-	1,100	26	1,126	-
2012	4,100	990	5,090	-	1,900	4,400	6,300	-
2013	6,200	2,000	8,200	-	4,200	24,000	28,200	-
2014	12,000	9,700	21,700	-	6,800	37,000	43,800	5,675
2015	21,000	27,000	48,000	2,283	9,400	78,000	87,400	7,860
2016	30,000	55,000	85,000	3,672	13,000	99,000	112,000	12,380
2017	42,000	87,000	129,000	5,111	21,000	98,000	119,000	16,043
2018	56,000	130,000	186,000	7,211	45,000	98,000	143,000	21,344
2019	91,000	160,000	251,000	10,309	110,000	96,000	206,000	29,025
2020	190,000	220,000	410,000	16,505	180,000	110,000	290,000	41,995
2021	380,000	330,000	710,000	20,775	250,000	140,000	390,000	54,000
2022	550,000	400,000	950,000	28,375	340,000	190,000	530,000	73,968

B | Summary of Actor Roles

The actor roles included in the tables of this Appendix were explained in Section 7.1.3. Table B.1 provides a summary of all interactions (sub-divided per actor role) present in case studies' feedback loop models, showing whether the interaction was included or not in each model. Lastly, Table B.2 shows which of the interactions were found in the refined version of the model of Geels and Ayoub (2023) (Figure 4.3), the version of the expanded feedback loop model based on literature (Figure 4.4), and the two final feedback loop models (Figure 7.1 and Figure 7.2).

The actor role of promoting education was not included in the final feedback loop diagrams (presented in Section 7.1.1), with the reasoning provided in Section 7.1.3. However, it did appear in the first version of the expanded feedback loop model, which was based on general literature alone. Therefore, it was included in the following tables.

Interaction label	Wind	energy	Electric vehicles	
Interaction label	UK	Portugal	UK	The Netherland
	(Figure 6.2)	(Figure 6.4)	(Figure 6.7)	(Figure 6.11)
Technology development				
Investment in R&D and production	1	1	1	1
Value	1	1	1	1
Research and test	1	1	1	1
Technology adoption				
Purchases	1	1	1	1
Revenue and profit	1	1	1	1
Further adoption	1	1	1	1
Expanded technology's visibility	1	1	1	1
Raising awareness				
Public awareness campaigns	1	1	1	1
Foster collaboration	×	×	×	×
Legitimising technology				
Strong policy plans and accompanying narratives	1	1	1	1
Solution to societal problems	1	1	1	1
Legitimacy	1	1	1	1
Public opinion	1	1	1	1
Providing financing incentives/options				
Loans	1	1	1	1
Capital grants	1	1	1	1

Table B.1: Summary of all interactions present in this report's feedback loop models, showing on which model(s) they were included. The interactions are sub-divided according to the overarching actor roles explained in Section 7.1.3

Continued on next page

Table B.1 – continued from previous page					
Interaction label	Wind	energy	Electric vehicles		
	UK	Portugal	UK	The Netherlands	
	(Figure 6.2)	(Figure 6.4)	(Figure 6.7)	(Figure 6.11)	
Support new ventures	1	1	1	1	
R&D subsidies	1	1	1	1	
Purchase subsidies	×	1	1	1	
Tax rebates	×	×	1	\checkmark	
Feed-in tariffs	1	1	×	×	
Funding			1	×	
Long-term certainty		1	X	×	
Financing solutions			X	×	
Public financing departments or businesses		1	×	×	
Establishing market-based mechanisms			1		
Tenders	1	1	×	×	
Investing in infrastructure					
Investment in infrastructure	1	1	1	✓	
Establishing legislation					
Regulations and standards	1	1	1	1	
(Binding) targets	1	1	1	1	
Policies, goals and coalitions adjustments	1	1	1	1	
Promoting education					
Training	×	×	×	×	
Retraining programs	×	×	1	×	
Promote education	×	×	1	1	
Fostering collaboration					
Foster collaboration	1	1	1	1	
Transferring knowledge					
Knowledge transfer	1	1	1	1	
Expert professionals	1	1	1	1	
Giving & receiving feedback					
Consultation	1	1	1	1	
User feedback	×	×	1	1	
Impact evaluations	×	×	×	×	
Convey and interpret data	×	×	×	×	
Lobbying					
Lobbying	1	1	1	1	
Litigation and electioneering	×	×	×	×	
Foster collaboration	1	1	1	1	
Public opinion	1	1	1	1	

Table B.1 – continued from previous page

 Table B.2: Summary of all interactions present in this report's feedback loop models, showing on which model(s) they were included. The interactions are sub-divided according to the overarching actor roles as defined in Section 7.1.3

(Figure 4.3)(Figure 7.1)(Figure 7.1)<	Interaction label	Geels & Ayoub	Literature	Wind energy	Electric vehicles
Investment in R&D and production 2 2 2 2 2 2 2 2 2 2		(Figure 4.3)	(Figure 4.4)	(Figure 7.1)	(Figure 7.2)
Value✓✓✓✓Research and testX✓✓✓Research and testX✓✓✓Technology adoption✓✓✓✓Purchases✓✓✓✓Revenue and profit✓✓✓✓Further adoption✓✓✓✓Expanded technology's visibility✓✓✓✓Raising awareness✓✓✓Public awareness campaignsX✓✓✓Foster collaborationX✓✓✓Storog policy plans and accompanying narratives✓✓✓Solution to societal problems✓✓✓✓Public opinion✓✓✓✓✓Providing financing incentives/options✓✓✓✓Loans✓✓✓✓✓✓Purchases subsidies✓✓✓✓✓Purchases subsidies✓✓✓✓✓ITar rebatesX✓✓✓✓Long-term certaintyX✓✓✓✓Publicing solutionsX✓✓✓✓Establishing market-based mechanisms✓✓✓✓TendersX✓✓✓✓✓Itar rebatesX✓✓✓✓Itar rebatesX✓✓✓✓	Technology development				
Research and test X ✓ ✓ ✓ Technology adoption Purchasses ✓ ✓ ✓ ✓ Purchasses ✓ ✓ ✓ ✓ ✓ ✓ Revenue and profit ✓ ✓ ✓ ✓ ✓ ✓ ✓ Further adoption ✓<	Investment in R&D and production	1	1	1	1
Technology adoption Purchases / / / Purchases / / / Revenue and profit / / / Further adoption / / / Expanded technology's visibility / / / Raising awareness // / / Public awareness campaigns X / / Foster collaboration X / X Strong policy plans and accompanying narratives / / / Solution to societal problems / / / Legitimacy / / / / Public opinion / / / / Verbilic opinion / / / / Support new ventures X / / / Support new ventures X / / / Tax rebates / / / / / Purchase subsidies / / / / / Support new ventures X / / / / R&D subsidies / / / / / Purchase subsi	Value	1	1	1	\checkmark
Purchases////Revenue and profit////Further adoption////Expanded technology's visibility////Raising awareness////Public awareness campaignsX///Foster collaborationX/X/Egitimising technology////Stong policy plans and accompanying narratives///Solution to societal problems////Legitimacy/////Public opinion/////Providing financing incentives/options////Loans//////Support new venturesX////Support new venturesX////Tax rebatesX/////Feed-in tariffsX/////Long-term certaintyX///X/Establishing market-based mechanismsX//X/EndersX////XInvesting in infrastructure/////Investing in infrastructure/////Establishing legislation// <td< td=""><td>Research and test</td><td>×</td><td>1</td><td>1</td><td>1</td></td<>	Research and test	×	1	1	1
Revenue and profit///Further adoption///Further adoption///Expanded technology's visibility///Raising awareness///Public awareness campaignsX//Foster collaborationX//X/X//Legitimising technology///Stong policy plans and accompanying narratives//Solution to societal problems///Legitimacy////Public opinion////Providing financing incentives/options///Loans////Support new venturesX///X/////Support new venturesX///X/////Purchase subsidies////X/////Inancing solutionsX///Long-term certaintyX///X/////Estabilishing market-based mechanisms///Etablishing in infrastructure////Investing in infrastructure////Estabilishing legislation//	Technology adoption				
Further adoption✓✓✓✓Expanded technology's visibility✓✓✓✓Raising awarenessPublic awareness campaignsX✓✓✓Public awareness campaignsX✓✓✓✓Foster collaborationX✓✓✓✓Legitimising technologyStrong policy plans and accompanying narratives✓✓✓✓✓Solution to societal problems✓✓✓✓✓✓Legitimacy✓✓✓✓✓✓Public opinion✓✓✓✓✓✓Providing financing incentives/options✓✓✓✓✓Loans✓✓✓✓✓✓Support new venturesX✓✓✓✓✓Support new venturesX✓✓✓✓✓R&D subsidies✓✓✓✓✓✓Purchase subsidies✓✓✓✓✓✓Tax rebatesX✓✓✓✓✓✓Financing solutionsX✓✓✓✓✓Lungterm certaintyX✓✓✓✓✓Estabilshing market-based mechanismsX✓✓✓✓TendersX✓✓✓✓✓✓Longterm certainty×✓✓✓✓	Purchases	1	1	1	1
Expanded technology's visibilityImage: science of the second	Revenue and profit	1	1	1	\checkmark
Raising awareness Public awareness campaigns X ✓ ✓ Foster collaboration X ✓ X ✓ Legitimising technology Strong policy plans and accompanying narratives ✓	Further adoption	1	1	1	\checkmark
Public awareness campaigns X / Legitimacy / <t< td=""><td>Expanded technology's visibility</td><td>1</td><td>1</td><td>1</td><td>\checkmark</td></t<>	Expanded technology's visibility	1	1	1	\checkmark
Foster collaborationX✓X✓Legitimising technologyStrong policy plans and accompanying narratives✓✓✓✓Solution to societal problems✓✓✓✓Legitimacy✓✓✓✓✓Public opinion✓✓✓✓✓Providing financing incentives/options✓✓✓✓Loans✓✓✓✓✓Capital grants✓✓✓✓✓Support new venturesX✓✓✓✓Purchase subsidies✓✓✓✓✓Purchase subsidies✓✓✓✓✓Feed-in tariffsX✓✓✓✓Long-term certaintyX✓✓✓✓Public financing departments orX✓✓✓Public financing departments orX✓✓✓Public financing in infrastructure✓✓✓✓Investing in infrastructure✓✓✓✓Establishing legislation✓✓✓✓✓	Raising awareness				
Legitimising technology Strong policy plans and accompanying narratives Nying narratives Solution to societal problems Legitimacy Public opinion V Providing financing incentives/options Loans V V Capital grants V V Support new ventures X V X V Purchase subsidies V V V Y X Punchase subsidies V V V Y X Purchase subsidies V V V Y X Purchase subsidies V Y X Purchase subsidies X Y X Purchase subsidies X Y X Solutiong X X Y Purchase subsidies X Y X Purchase subsidies X <	Public awareness campaigns	×	1	1	1
Strong policy plans and accompanying narratives ✓ ✓ ✓ ✓ Solution to societal problems ✓ ✓ ✓ ✓ Legitimacy ✓ ✓ ✓ ✓ ✓ Public opinion ✓ ✓ ✓ ✓ ✓ Providing financing incentives/options ✓ ✓ ✓ ✓ Loans ✓ ✓ ✓ ✓ ✓ Capital grants ✓ ✓ ✓ ✓ ✓ Support new ventures X ✓ ✓ ✓ R&D subsidies ✓ ✓ ✓ ✓ Purchase subsidies ✓ ✓ ✓ ✓ Tax rebates X ✓ ✓ ✓ Feed-in tariffs X ✓ ✓ ✓ Funding X ✓ ✓ ✓ Long-term certainty X ✓ ✓ X Public financing departments or X ✓ ✓ businesses X ✓ ✓ ✓ Establishing market-based mechanisms X ✓ ✓ Tenders X ✓ ✓ ✓ Investment in infrastructure ✓ ✓ ✓ ✓ Investment in infrastructure ✓ ✓ ✓ ✓	Foster collaboration	×	✓	×	✓
nying narratives Solution to societal problems ✓ ✓ ✓ ✓ Legitimacy ✓ ✓ ✓ ✓ Public opinion ✓ ✓ ✓ ✓ Providing financing incentives/options ✓ ✓ ✓ ✓ Loans ✓ ✓ ✓ ✓ ✓ Capital grants ✓ ✓ ✓ ✓ ✓ Support new ventures X ✓ ✓ ✓ ✓ Purchase subsidies ✓ ✓ ✓ ✓ ✓ Purchase subsidies ✓ ✓ ✓ ✓ ✓ ✓ Feed-in tariffs X ✓ ✓ ✓ ✓ ✓ ✓ Funding X ✓ <	Legitimising technology				
Solution to societal problems✓✓✓✓Legitimacy✓✓✓✓✓Public opinion✓✓✓✓✓Providing financing incentives/options✓✓✓✓Loans✓✓✓✓✓Capital grants✓✓✓✓✓Support new venturesX✓✓✓✓R&D subsidies✓✓✓✓✓Purchase subsidies✓✓✓✓✓Purchase subsidies✓✓✓✓✓Feed-in tariffsX✓✓✓✓FundingX✓✓✓✓✓Long-term certaintyX✓✓✓XPublic financing departments orX✓✓✓Xbusinesses✓✓✓✓✓Establishing market-based mechanisms✓✓✓✓Investing in infrastructure✓✓✓✓✓✓Investing in infrastructure✓✓✓✓✓✓Establishing legislation✓✓✓✓✓		 ✓ 	1	1	1
Legitimacy✓✓✓✓✓Public opinion✓✓✓✓✓Providing financing incentives/optionsLoans✓✓✓✓Capital grants✓✓✓✓Capital grants✓✓✓✓Support new venturesX✓✓✓R&D subsidies✓✓✓✓Purchase subsidies✓✓✓✓Purchase subsidies✓✓✓✓Feed-in tariffsX✓✓✓Feed-in tariffsX✓✓✓FundingX✓✓✓Long-term certaintyX✓✓✓Establishing market-based mechanismsX✓✓XInvesting in infrastructure✓✓✓✓Investing in infrastructure✓✓✓✓Establishing legislation✓✓✓✓	nying narratives				
Public opinion✓✓✓✓✓Providing financing incentives/optionsLoans✓✓✓✓Capital grants✓✓✓✓Capital grants✓✓✓✓Support new venturesX✓✓✓R&D subsidies✓✓✓✓Purchase subsidies✓✓✓✓Purchase subsidies✓✓✓✓Feed-in tariffsX✓✓✓Feed-in tariffsX✓✓✓FundingX✓✓✓Long-term certaintyX✓✓✓Financing solutionsX✓✓✓Establishing market-based mechanismsX✓✓✓Investing in infrastructure✓✓✓✓Investing in infrastructure✓✓✓✓Establishing legislation✓✓✓✓		1	1	1	1
Providing financing incentives/options Loans ✓ ✓ ✓ ✓ Capital grants ✓ ✓ ✓ ✓ Support new ventures X ✓ ✓ ✓ ✓ R&D subsidies ✓ ✓ ✓ ✓ ✓ ✓ Purchase subsidies ✓ ✓ ✓ ✓ ✓ ✓ ✓ Purchase subsidies ✓ <	Legitimacy	1	1	1	1
Loans✓✓✓✓Capital grants✓✓✓✓✓Support new venturesX✓✓✓✓R&D subsidies✓✓✓✓✓Purchase subsidies✓✓✓✓✓Purchase subsidies✓✓✓✓✓Tax rebatesX✓✓✓✓Feed-in tariffsX✓✓✓✓FundingX✓✓✓✓Long-term certaintyX✓✓✓✓Financing solutionsX✓✓✓✓Public financing departments or businessesX✓✓✓Establishing market-based mechanismsX✓✓✓Investing in infrastructure✓✓✓✓✓Investing in infrastructure✓✓✓✓✓Establishing legislation✓✓✓✓	Public opinion	1	1	1	1
Capital grantsImage: scalar scala	Providing financing incentives/option	าร			
Support new venturesX✓XXR&D subsidies✓✓✓✓Purchase subsidies✓✓✓✓Purchase subsidies✓✓✓✓Tax rebatesX✓✓✓Feed-in tariffsX✓✓✓FundingX✓✓✓Long-term certaintyX✓✓✓Financing solutionsX✓✓✓Public financing departments or businessesX✓✓Establishing market-based mechanisms✓✓✓Investing in infrastructure✓✓✓Investment in infrastructure✓✓✓Establishing legislation✓✓✓	Loans	1	1	1	\checkmark
R&D subsidies✓✓✓✓Purchase subsidies✓✓✓✓Tax rebatesX✓X✓Feed-in tariffsX✓✓XFundingX✓✓✓Long-term certaintyX✓✓Financing solutionsX✓✓Public financing departments or businessesX✓✓Establishing market-based mechanisms✓✓✓Investing in infrastructure✓✓✓Investment in infrastructure✓✓✓Establishing legislation✓✓✓	Capital grants	1	1	1	\checkmark
Purchase subsidiesImage: subsidiesIma	Support new ventures	×	1	×	×
Tax rebatesXIXIFeed-in tariffsXIIXFundingXIIILong-term certaintyXIIILong-term certaintyXIIIFinancing solutionsXIIIPublic financing departments or businessesXIIIEstablishing market-based mechanismsIIIIInvesting in infrastructureIIIIEstablishing legislationIIII	R&D subsidies	1	1	1	\checkmark
Feed-in tariffsX✓✓XFundingX✓✓✓Long-term certaintyX✓✓✓Financing solutionsX✓✓✓Public financing departments or businessesX✓✓Establishing market-based mechanisms✓✓✓TendersX✓✓✓Investing in infrastructure✓✓✓Investment in infrastructure✓✓✓Establishing legislation✓✓✓	Purchase subsidies	1	1	1	1
Funding X ✓ ✓ ✓ Long-term certainty X ✓ ✓ X Financing solutions X ✓ ✓ X Public financing departments or businesses X ✓ ✓ X Establishing market-based mechanisms X ✓ ✓ X Tenders X ✓ ✓ X Investing in infrastructure ✓ ✓ ✓ ✓ Establishing legislation Investing legislation ✓ ✓ ✓	Tax rebates	×	1	×	1
Long-term certainty X ✓ X Financing solutions X ✓ X Public financing departments or businesses X X ✓ X Establishing market-based mechanisms X ✓ X X Tenders X ✓ ✓ X Investing in infrastructure ✓ ✓ ✓ ✓ Establishing legislation ✓ ✓ ✓ ✓	Feed-in tariffs	×	1	1	×
Financing solutions X ✓ X Public financing departments or businesses X X X X Establishing market-based mechanisms X ✓ X X Tenders X ✓ ✓ X Investing in infrastructure ✓ ✓ ✓ ✓ Establishing legislation ✓ ✓ ✓ ✓	Funding	×	1	1	\checkmark
Public financing departments or businesses X X ✓ X Establishing market-based mechanisms Establishing market-based mechanisms ✓ ✓ ✓ X Tenders X ✓ ✓ ✓ X ✓ ✓ Investing in infrastructure ✓ ✓ ✓ ✓ ✓ ✓ Investment in infrastructure ✓ ✓ ✓ ✓ ✓ ✓ Establishing legislation ✓ ✓ ✓ ✓ ✓ ✓	Long-term certainty	×	1	1	×
businesses Establishing market-based mechanisms Tenders X Investing in infrastructure Investment in infrastructure Investment in infrastructure Istablishing legislation	Financing solutions	×	1	1	×
X ✓ X X Investing in infrastructure Investment in infrastructure ✓ ✓ Investment in infrastructure ✓ ✓ ✓ Establishing legislation ✓ ✓ ✓		×	×	1	×
Investing in infrastructure Investment in infrastructure Establishing legislation	Establishing market-based mechani	sms			
Investment in infrastructure	Tenders	×	1	1	×
Establishing legislation	Investing in infrastructure				
	Investment in infrastructure	1	1	1	1
	Establishing legislation				
Regulations and standards VVVV	Regulations and standards	1	1	1	1

Interaction label	Geels & Ayoub	General literature	Wind energy	Electric vehicles
	(Figure 4.3)	(Figure 4.4)	(Figure 7.1)	(Figure 7.2)
(Binding) targets	×	×	1	1
Policies, goals and coalitions adjustments	1	1	1	1
Promoting education ¹				
Training	×	1	×	×
Retraining programs	×	×	×	×
Promote education	×	1	×	×
Fostering collaboration				
Foster collaboration	×	1	1	1
Transferring knowledge				
Knowledge transfer	×	1	1	1
Expert professionals	×	1	1	\checkmark
Giving & receiving feedback				
Consultation	×	×	1	1
User feedback	×	×	×	1
Impact evaluations	×	1	×	×
Convey and interpret data	×	1	×	×
Lobbying				
Lobbying	1	1	1	✓
Litigation and electioneering	×	✓	×	×
Foster collaboration	×	✓	1	\checkmark
Public opinion	1	1	1	1

Table B.2 – continued from previous page

C | Summary of Policies

Section C.1 complements the explanation provided in Section 4.3.1 about different types of policy instruments. Secondly, Section C.2 provides a summary of the policies found in each case study.

C.1. Policy Categorisation

In the framework of Geels and Ayoub (2023) the following policy instruments are included: loans, capital grants, R&D subsidies, infrastructure investment, regulations, standards, and purchase subsidies (see Figure 4.2). However, due to the plenitude of policy instruments, it is unlikely that these were the only ones at play. Perhaps additional policies were included in terms such as "stronger policy support" and "climate-related policy pressures", which do not say much. This thesis aimed to understand which policies were employed in the four case studies, as well as which ones were the most important. Specific policies differ when looking at different technologies and/or different countries. Thus, to compare the policies of each case study, policy instruments were categorised according to their objective.

Policies can be divided into three main groups: economic, regulatory, and information instruments (Dubois & Eyckmans, 2014; Rogge & Reichardt, 2016; van Calster, 2014; Worrell, 2014). These three policy types are explained below. Each instrument was further sub-divided into three categories, allowing the comparison of the policies employed in the different case studies according to their objective (e.g., feed-in-tariffs and contracts-for-difference are two instruments that have a similar goal).

Economic Instruments

Economic instruments are policy tools designed to compensate for externalities and can be employed to internalise environmental costs, applying the polluter-pays principle. Economic instruments enhance the capacity of governments to address environmental and development issues cost-effectively. These instruments aim to influence behaviour such that economic activities align with specific policy objectives. By providing financial incentives for environmentally improved behaviour or disincentives for harmful practices, governments can encourage technological innovation and influence consumption and production patterns. This is done through alleviating financial barriers, promoting investments, and making certain products more economically viable (Johansson et al., 2021; Panayotou, 1995; Roseland, 1996).

It was decided to divide economic instruments into three sub-categories: financial incentives, marketbased mechanisms, and R&D support schemes. Examples of economic instruments falling under each category are provided in Table C.1. In particular, *financial incentives* aim to stimulate investment in renewable energy projects and promote the adoption of sustainable practices. Examples include purchase subsidies, feed-in tariffs, tax rebates, and green bonds (Cox, 2016). *Market-based mechanisms* leverage market forces to internalise the costs of environmental externalities and promote emissions reductions. Examples are carbon taxes, tenders, net metering, and emissions trading (Joskow & Schmalensee, 1998). *R&D support* aims to foster innovation and technological advancements through financial support for R&D and demonstration projects (Huergo & Moreno, 2014).

Examples	Source		
Financial incentives			
Investment subsidies or grants	(Gregersen & Johnson, 2010)		
Purchase subsidies	(Besley & Persson, 2023; Jaffe & Stavins, 1994; Rogge & Reichardt, 2016; Zhang & He, 2013)		
Renewable energy financing programs	(Q. Chai & Zhang, 2010)		
Feed-in tariffs	(Gregersen & Johnson, 2010; Rogge & Reichardt, 2016; Zhang & He, 2013)		
Tax credits, rebates and incentives	(Q. Chai & Zhang, 2010; Gregersen & Johnson, 2010; Jaffe & Stavins, 1994; Rogge & Reichardt, 2016)		
Green bonds	(Newell & Simms, 2021; Zhang & He, 2013)		
State equity assistance	(Rogge & Reichardt, 2016)		
Sustainable public procurement	(Gregersen & Johnson, 2010; Rogge & Reichardt, 2016)		
Market-based mechanisms			
Carbon taxes	(Besley & Persson, 2023; Rogge & Reichardt, 2016)		
Emissions trading	(Gregersen & Johnson, 2010; Jaffe & Stavins, 1994; Rogge & Reichardt, 2016)		
Tenders	(Metcalfe & Sasse, 2023)		
Net metering or billing	(Gregersen & Johnson, 2010)		
R&D support			
R&D incentives	(Jaffe & Stavins, 1994; Newell & Simms, 2021; Rogge & Reichardt, 2016; Zhang & He, 2013)		
Financial and technical support for demonstration projects	(Gregersen & Johnson, 2010; Zhang & He, 2013)		

Table C.1: Examples of economic instruments per sub-category, and sources where more information can be found

Regulatory Instruments

Regulatory instruments are policy tools that involve the use of informal and formal orders, decrees, laws, regulations, and standards to achieve policy objectives. These instruments are typically mandated by government authorities and enforce compliance through legal and administrative measures (Astuti, Day, & Emery, 2019; Entsalo, Kalimo, Kautto, & Turunen, 2023). Three sub-categories were created: standards, strategic planning, and market facilitation. Examples of regulatory instruments falling under each category are provided in Table C.2. *Standards* ensure that products, practices, and operations meet specific requirements or guidelines. Examples include mandating products or practices, intellectual property rights, renewable portfolio standards, technology/performance standards, and product labelling (ISO, 2024). *Strategic planning* involves the setting of goals, targets, and long-term roadmaps to guide policy development and decision-making (Gregersen & Johnson, 2010; Newell & Simms, 2021). *Market facilitation* focuses on creating conducive environments for the development, growth, and integration of renewable energy markets and infrastructure. This is done via local market design/development, infrastructure provision, and grid access guarantees (Steinbach & Bunk, 2024).

Table C.2: Examples of regulatory instruments per sub-category, and sources where more information c	an be found

Examples	Source
Standards	
Mandate products or practices	(Gregersen & Johnson, 2010; Jaffe & Stavins, 1994; Rogge & Reichardt, 2016)
Application constraints	(Rogge & Reichardt, 2016)
Mandate environmental reporting	(Jaffe & Stavins, 1994)
Environmental liability law	(Rogge & Reichardt, 2016)
Intellectual property rights	(Gregersen & Johnson, 2010; Rogge & Reichardt, 2016)
	Continued on next page

Examples	Source		
Patent law	(Rogge & Reichardt, 2016)		
Renewable portfolio standard	(Gregersen & Johnson, 2010)		
Technology/Performance standard	(Gregersen & Johnson, 2010; Jaffe & Stavins, 1994; Rogge & Reichardt, 2016)		
Product labelling	(Jaffe & Stavins, 1994)		
Strategic planning			
(Binding) targets	(Q. Chai & Zhang, 2010; Gregersen & Johnson, 2010; Newell & Simms, 2021; Zhang & He, 2013)		
Roadmap	(Gregersen & Johnson, 2010; Newell & Simms, 2021)		
Market facilitation			
Local market design/development	(Newell & Simms, 2021; Rogge & Reichardt, 2016)		
Infrastructure provision (Rogge & Reichardt, 2016; The World Bank, 2023)			
Priority feed-in (Rogge & Reichardt, 2016)			
Grid access guarantee	(Rogge & Reichardt, 2016)		

Information Instruments

Information instruments intend to change individuals' behaviour by making sure they are better informed about the necessity or advantages of certain choices. To do so effectively, information campaigns should be customised to the target group, and fundamental information should be repeated often (Orset, 2021; Worrell, 2014). Information instruments were sub-divided into knowledge transfer, collaboration and networking, and outreach initiatives. Table C.3 shows examples for each. *Knowledge transfer* is designed to facilitate information, expertise, and skills exchange among stakeholders via e.g., professional and entrepreneurship trainings and scientific workshops (Kochenkova et al., 2015). *Collaboration and networking* incentives aim to facilitate partnerships, cooperation, and knowledge sharing among stakeholders, which is accomplished through public-private partnerships, cooperative RD&D programs, thematic meetings, and clusters (Q. Chai & Zhang, 2010; Lehmann, 2006). *Outreach* instruments aim to raise awareness, educate the public, and engage stakeholders. Examples include rating and labelling programs, public information campaigns, and organised public debates (Orset, 2021).

Examples	Source
Knowledge transfer	
Professional training and qualification Entrepreneurship training Scientific workshops	(Newell & Simms, 2021; Rogge & Reichardt, 2016) (Kochenkova et al., 2015; Rogge & Reichardt, 2016) (Kochenkova et al., 2015; Rogge & Reichardt, 2016)
Collaboration & networking	
Public-private partnerships Cooperative RD&D programmes Thematic meetings Clusters	(Gregersen & Johnson, 2010; Lehmann, 2006) (Q. Chai & Zhang, 2010; Newell & Simms, 2021; Rogge & Reichardt, 2016) (Rogge & Reichardt, 2016) (Rogge & Reichardt, 2016)
Outreach	
Rating and labelling programmes Public information campaigns Public debates	(Rogge & Reichardt, 2016) (Jaffe & Stavins, 1994; Orset, 2021; Rogge & Reichardt, 2016) (Rogge & Reichardt, 2016)

Table C.3: Examples of information instruments per sub-category, and sources where more information can be found

Therefore, many policy instruments exist to foster the diffusion of SETs. It is thus unlikely that there were not more relevant policies employed in the case studies analysed by Geels and Ayoub (2023).

C.2. Case Studies Policies

In this section, a summary of the policies found in each case study is provided, in the same order as the case studies appear in Chapter 6. The policies are allocated to the respective sub-category (explained in Section C.1). Furthermore, their implementation year and source are stated.

Wind Energy in the UK (2002 - 2014)

Table C.4 gives an overview of the policies employed between 2002 and 2014 that had the largest impact on the uptake of offshore wind energy in the UK. This summary was constructed based on the information in Section 6.1.

 Table C.4: Policies in place during the enabling phase and part of the accelerating phase (2002 - 2014) of the deployment of offshore wind energy in the UK, as well as the year in which they were introduced and the source where this information can be obtained

Instrument type	Instruments employed	Year	Source
Economic instruments			
Financial incentives	Renewable Obligation Certificates	2002 (adjusted in 2009)	(Geels & Ayoub, 2023)
	Interim Connect and Manage	2009-2011	(Ofgem, 2013)
	Connect and Manage	2011	(Metcalfe & Sasse, 2023; Ofgem, 2013)
	Green Investment Bank	2012	(Green Investment Group, 2024)
	Final Investment Decision Enabling for Renewables (FiDeR)	2013-2014	(BVG Associates, 2021)
	Contracts for Difference	2014	(BVG Associates, 2021)
	Offshore Wind Capital Grant Scheme	2003	(IEA, 2014)
	Environmental Transformation Fund	2009-2011	(Kern et al., 2014)
	Offshore Wind Manufacturing Funding	2011-2015	(Kern et al., 2014)
Market-based mecha- nisms	EUETS	2005	(European Commission, 2021)
	Crown Estate seabed leasing auctions	2003 and 2008	(Kern et al., 2014)
	Tenders for transmission licenses	2009	(Metcalfe & Sasse, 2023)
	Carbon Price Floor	2013	(HM Revenue & Customs, 2018)
R&D support	Offshore Wind Accelerator	2008-2014	(Kern et al., 2014)
Regulatory instruments			
Standards	Renewable Obligation	2002 (amended in 2009)	(Geels & Ayoub, 2023)
Strategic planning	EU targets	2007	(McNally, 2022)
	Climate Change Act	2008	(Metcalfe & Sasse, 2023)
	Renewable Energy Roadmap	2011	(DECC, 2011)
	Offshore Wind Industrial Strategy	2013	(BIS & DECC, 2013)
Market facilitation ¹	Interim Connect and Manage	2009-2011	(Ofgem, 2013)
	Connect and Manage	2011	(Metcalfe & Sasse, 2023; Ofgem, 2013)
Information instruments			
Knowledge transfer	Test and technical services	2003	(BVG Associates, 2021)
Collaboration & network- ing	Offshore Wind Cost Reduction Task Force	2011	(RenewableUK, 2012)
	Supply chain events	2001	(Kern et al., 2014)
Outreach	-	-	-

Wind Energy in Portugal (2000 - 2007)

The policies that mostly influenced the deployment of onshore wind energy in Portugal between 2000 and 2007 are summarised in Table C.5, according to the information provided in Section 6.2.

 Table C.5: Policies in place during the enabling phase and part of the accelerating phase (2000 - 2007) of the deployment of offshore wind energy in Portugal, as well as the year in which they were introduced and the source where this information can be obtained

Instrument type	Instruments employed	Year	Source
Economic instruments			
Financial incentives	Feed-in tariffs	1995	(IRENA, 2013b)
	Incentive Scheme for Rational Use of Energy (SIRUE)	2001-2005	(IRENA, 2013b)
	PRIME/MAPE subsidy	2007	(IEA, 2008)
	Investments in (reinforcement of) the trans- mission and distribution grid	2000	(Agência LUSA, 2004; IEA, 2005, 2009a)
	Financing from Caixa BI	2006	(TVI Notícias, 2006)
	European Investment Bank	Since 1986	(EIB, 2007)
Market-based mecha- nisms	Tenders	2005	(IEA, 2009a)
R&D support	Grants given by the countries where compa- nies had manufacturing facilities	-	(BMWi, 2020; IRENA, 2013a; MITECO, 2005)
	Partial funding of INETI	1977	(IEA, 2004b)
Regulatory instruments			
Standards	Simplification of license-granting process	2001	(IRENA, 2013b)
	Market liberalisation	2004	(IEA, 2004a)
Strategic planning	Renewables electricity generation share of 39% (later changed to 45%) by 2010	2001	(IEA, 2004a)
	3,750 MW of wind power until 2010	2003	(IEA, 2004b)
	5,100 MW of wind power until 2013	2005	(IEA, 2006a)
Market facilitation	Guaranteed grid access for Independent Power Producers	1999	(IRENA, 2013b)
Information instruments			
Knowledge transfer	-	-	-
Collaboration & network- ing	IEA R&D Wind Executive Committee	2003	(IEA, 2004b)
	Wind energy industrial cluster	2005	(J. R. Ferreira & Martins, 2009)
Outreach	-	-	-

Electric Vehicles in the UK (2015 - 2021)

A summary of the relevant policies to EV uptake in the UK between 2015 and 2021 is provided in Table C.6. An explanation of each policy is given throughout Section 6.3.

 Table C.6: Policies in place during the enabling phase and part of the accelerating phase (2015 - 2021) of the deployment of EVs in the UK, as well as the year in which they were introduced and the source where this information can be obtained

Instrument type	Instruments employed	Year	Source
Economic instruments			
Financial incentives	Plug-in Car Grant	2011	(Geels & Turnheim, 2022)
	Vehicle Excise Duty (VED) rebates	2010	(Zapmap, 2023)
	Fuel Excise Duty (FED) rebates	-	(Zapmap, 2023)
	Company Car Tax rebates	2002	(Zapmap, 2023)
	Free parking	Ongoing	(Rietmann & Lieven, 2019)
	Electric Vehicle Homecharge Scheme (EVHS)	2014	(British Gas, 2024)
	Workplace Charging Scheme	2016	(OZEV, 2016b)
	On-street Residential Chargepoint Scheme (ORCS)	2016	(OZEV, 2016a)
	Go Ultra Low Scheme	2016	(OLEV, 2014)
	Further charging infrastructure investments	2017 and 2018	(BEIS, 2017, 2018)
Market-based mecha- nisms	-	-	-
R&D support	Faraday Battery Challenge	2017	(BEIS, 2017)
	Automotive Sector Deal	2018	(BEIS, 2018)
	Innovate UK	Ongoing	(Innovate UK, 2024)
	UK Research and Innovation	2018	(UK Research and Innova tion, 2024a)
Regulatory instruments		1	
Standards	EU CO ₂ emission standards	2009	(ICCT, 2014)
	Phase-out of petrol/diesel cars	2017 (adjusted in 2020)	(BEIS, 2017)
	Clean Air Zones	2019 and 2020	(Department for Environment Food & Rural Affairs, 2020)
Strategic planning	Clean Growth Strategy	2017	(BEIS, 2017)
	Road to Zero Strategy	2018	(UK Government, 2018)
Market facilitation ²	Electric Vehicle Homecharge Scheme (EVHS)	2014	(British Gas, 2024)
	Workplace Charging Scheme	2016	(OZEV, 2016b)
	On-street Residential Chargepoint Scheme (ORCS)	2016	(OZEV, 2016a)
	Go Ultra Low Scheme	2016	(Department for Transport & OZEV, 2020; Fleet News 2016; OLEV, 2014)
	Further charging infrastructure investments	2017 and 2018	(BEIS, 2017, 2018)
Information instruments			
Knowledge transfer	Investment in education system	2018	(BEIS, 2018)
-	National Retraining Scheme	2018	(BEIS, 2018)
Collaboration & network- ing	Faraday Battery Challenge	2017	(UK Research and Innova tion, 2023)
	Automotive Sector Deal	2018	(BEIS, 2018)
Outreach	-	-	-

²Market creation/development was mainly done through the provision of a public charging infrastructure, which showed a clear investment from the UK government in creating a mass market for EVs (Geels & Ayoub, 2023).

Electric Vehicles in the Netherlands (2015 - 2021)

Table C.7 summarises the policies that targeted and affected the uptake of EVs in the Netherlands between 2015 and 2021, according to the description given in Section 6.4.

 Table C.7: Policies in place during the enabling phase and part of the accelerating phase (2015 - 2021) of the deployment of EVs in the Netherlands, as well as the year in which they were introduced and the source where this information can be obtained

Instrument type	Instruments employed	Year	Source
Economic instruments	•		
Financial incentives	Climate Agreement (infrastructure)	2019	(Nederland Elektrisch, 2020)
	Green Deal (infrastructure)	2016	(Nederland Elektrisch, 2019)
	Purchase subsidy (SEPP)	2020	(RVO, 2022b)
	Free parking	Until 2018 ³	(Yellowbrick, 2018)
	Registration and circulation tax exemption	2011 (ended in 2016 for PHEVs)	(Deuten et al., 2020)
	Reduced addition to taxable income	2014	(Deuten et al., 2020)
Market-based mecha- nisms	-	-	-
R&D support	Research and Development Promotion Act (WBSO)	1994	(RVO, 2015)
	Region and Top Sectors (MIT)	2014	(RVO, 2023b)
	Demonstration of Energy and Climate Inno- vation (DEI+)	2014	(RVO, 2023a)
	Subsidy Scheme for Demonstration of Cli- mate Technologies and Innovations in Trans- port (DKTI-Transport)	2017	(RVO, 2022a)
	Interreg	1990	(European Commission, 2020)
	LIFE	1992	(European Commission, 2022)
	Eurostars	2007	(RVO, 2011)
	Connecting Europe Facility (CEF) Transport	2014	(RVO, 2014)
Regulatory instruments			
Standards	EU CO ₂ emission standards	2009	(ICCT, 2014)
	Low-emission zones (LEZs)	2015	(Auto Siero, 2021; IEA, 2023b)
Strategic planning	100% sale of zero-emission vehicles by 2030	2017	(Deuten et al., 2020)
	EU CO ₂ emission targets for new fleets	2015	(Agency, 2016)
	Fit for 55	2021	(Nationale Agenda Laadin- frastructuur, 2022)
Market facilitation	Climate Agreement (infrastructure)	2019	(Nederland Elektrisch, 2020)
	Green Deal (infrastructure)	2016	(Nederland Elektrisch, 2019)
Information instruments			
Knowledge transfer	Formula E-team	2016	(Nederland Elektrisch, 2024a)
Collaboration & network- ing	Formula E-team	2016	(Nederland Elektrisch, 2024a)
	Dutch National Charging Infrastructure Agenda	2022	(Nationale Agenda Laadin- frastructuur, 2022)
Outreach	-	-	-

³Before 2018, when at a charging point EVs did not have to pay for parking in most cities (Yellowbrick, 2018). However, since this was different for each city it is hard to specify the first year in which EVs did not have to pay for parking.

D | Summary of Validation Interviews

In this chapter a summary is provided, per interviewee, of the interviews conducted as part of the validation process. While for most part the interviewees agreed with was presented to them, they also shared interesting insights and recommendations about certain points, which are described below.

D.1. Interviewee 1

Interviewee 1 is a manager at one of UK's Big Four consultancy firms, and has worked in the renewable energy sector. They contributed to the validation of the wind energy (Section 6.1) and the EV (Section 6.3) case studies in the UK. Furthermore, other general points were discussed with the interviewee.

When inquired about where consultancy firms would fit within the model, they agreed that it made sense to frame them as technology firms. This is because these companies often work on projects aimed at developing the technology or at defining the strategy of its implementation. Nonetheless, the interviewee mentioned that consultancy firms typically fit in different boxes.

The fact that the incumbents appeared very active in all case studies was discussed with the interviewee. They considered that generally the very large incumbents were risk averse. While that does not mean that large firms like Siemens do not innovate, according to the interviewee's experience in the UK the majority of innovations come from small and medium enterprises trying to set themselves up. This is then followed by commitment from the incumbents once the technology is more developed. For example, most people associate the large growth in EVs with Tesla, who put a lot of effort into developing and marketing the technology. However, since the beginning it was expected that once the large incumbents such as Mercedes and Volkswagen committed to EVs, they would overcome Tesla in many aspects, which was observed in reality. These insights were integrated into Chapter 7 and Chapter 8.

There was one detail concerning the role of incumbents in which the interviewee's opinion differed slightly from the case studies' findings. Based on intuition, interviewee 1 suggested that most incumbents wait for the tipping point before committing to a new technology. This did not agree with the data found in all case studies and therefore was not implemented.

D.1.1. Wind Energy in the UK

For the wind energy in the UK case study, interviewee 1 mentioned that for instance the TSB is currently very focused on collaboration agreements, providing a lot of data to other parties. In fact, many of the projects in which the interviewee has been involved stipulate that they must perform the data work in collaboration with a university. Furthermore, they stated that, apart from the academic and research institutes mentioned in the presentation, ORE Catapult is also involved in performing research.

Concerning trade associations, interviewee 1 highlighted the mutual relationship between these and policymakers. When the government is looking for feedback on policies, they perform a request for

evidence, typically next to trade associations. This helps set-out policies that are implementable, such that time and money are not wasted. For instance, interviewee 1 mentioned that the switch from Renewable Obligation Certificates to Contracts for Difference was a consequence of feedback from the industry. At the time, the government wanted to bring the incentive to a halt since it was only meant to kick-start the industry, not maintain it. However, the industry players claimed they still needed support, from which the new system emerged. Therefore, the interviewee proposed having a feedback loop between policymakers and trade associations to show this mutual interaction.

They argued that a lot of the political influence and lobbying from technology firms towards policymakers goes through trade associations. This is because in this industry there are many small businesses that do not have the weight of a trade association. Thus, the latter act as the middle man. This observation was, however, already included in the description of the case studies, although perhaps it is not immediately clear from the feedback loop models. Besides, interviewee 1 pointed out that apart from the trade associations mentioned in the presentation there was also the Renewable Energy Association.

In terms of financial institutions, the interviewee confirmed the importance of the Green Investment Bank, and stated that currently the UK Infrastructure Bank is a big player in this field. Nonetheless, the latter was only established in 2021 (UK Infrastructure Bank, 2022).

They also mentioned that the Crown Estate (CE) is a very particular organisation since they are an independent private organisation set up by an active government. They report to the government as their main shareholder and most of the CE's revenue goes to the country's Treasury.

Moreover, they said that financial institutions could be linked directly with technology firms as they provide most firms require on loans, especially the smaller ones. However, I argued that these financing solutions toward technology firms have always been there. Therefore, they were probably not critical in triggering the positive tipping point and can be left out to avoid adding unnecessary complexity to the model. The interviewee agreed with this point.

Finally, interviewee 1 argued that academic and research institutes could link directly to the sustainable energy technology block. In the model presented to them these institutes interacted with technology firms who in turn interacted with the technology. However, academic and research institutes directly research and test new technologies. Nonetheless, they said that if the intention was to show how the investment flows then probably not having them link directly to the technology made sense. Still on academic and research institutes, the interviewee agreed that technology firms provided them training but they argued that direct R&D investment was also involved.

D.1.2. EVs in the UK

Again, interviewee 1 argued that there was investment from technology firms toward academic and research institutes. As an example they mentioned the Work Manufacturing Group at the Warwick Business School, which worked very closely with Jaguar, Land Rover, among other manufacturers. This collaboration included the Advanced Propulsion Centre as well. Such investments resulted in new buildings at engineering schools, some of which dedicated solely to EVs, for instance at the new engineering school of Coventry University. Furthermore, they argued that training in both directions was definitely involved. This was already captured by the interactions named *knowledge transfer*.

The interviewee agreed with not having a direct link between trade associations and adopters. They believe that trade associations undertake surveys of EV users, but so do technology firms, consultancies, and the government. Therefore, this does seem like standard practice and the interviewee agreed that it did not need its own arrow in the model. They noted that it differed from wind energy, in which adopters are typically companies and not individuals.

Interviewee 1 also highlighted the important role of infrastructure, as the lack of charging points is one of the main reasons for not adopting an EV according to them. They consider that there have been affordable EVs available for a long time already, and that the charging infrastructure was one of the main constraints for the technology's deployment. They noted that there is still some way to go. For instance, there are lamppost chargers on some streets, but those parking spots are often occupied by cars that are not plugged in. Another concern is the fact that the network does not have enough capacity to, e.g., change all the lamppost on a street to charging points.

Concerning the contribution of incumbents, they argued that the 2015 Diesel scandal was responsible both for the tipping point in adoption, and the commitment of incumbents, with the two latter occurring at a similar time. Their opinion was based on the fact that EVs depend heavily on behavioural aspects and that the Diesel scandal showed people that their car was not as environmentally friendly as they thought, making them realise they had to do something different. However, this opinion is not supported by the case study findings. Rather the analysis suggests that the Diesel scandal led technology firms to reorient shortly after, with the tipping point occurring roughly four years after, in 2015.

D.2. Interviewee 2 - Wind Energy in Portugal

Interviewee 2 validated the case study concerning wind energy in Portugal (Section 6.2). They work at a company focused on wind energy production as a project manager.

The interviewee mentioned that the stabilisation period after 2010 was not solely due to the economic crisis but also due to the inability of the electrical grid to accommodate the energy that could be generated in locations suitable for wind projects, which has remained insufficient up to today. In the case study it was mentioned that the tender process specified the preferred points on the grid for new projects. As discussed with interviewee 2, if the grid is not expanded, these points will eventually run out.

In terms of the actors involved, interviewee 2 acknowledged that project developers do sometimes sell parks that have been built. This is common when investment funds - which specialise in managing financial investments rather than in developing or building wind parks themselves - seek assets that are already operational. Consequently, they often hire other companies to handle the technical management of the parks on their behalf. In this case, the project developers would fit as *adopters*.

However, interviewee 2 highlighted that project developers do not always correspond to *adopters* as in "people who actually buy the technology". They mentioned that it was very common for project developers to take projects to the ready-to-build phase by acquiring land, conducting preliminary studies, engineering the designs, and obtaining the necessary permits. They then sell these ready-to-build projects already with the technology (e.g., wind turbines or solar panels) to the final owner. This is very common nowadays and occurs via milestone fee payments following a co-development logic.

Other actors to consider might be consulting firms (technical, energy, environmental, legal, etc.), providing reassurance to adopters for their investments, including validating the work done by project developers. Interviewee 2 mentioned that consultancy has always been important for validating the reliability of projects in due diligence processes for bank financing.

Concerning financial institutions, interviewee 2 made a specific comment about banks, saying that these can play a significant role in providing the capital that allows these projects to materialise. Banks have always been very interested in renewable energy projects, especially when their production can be predicted with considerable certainty in the medium- to long-term, and feed-in tariffs are involved. The production estimations are then typically validated via due diligence by consultants.

With respect to the wind atlas developed by INETI, the interviewee mentioned that it serves only to provide an initial perception of the wind potential and to identify suitable locations. It is still necessary to conduct wind measurement campaigns, with the installation of measurement towers (met masts), to reliably anticipate investment returns and obtain project financing.

In specific regarding policies, nowadays the use of what could be called *compensatory instruments* is widespread. Despite them being renewable energies, these technologies still have a significant impact on the environment, territory, and/or local communities. Thus, it is common for administrative entities (e.g., municipalities) to request a compensatory measure. At times, these measures are defined by law, such as compensations to municipalities. Nowadays, it is even common for promoters to suggest compensatory measures right away (tree planting, offering vehicles for fire prevention, habitat protection measures, among others). This is indeed observed in the wind energy in Portugal case study.

The interviewee commented that public incentives were not solely financial benefits. The tender process offered the opportunity for project promoters proposing technological clusters to install capacity in the country. This setup encouraged manufacturers to participate and establish factories in Portugal. By joining these clusters and winning the auctions, they not only had access to incentives but also had a market to sell their products in, which facilitated their investment in the country.

When it comes to the feed-in tariffs, they highlighted that these were not exactly a fixed rate per MWh, but were calculated with a complex formula. The latter involved indices associated with e.g., production hours. For instance, a wind park could have an average tariff of €80/MWh and another of €100/MWh. Hence, the feed-in tariff I specify in my case study should probably be referred to as "average tariff".

Moreover, the feed-in tariffs were not strictly given for 15 years (after the revision of 2005). They were at most 15 years or until an accumulated production of a certain amount of GWh was reached. For instance, there were parks with good production whose feed-in tariff only lasted for 13 years. Therefore, the correct term should be "up to a maximum of 15 years".

Concerning feedback loops 1 and 2, the interviewee noted that in the past years the cost per MW of installing wind farms has not decreased much, even though they are not sure how it was between 2000 and 2007. Since 2007, when the interviewee joined the industry, turbines have indeed become larger, with higher power and rotor diameters. This has allowed for the installation of more MW in smaller areas and increased production in locations with less wind, which might have been disregarded previously.

D.3. Interviewee 3 - EVs in the Netherlands

Interviewee 3 is a senior manager at one of the Netherlands' Big Four accounting firms, and works in renewable energy projects. They helped validate the EVs in the Netherlands case study (Section 6.4).

One of the remarks they made concerned the importance of the infrastructure and accessibility in reaching the tipping point in EV adoption in the Netherlands. This is because, once the infrastructure provides sufficient coverage, the technology becomes usable and adaptation begins. Associated with infrastructure development they mentioned the role of national government and municipality-level policies. For instance, the environmental areas (*milieuzones*) within city centres proved to be a relevant incentive in the transition. On this note, they stressed the importance of polycentric efforts, i.e. having policymaking at different levels, in enabling positive tipping points (Geels et al., 2017).

Interviewee 3 pointed out that the Netherlands, lacking a domestic car industry, is an unexpected leader in electric vehicle adoption. Typically, one would anticipate robust government support and subsidies in countries with significant car manufacturing sectors. However, powerful lobbying efforts, particularly evident in Germany, have instead hindered progress in such countries, resulting in their lagging behind. This discrepancy highlights the irrationality of political influence on policy decisions.

When it comes to feedback loop 4, interviewee 3 mentioned that municipalities, particularly Amsterdam, were eager to establish a public charging network and demonstrated flexibility in the permitting process. Erecting these stations in public areas posed challenges, requiring adherence to a rigorous permitting procedure. The national government provided subsidies to municipalities for charger installation, enabling individuals without home chargers to request public ones. Additionally, an auction was held for fast charging spots near highways, which was overlooked by oil and gas companies but seized by Fastned at low cost. Fastned has since expanded into a major company, though confrontations with existing gas stations persist, as they cannot sell beverages, for instance.

In terms of private infrastructure investment, they also highlighted Tesla's innovative business model, performed in partnership with the Van der Valk hotels. Having chargers in the hotel's premises is beneficial for both parties. For Tesla, one of the advantages is that hotels are always closely located to highways. Interviewee 3 outlined that these collaborations helped make the tipping point realistic.

Concerning feedback loop 6 and the fact that framing EVs as offering a solution to societal problems legitimises the technology, interviewee 3 mentioned that one other very important factor is simply that owning an EV is 'cool'. For instance, even though Tesla does tick most of the ten types of innovation (Deloitte Digital, 2024), it also brings with it a unique experience that affects behavioural aspects.

Another aspect they saw as important to consider was the fact that in 2013 the Dutch Government set ambitious climate goals, yet certain aspects of the transition seemed unattainable at the time. For instance, decarbonising the steel industry posed significant challenges. However, progress could be swiftly achieved in sectors like solar, wind, and electric vehicles. These were areas ready for transition, whereas progress in other sectors lagged behind. This does not, nonetheless, imply that wind and EVs are "easy" sectors, as they faced significant challenges one or two decades ago, too. The difficulty lies in kick starting progress and establishing the necessary infrastructure. Additionally, interviewee 3 stated that subsidising early adopters is crucial. Choosing where to focus efforts is also challenging as making the wrong choice could result in delays spanning decades.

Lastly, they mentioned that general lessons from these case studies can be incredibly helpful and interesting for the industry. They highlighted that although comparing the case studies and trying to find patterns among them is perhaps this thesis' most difficult part, it is also the most interesting one.

D.4. Interviewee 4 - Final Feedback Loop Models

Interviewee 4 is a postdoctoral researcher at a Dutch technical university working closely to positive tipping points. Therefore, they validated the overall final feedback loop models (Section 7.1.1), as well as the conclusions about actor groups and actor roles (Section 7.1.2 and Section 7.1.3).

One point made by interviewee 4 concerned the definition of wider publics, especially when it comes to onshore wind. They mentioned there is a lot of literature on social acceptance of wind energy and renewable energy. That literature often makes a distinction between the general public and specific communities where the technology is located. They suggested this distinction could be integrated into this thesis' models. Besides, interviewee 4 said there was typically a difference in the level of acceptance of the general public. For instance, while the wider public might generally support wind energy, a local community might not. This difference is called the social gap.

Initially, interviewee 4 agreed that it made sense for financial institutions to have a larger impact for B2B than B2C technologies. The interviewee argued that, although it might not be as obvious as the role of large capital investments or debt provided to wind farm developers, financial institutions could still play a role in EV adoption. However, they said including this actor group on the EV case studies or not depended on the case studies' findings.

Concerning promoting education, I commented that perhaps creating these education programs did not have a direct impact on the deployment of the technologies in the time period analysed. The interviewee agreed that it was hard to be sure of the effect of this actor role and that they would also leave it out.

The interviewee pointed out that policymakers seemed to have a substantial role in most of the actor roles. They commented that oftentimes when people look at different systems policymakers come across as the most important actors, but that is not always the case. They are the ones that are in the public eye the whole time and so have the most explicit public goals and visions. However, sometimes systems are driven by other actors whose role is not so visible. For instance, global communities and wider publics can also play an important role in shaping the positions of policymakers.

Finally, interviewee 4 noted that three or two new feedback loops had been added to the model, pointing out these were 'smaller' feedbacks. They questioned whether there were any bigger feedback loops created by these additions. This comment was incorporated in Section 7.1.1 and Section 8.3.